UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

Form 1	0-Q
--------	------------

(Mark	Ono)

■ QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended September 30, 2011

or

☐ TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from

Commission File Number: 001-34186

VANDA PHARMACEUTICALS INC.

(Exact name of registrant as specified in its charter)

Delaware

(State or other jurisdiction of incorporation or organization)

03-0491827 (I.R.S. Employer Identification No.)

9605 Medical Center Drive, Suite 300 Rockville, Maryland (Address of principal executive offices)

20850 (Zip Code)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes \boxtimes No \square									
Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T (§232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). Yes \boxtimes No \square									
Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See the definitions of "large accelerated filer," "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act.									
Large accelerated filer	Accelerated filer	\boxtimes							
Non-accelerated filer \Box (Do not check if a smaller reporting company)	Smaller reporting company								
ndicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes 🗆 No 🗵									
as of November 2, 2011, there were 28,117,026 shares of the registrant's common stock issued and outstanding.									
		_							

Vanda Pharmaceuticals Inc.

Quarterly Report on Form 10-Q For the Quarter Ended September 30, 2011

INDEX

		Page
	PART I — FINANCIAL INFORMATION	
ITEM 1.	<u>Financial Statements (Unaudited):</u>	3
	Condensed Consolidated Balance Sheets as of September 30, 2011 and December 31, 2010	3
	Condensed Consolidated Statements of Operations for the three and nine months ended September 30, 2011 and 2010	4
	Condensed Consolidated Statement of Changes in Stockholders' Equity for the nine months ended September 30, 2011	5
	Condensed Consolidated Statements of Cash Flows for the nine months ended September 30, 2011 and 2010	ϵ
	Notes to Condensed Consolidated Financial Statements	7
ITEM 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	20
ITEM 3.	Quantitative and Qualitative Disclosures about Market Risk	30
ITEM 4.	Controls and Procedures	31
	PART II — OTHER INFORMATION	
ITEM 1.	<u>Legal Proceedings</u>	32
ITEM 1A.	Risk Factors	32
ITEM 2.	<u>Unregistered Sales of Equity Securities and Use of Proceeds</u>	32
ITEM 3.	<u>Defaults Upon Senior Securities</u>	32
ITEM 4.	Removed and Reserved	32
ITEM 5.	Other Information	32
ITEM 6.	<u>Exhibits</u>	32
<u>Signatures</u>		33
Exhibits		34

Part I — FINANCIAL INFORMATION

Item 1. Financial Statements (Unaudited).

VANDA PHARMACEUTICALS INC.

CONDENSED CONSOLIDATED BALANCE SHEETS (Unaudited)

(in thousands except for share amounts)	September 30, 2011	December 31, 2010
(in thousands, except for share amounts) ASSETS		
Current assets:		
Cash and cash equivalents	\$ 60,326	\$ 42,559
Marketable securities, current	101,122	155,478
Accounts receivable	1,216	511
Prepaid expenses, deposits and other current assets	1,873	1,843
Deferred tax, current portion	182	182
Total current assets	164,719	200,573
Marketable securities, non-current	19,011	_
Property and equipment, net	851	937
Other non-current assets	84	_
Intangible asset, net	8,404	9,522
Deferred tax, noncurrent portion	1,639	1,639
Restricted cash	1,030	430
Total assets	\$ 195,738	\$ 213,101
LIABILITIES AND STOCKHOLDERS' EQUITY		
Current liabilities:		
Accounts payable	\$ 1,333	\$ 648
Accrued liabilities	3,410	1,324
Accrued income taxes	2,108	2,266
Deferred revenues, current portion	26,789	26,789
Total current liabilities	33,640	31,027
Deferred rent	600	490
Deferred revenues, noncurrent portion	123,816	143,853
Total liabilities	158,056	175,370
Commitments		
Stockholders' equity:		
Preferred stock, \$0.001 par value; 20,000,000 shares authorized and none issued and outstanding at September 30, 2011 and December 31, 2010	_	_
Common stock, \$0.001 par value; 150,000,000 shares authorized as of September 30, 2011 and December 31, 2010;		
and 28,110,619 and 28,041,379 shares issued and outstanding as of September 30, 2011 and December 31, 2010, respectively	28	28
Additional paid-in capital	295,530	291,342
Accumulated other comprehensive income	293,330	291,342
Accumulated other comprehensive income Accumulated deficit	(257,920)	(253,641)
Total stockholders' equity	37,682	37,731
Total liabilities and stockholders' equity	\$ 195,738	\$ 213,101

VANDA PHARMACEUTICALS INC.

CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS (Unaudited)

		Three Mo	Nine Months Ended					
(in thousands, except for share amounts)	September 30, 2011		September 30, 2010		September 30, 2011		September 30, 2010	
					_			2010
Revenues:	.			0.000		20.02	•	20.02
Licensing agreement	\$	6,753	\$	6,753	\$	20,037	\$	20,037
Royalty revenue		1,216		493		2,863		2,630
Product sales								5,290
Total revenues		7,969		7,246		22,900		27,957
Operating expenses:								
Cost of sales, product		—		_		_		2,891
Research and development		8,174		4,072		18,440		8,516
General and administrative		2,711		2,054		8,141		7,385
Intangible asset amortization		377		377		1,118		1,118
Total operating expenses		11,262		6,503		27,699		19,910
Income (loss) from operations		(3,293)		743		(4,799)		8,047
Interest income		106		156		362		289
Income (loss) before tax provision		(3,187)		899		(4,437)		8,336
Tax provision (benefit)		(113)		(2,285)		(158)		3,343
Net income (loss)	\$	(3,074)	\$	3,184	\$	(4,279)	\$	4,993
Net income (loss) per share:								
Basic	\$	(0.11)	\$	0.11	\$	(0.15)	\$	0.18
Diluted	\$	(0.11)	\$	0.11	\$	(0.15)	\$	0.18
Shares used in calculation of net income (loss) per share:								
Basic	28	,107,363	28,	003,453	28	,104,749	27	,872,542
Diluted	28	,107,363	28,	466,532	28	,104,749		,429,223

VANDA PHARMACEUTICALS INC.

CONDENSED CONSOLIDATED STATEMENT OF CHANGES IN STOCKHOLDERS' EQUITY (Unaudited)

	Common Stock		Additional	Other			
(in thousands, except for share amounts)	Shares	Par Value	Paid-In Capital	Comprehensi Income	ve Accumulated <u>Deficit</u>	Comprehensive Income (Loss)	Total
Balances at December 31, 2010	28,041,379	\$ 28	\$291,342	\$	2 \$ (253,641)		\$37,731
Issuance of common stock from exercised stock							
options/restricted stock units	69,240	_	5	_	_		5
Employee and non-employee stock-based compensation	_	_	4,183	_	- <u>-</u>		4,183
Comprehensive loss:							
Net loss	_	_	_	_	- (4,279)	\$ (4,279)	
Net unrealized gain on marketable securities	_	_	_	۷	12 —	42	
Comprehensive loss	_	_	_	_	_	\$ (4,237)	(4,237)
Balances at September 30, 2011	28,110,619	\$ 28	\$295,530	\$ 4	\$ (257,920)		\$37,682

VANDA PHARMACEUTICALS INC.

CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (Unaudited)

	Nine Mon	onths Ended	
(in thousands)	September 30, 2011	September 30, 2010	
Cash flows from operating activities		2010	
Net income (loss)	\$ (4,279)	\$ 4,993	
Adjustments to reconcile net income (loss) to net cash used in operating activities:	,		
Depreciation and amortization	284	258	
Employee and non-employee stock-based compensation	4,183	3,647	
Loss on disposal of assets	<u> </u>	(23)	
Amortization of premiums and discounts on marketable securities	774	(12)	
Amortization of intangible asset	1,118	1,118	
Deferred tax benefit	_	(1,554)	
Changes in assets and liabilities:			
Accounts receivable	(705)	2,670	
Inventory	_	2,399	
Prepaid expenses, deposits and other current assets	(114)	190	
Accounts payable	685	(2,128)	
Accrued liabilities	2,086	(545)	
Accrued income taxes	(158)	3,235	
Other liabilities	110	(13)	
Deferred revenue	(20,037)	(20,037)	
Net cash used in operating activities	(16,053)	(5,802)	
Cash flows from investing activities			
Purchases of property and equipment	(198)	_	
Proceeds from sale of property and equipment	_	66	
Purchases of marketable securities	(140,637)	(124,028)	
Maturities of marketable securities	175,250	24,500	
Change in restricted cash	(600)		
Net cash provided by (used in) investing activities	33,815	(99,462)	
Cash flows from financing activities			
Excess tax benefits from stock-based compensation	<u> </u>	1,662	
Proceeds from exercise of stock options	5	775	
Net cash provided by financing activities	5	2,437	
Net change in cash and cash equivalents	17,767	(102,827)	
Cash and cash equivalents			
Beginning of period	42,559	205,295	
End of period	\$ 60,326	\$ 102,468	

VANDA PHARMACEUTICALS INC.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (unaudited)

1. Business Organization and Presentation

Business organization

Vanda Pharmaceuticals Inc. (Vanda or the Company) is a biopharmaceutical company focused on the development and commercialization of products for the treatment of central nervous system disorders. Vanda commenced its operations in 2003. The Company's lead product, Fanapt® (iloperidone), which Novartis Pharma AG (Novartis) began marketing in the U.S. in the first quarter of 2010, is a compound for the treatment of schizophrenia. On May 6, 2009, the U.S. Food and Drug Administration (FDA) granted U.S. marketing approval of Fanapt® for the acute treatment of schizophrenia in adults. On October 12, 2009, Vanda entered into an amended and restated sublicense agreement with Novartis. Vanda had originally entered into a sublicense agreement with Novartis on June 4, 2004 pursuant to which Vanda obtained certain worldwide exclusive licenses from Novartis relating to Fanapt®. Pursuant to the amended and restated sublicense agreement, Novartis has exclusive commercialization rights to all formulations of Fanapt® in the U.S. and Canada. On January 11, 2010, Novartis launched Fanapt® in the U.S. Novartis is responsible for the further clinical development activities in the U.S. and Canada, including the development of a long-acting injectable (or depot) formulation of Fanapt®. Pursuant to the amended and restated sublicense agreement, Vanda received an upfront payment of \$200.0 million at the end of 2009 and is eligible for additional payments totaling up to \$265.0 million upon the achievement of certain commercial and development milestones for Fanapt® in the U.S. and Canada. Vanda also receives royalties, which, as a percentage of net sales, are in the low double-digits, on net sales of Fanapt® in the U.S. and Canada. In addition, Vanda is no longer required to make any future milestone payments with respect to sales of Fanapt® or any future royalty payments with respect to sales of Fanapt® in the U.S. and Canada. Vanda retains exclusive rights to Fanapt® outside the U.S. and Canada and Vanda has exclusive rights to use any of Novartis' data for Fanapt® for developing and commercializing Fanapt® outside the U.S. and Canada. At Novartis' option, Vanda will enter into good faith discussions with Novartis relating to the co-commercialization of Fanapt® outside of the U.S. and Canada or, alternatively, Novartis will receive a royalty on net sales of Fanapt® outside of the U.S. and Canada. Novartis has chosen not to co-commercialize Fanapt® with Vanda in Europe and certain other countries and will instead receive a royalty on net sales in those countries. These include, but are not limited to, the countries in the European Union, as well as Switzerland, Norway, Liechtenstein, and Iceland. Vanda continues to explore the regulatory path and commercial opportunity for Fanapt® oral formulation outside of the U.S. and Canada. On November 1, 2010, the Therapeutic Goods Administration of Australia's Department of Health and Ageing, accepted for evaluation Vanda's application for marketing approval for the Fanapt® oral formulation. On July 22, 2011, the European Medicines Agency (EMA) notified Vanda that it had accepted for evaluation the Marketing Authorization Application (MAA) for oral iloperidone tablets. Vanda has entered into agreements with the following partners for the commercialization of Fanapt® in the countries set forth below:

> Country Mexico Argentina

Probiomed S.A. de C.V. Biotoscana Farma S.A.

Tasimelteon is an oral compound in development for the treatment of sleep and mood disorders including Circadian Rhythm Sleep Disorders (CRSD). On January 19, 2010, the FDA granted orphan drug designation status for tasimelteon in a specific CRSD, Non-24-Hour Sleep-Wake Disorder (N24HSWD) in blind individuals without light perception. The FDA grants orphan drug designation to drugs that may provide significant therapeutic advantage over existing treatments and target conditions affecting 200,000 or fewer U.S. patients per year. Orphan drug designation provides potential financial and regulatory incentives including, study design assistance, waiver of FDA user fees, tax credits, and up to seven years of market exclusivity upon marketing approval. On February 23, 2011, the European Commission (EC) designated tasimelteon as an orphan medicinal product for the same indication. Vanda initiated four clinical trials to pursue FDA approval of tasimelteon for the treatment of N24HSWD in blind individuals without light perception. Two of the clinical trials were initiated in the third quarter of 2010, the third was initiated in the third quarter of 2011 and the fourth was initiated in October 2011. The first clinical trial (3201) is a randomized, double-blind, placebo-controlled study with a planned enrollment of approximately 100 patients with N24HSWD. The trial has a six month treatment period and includes measures of both nighttime and daytime sleep, as well as laboratory measures of the synchronization between the internal body clock and the 24-hour environmental light/dark cycle. The second clinical trial (3202) is a one-year safety study of tasimelteon for the treatment of N24HSWD. This trial is an openlabel safety study with a planned enrollment of up to 140 patients with N24HSWD. The third clinical trial (3203) is a placebo-controlled, randomized withdrawal study to examine the maintenance of effect of tasimelteon for the treatment of N24HSWD. Patients will be observed for 8 weeks during which nighttime and daytime sleep as well as synchronization of their internal body clock to the 24-hour day will continue to be evaluated. The fourth clinical trial (3204) is a two year open-label, multicenter, study in blind subjects with N24HSWD to assess the safety of tasimelteon. Vanda plans to conduct these clinical trials over the next one to two years to support the use of tasimelteon as a circadian regulator and the submission of a new drug application (NDA) to the FDA and a marketing authorization application to the EMA. Vanda initiated a Phase IIb/III clinical trial to study the efficacy of tasimelteon for the treatment of Major Depressive Disorder (MDD) in the third quarter of 2011. The clinical trial is a randomized, double-blind, placebo-controlled study with planned enrollment of approximately 500 patients with MDD. The trial has an eight-week treatment period, followed by an optional one-year open-label extension, and includes measures of depression and anxiety symptoms, nighttime and daytime sleep, as well as laboratory measures of the internal body clock. Given the range of potential indications for tasimelteon, Vanda may pursue one or more partnerships for the development and commercialization of tasimelteon worldwide.

VANDA PHARMACEUTICALS INC.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (unaudited) - (continued)

Throughout this quarterly report on Form 10-Q, Vanda refers to Fanapt® within the U.S. and Canada as its partnered product and Vanda refers to Fanapt® outside the U.S. and Canada and tasimelteon as its products. All other compounds are referred to as Vanda's product candidates. In addition, Vanda refers to its partnered products, products and product candidates collectively as its compounds. Moreover, Vanda refers to drug products generally as drugs or products.

Basis of presentation

The accompanying unaudited condensed consolidated financial statements have been prepared in accordance with U.S. generally accepted accounting principles (GAAP) for interim financial information and the instructions to Form 10-Q and Article 10 of Regulation S-X. Accordingly, they do not include all the information and footnotes required by GAAP for complete financial statements and should be read in conjunction with the Company's consolidated financial statements for the year ended December 31, 2010 included in the Company's annual report on Form 10-K. The financial information as of September 30, 2011 and for the three and nine months ended September 30, 2011 and 2010, is unaudited, but in the opinion of management, all adjustments, consisting only of normal recurring accruals, considered necessary for a fair statement of the results of these interim periods have been included. The condensed consolidated balance sheet data as of December 31, 2010 was derived from audited financial statements but does not include all disclosures required by GAAP.

The results of the Company's operations for any interim period are not necessarily indicative of the results that may be expected for any other interim period or for a full fiscal year. The financial information included herein should be read in conjunction with the consolidated financial statements and notes in the Company's Annual Report on Form 10-K for the year ended December 31, 2010.

2. Summary of Significant Accounting Policies

Use of estimates

The preparation of financial statements in conformity with GAAP requires management to make estimates that affect the reported amounts of assets and liabilities at the date of the financial statements, disclosure of contingent assets and liabilities, and the reported amounts of revenue and expenses during the reporting period. Actual results could differ from those estimates.

Cash and cash equivalents

For purposes of the condensed consolidated balance sheets and condensed consolidated statements of cash flows, cash equivalents represent highly-liquid investments with a maturity date of three months or less at the date of purchase.

Marketable securities

The Company classifies all of its marketable securities as available-for-sale securities. The Company's investment policy requires the selection of high-quality issuers, with bond ratings of AAA to A1+/P1. Available-for-sale securities are carried at fair market value, with unrealized gains and losses reported as a component of stockholders' equity in accumulated other comprehensive income/loss. Interest and dividend income is recorded when earned and included in interest income. Premiums and discounts on marketable securities are amortized and accreted, respectively, to maturity and included in interest income. The Company uses the specific identification method in computing realized gains and losses on the sale of investments, which would be included in the condensed consolidated statements of operations when generated. Marketable securities with a maturity of more than one year as of the balance sheet date, and for which the Company does not intend to sell within the next twelve months are classified as non-current. All other marketable securities are classified as current.

VANDA PHARMACEUTICALS INC.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (unaudited) - (continued)

Intangible asset, net

Costs incurred for products or product candidates not yet approved by the FDA and for which no alternative future use exists are recorded as expense. In the event a product or product candidate has been approved by the FDA or an alternative future use exists for a product or product candidate, patent and license costs are capitalized and amortized over the expected patent life of the related product or product candidate. Milestone payments to the Company's partners are recognized when it is deemed probable that the milestone event will occur.

As a result of the FDA's approval of the NDA for Fanapt® in May 2009, the Company met a milestone under its original sublicense agreement with Novartis which required the Company to make a payment of \$12.0 million to Novartis. The \$12.0 million is being amortized on a straight line basis over the remaining life of the U.S. patent for Fanapt®, which the Company expects to last until May 15, 2017. This includes the Hatch-Waxman extension that extends patent protection for drug compounds for a period of up to five years to compensate for time spent in development and a six-month pediatric term extension. This term is the Company's best estimate of the life of the patent; if, however, the Hatch-Waxman or pediatric extensions are not granted, the intangible asset will be amortized over a shorter period.

The carrying values of intangible assets are periodically reviewed to determine if the facts and circumstances suggest that a potential impairment may have occurred. The Company had no impairments of its intangible assets for the nine months ended September 30, 2011.

Fair value of financial instruments

The carrying amounts of the Company's financial instruments, which include cash and cash equivalents, marketable securities and accounts receivable, approximate their fair values due to their short term nature.

Property and equipment

Property and equipment are stated at cost less accumulated depreciation and amortization. Depreciation of property and equipment is provided on a straight-line basis over the estimated useful lives of the assets. Amortization of leasehold improvements is provided on a straight-line basis over the shorter of their estimated useful life or the lease term. The costs of additions and improvements are capitalized, and repairs and maintenance costs are charged to operations in the period incurred. Upon retirement or disposition of property and equipment, the cost and accumulated depreciation and amortization are removed from the accounts and any resulting gain or loss is reflected in the Company's statement of operations for that period.

Revenue Recognition

The Company's revenues are derived primarily from the amended and restated sublicense agreement with Novartis and include an up-front payment, product sales and future milestone and royalty payments. Revenue is considered both realizable and earned when each one of the following four conditions is met: (1) persuasive evidence of an arrangement exists, (2) the arrangement fee is fixed or determinable, (3) delivery or performance has occurred and (4) collectability is reasonably assured. Pursuant to the amended and restated sublicense agreement, Novartis has the right to commercialize and develop Fanapt® in the U.S. and Canada. Under the amended and restated sublicense agreement, the Company received an upfront payment of \$200.0 million in December of 2009. Pursuant to the amended and restated sublicense agreement, the Company and Novartis established a Joint Steering Committee (JSC) following the effective date of the amended and restated sublicense agreement. The Company expects to have an active role on the JSC and concluded that the JSC constitutes a deliverable under the amended and restated sublicense agreement and that revenue related to the upfront payment will be recognized ratably on a straight-line basis over the term of the JSC; however, the delivery or performance has no term as the exact length of the JSC is undefined. As a result, the Company deems the performance period of the JSC to be the life of the U.S. patent of Fanapt®, which the Company expects to last until May 15, 2017. This includes the Hatch-Waxman extension that provides patent protection for drug compounds for a period of up to five years to compensate for time spent in development and a sixmonth pediatric term extension. This term is the Company's best estimate of the life of the patent. Revenue will be recognized ratably on a straight-line basis from the date the amended and restated sublicense agreement became effective (November 27, 2009) through the expected life of the U.S. patent for Fanapt® (May 15, 2017). Revenue related to p

VANDA PHARMACEUTICALS INC.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (unaudited) - (continued)

Concentrations of credit risk

Financial instruments which potentially subject the Company to significant concentrations of credit risk consist primarily of cash, cash equivalents and marketable securities. The Company places its cash, cash equivalents and marketable securities with what the Company believes to be highly-rated financial institutions. At September 30, 2011, the Company maintained all of its cash, cash equivalents and marketable securities in three financial institutions. Deposits held with these institutions may exceed the amount of insurance provided on such deposits. Generally, these deposits may be redeemed upon demand, and the Company believes there is minimal risk of losses on such balances.

Accrued expenses

The Company's management is required to estimate accrued expenses as part of the process of preparing the financial statements. The estimation of accrued expenses involves identifying services that have been performed on the Company's behalf, and then estimating the level of service performed and the associated cost incurred for such services as of each balance sheet date in the financial statements. Accrued expenses include professional service fees, such as lawyers and accountants, contract service fees, such as those under contracts with clinical monitors, data management organizations and investigators in conjunction with clinical trials, fees to contract manufacturers in conjunction with the production of clinical materials, and fees for marketing and other commercialization activities. Pursuant to management's assessment of the services that have been performed on clinical trials and other contracts, the Company recognizes these expenses as the services are provided. Such management assessments include, but are not limited to: (1) an evaluation by the project manager of the work that has been completed during the period, (2) measurement of progress prepared internally and/or provided by the third-party service provider, (3) analyses of data that justify the progress, and (4) management's judgment. In the event that the Company does not identify certain costs that have begun to be incurred or the Company under- or over-estimates the level of services performed or the costs of such services, the Company's reported expenses for such period would be too low or too high.

Research and development expenses

The Company's research and development expenses consist primarily of fees for services provided by third parties in connection with the clinical trials, costs of contract manufacturing services, milestone license fees, costs of materials used in clinical trials and research and development, costs for regulatory consultants and filings, depreciation of capital resources used to develop products, related facilities costs, and salaries, other employee related costs and stock-based compensation for the research and development personnel. The Company expenses research and development costs as they are incurred for compounds in the development stage, including certain payments made under the license agreements prior to FDA approval. Prior to FDA approval, all Fanapt® manufacturing-related and milestone license costs were included in research and development expenses. Subsequent to FDA approval of Fanapt®, manufacturing and milestone license costs related to this product are being capitalized. Costs related to the acquisition of intellectual property have been expensed as incurred since the underlying technology associated with these acquisitions were made in connection with the Company's research and development efforts and have no alternative future use. Milestone license payments are accrued in accordance with the FASB guidance on accounting for contingencies which states that milestone payments be accrued when it is deemed probable that the milestone event will be achieved.

General and administrative expenses

General and administrative expenses consist primarily of salaries, other employee related costs and stock-based compensation for personnel serving executive, business development, marketing, finance, accounting, information technology, marketing and human resource functions, facility costs not otherwise included in research and development expenses, insurance costs and professional fees for legal, accounting and other professional services. General and administrative expenses also include third party expenses incurred to support business development, marketing and other business activities related to Fanapt®.

Employee stock-based compensation

The Company accounts for its stock-based compensation expenses in accordance with the FASB guidance on share-based payments which were adopted on January 1, 2006. Accordingly, compensation costs for all stock-based awards to employees and directors are measured based on the grant date fair value of those awards and recognized over the period during which the employee or director is required to perform service in exchange for the award. The Company generally recognizes the expense over the award's vesting period.

VANDA PHARMACEUTICALS INC.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (unaudited) - (continued)

The fair value of stock options granted is amortized using the accelerated attribution method. The fair value of restricted stock units (RSUs) awarded is amortized using the straight line method. As stock-based compensation expense recognized in the consolidated statements of operations is based on awards ultimately expected to vest, it has been reduced for estimated forfeitures. Forfeitures are required to be estimated at the time of grant and revised, if necessary, in subsequent periods if actual forfeitures differ from those estimates. Pre-vesting forfeitures on the options granted prior to 2009 were estimated to be approximately 2%. The forfeiture rate was increased to 4% in 2009, which has been utilized for all subsequently granted options based on the Company's historical experience.

Total employee stock-based compensation expense recognized during the three and nine months ended September 30, 2011 and 2010 was comprised of the following:

	 Three Mo	Nine Months Ended					
	September 30, 2011		September 30, 2010		September 30, 2011		ember 30, 2010
(in thousands)	 						
Research and development	\$ 558	\$	429	\$	1,896	\$	1,983
General and administrative	 704		368		2,278		1,547
Stock-based compensation expense	\$ 1,262	\$	797	\$	4,174	\$	3,530

As of September 30, 2011, \$6.3 million of total unrecognized compensation costs related to non-vested awards are expected to be recognized over a weighted average period of 1.21 years.

As of September 30, 2011, the Company had two equity incentive plans, the Second Amended and Restated Management Equity Plan (the 2004 Plan) and the 2006 Equity Incentive Plan (the 2006 Plan) that were adopted in December 2004 and April 2006, respectively. An aggregate of 677,145 shares were subject to outstanding options granted under the 2004 Plan as of September 30, 2011, and no additional options will be granted under this plan. As of September 30, 2011, there were 6,741,579 shares of the Company's common stock reserved under the 2006 Plan of which 3,859,230 shares were subject to outstanding options and RSUs issued to employees and non-employees.

Options are subject to terms and conditions established by the compensation committee of the board of directors. None of the stock-based awards are classified as a liability as of September 30, 2011. Option awards have 10-year contractual terms and all options granted prior to December 31, 2006, options granted to new employees, and certain options granted to existing employees vest and become exercisable on the first anniversary of the grant date with respect to 25% of the shares subject to the option awards. The remaining 75% of the shares subject to the option awards vest and become exercisable monthly in equal installments thereafter over three years. Certain option awards granted to existing employees after December 31, 2006 vest and become exercisable monthly in equal installments over four years. The initial stock options granted to directors upon their election vest and become exercisable in equal monthly installments over a period of four years, while the subsequent annual stock option grants to directors vest and become exercisable in equal monthly installments over a period of one year. Certain option awards to executives and directors provide for accelerated vesting if there is a change in control of the Company. Certain option awards to employees and executives provide for accelerated vesting if the respective employee's or executive's service is terminated by the Company for any reason other than cause or permanent disability.

The fair value of each option award is estimated on the date of grant using the Black-Scholes-Merton option pricing model that uses the assumptions noted in the following table. Expected volatility rates are based on the Company's historical volatility of its publicly traded common stock blended with the historical volatility of the common stock of comparable entities. The expected term of options granted is based on the transition approach provided by FASB guidance as the options meet the "plain vanilla" criteria required by this guidance. The risk-free interest rates are based on the U.S. Treasury yield for a period consistent with the expected term of the option in effect at the time of the grant. The Company has not paid cash dividends to its stockholders since its inception and does not plan to pay any such dividends in the foreseeable future.

VANDA PHARMACEUTICALS INC.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (unaudited) - (continued)

Assumptions used in the Black-Scholes-Merton option pricing model for employee and director stock options granted during the nine months ended September 30, 2011 and 2010 were as follows:

	Nine Months	Ended
	September 30, 2011	September 30, 2010
Expected dividend yield	0%	0%
Weighted average expected volatility	73%	68%
Weighted average expected term (years)	6.03	6.03
Weighted average risk-free rate	2.42%	2.38%

A summary of option activity for the 2004 Plan as of September 30, 2011, and changes during the nine months then ended is presented below:

(in thousands, except for share amounts)	Number of Shares	Weighted Average Exercise Price at Grant Date		Average Exercise Price at		Average Exercise Price at Grant Date		Weighted Average Remaining Term (Years)		
Outstanding at December 31, 2010	680,754	\$	1.77							
Exercised	(3,609)	\$	0.33							
Forfeited	_		_							
Cancelled										
Outstanding at September 30, 2011	677,145	\$	1.78	4.03	\$	2,145				
Exercisable at September 30, 2011	677,145	\$	1.78	4.03	\$	2,145				

A summary of option activity for the 2006 Plan as of September 30, 2011, and changes during the nine months then ended is presented below:

(in thousands, except for share amounts)	Number of Shares	Weighted Average Exercise Price at Grant Date		Weighted Average Remaining Term (Years)	ite Intrinsic /alue
Outstanding at December 31, 2010	3,324,790	\$	14.07		
Granted	184,500	\$	7.09		
Exercised	(3,569)	\$	1.02		
Forfeited	(26,764)	\$	9.24		
Cancelled	(2,790)	\$	11.66		
Outstanding at September 30, 2011	3,476,167	\$	13.75	7.38	\$ 431
Exercisable at September 30, 2011	2,168,193	\$	16.42	6.65	\$ 307

The weighted average grant-date fair value of options granted during the nine months ended September 30, 2011 was \$4.63 per share. For the nine months ended September 30, 2011 and 2010, the amounts received by the Company in cash from options exercised under the stock-based arrangements were not material.

A RSU is a stock award that entitles the holder to receive shares of the Company's common stock as the award vests. The fair value of each RSU is based on the closing price of the Company's stock on the date of grant which equals the RSUs intrinsic value. As of September 30, 2011, there was \$2.9 million of total unrecognized compensation cost related to unvested RSU awards granted under the Company's stock incentive plans.

A summary of RSU activity for the 2006 Plan as of September 30, 2011, and changes during the nine months then ended are as follows:

(in thousands, except for share amounts)	Number of Shares	Weighted Average Price/Share				gregate nsic Value
Unvested at December 31, 2010	359,563	\$	9.75	\$ 3,401		
Granted	34,000	\$	7.11			
Vested	(2,500)	\$	0.80			
Cancelled	(8,000)	\$	9.57			
Unvested at September 30, 2011	383,063	\$	9.56	\$ 1,896		

VANDA PHARMACEUTICALS INC.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (unaudited) - (continued)

Income taxes

The Company accounts for income taxes under the liability method in accordance with the FASB provisions on accounting for income taxes, which requires companies to account for deferred income taxes using the asset and liability method. Under the asset and liability method, current income tax expense or benefit is the amount of income taxes expected to be payable or refundable for the current year. A deferred income tax asset or liability is recognized for future tax consequences attributable to differences between the financial statement carrying amounts of existing assets and liabilities and their respective tax bases and tax credits and loss carryforwards. Deferred tax assets are reduced by a valuation allowance when, in the opinion of management, it is more likely than not that some portion or all of the deferred tax assets will not be realized. Tax rate changes are reflected in income during the period such changes are enacted. Changes in ownership may limit the amount of net operating loss carryforwards that can be utilized in the future to offset taxable income.

Recent Accounting Pronouncements

In June 2011, the FASB issued an Accounting Standards Update which eliminates the option to report other comprehensive income and its components in the statement of changes in stockholders' equity. It requires an entity to present total comprehensive income, which includes the components of net income and the components of other comprehensive income either in a single continuous statement or in two separate but consecutive statements. This pronouncement is effective for financial statements issued for annual and interim periods within the first annual period beginning after December 15, 2011. The Company believes the adoption of this pronouncement will not have a material impact on its financial position or results of operations.

3. Earnings per Share

Net income (loss) is calculated in accordance with FASB guidance on earnings per share. Basic earnings per share (EPS) is calculated by dividing the net income (loss) by the weighted average number of shares of common stock outstanding, reduced by the weighted average unvested shares of common stock subject to repurchase. Diluted EPS is computed by dividing the net income (loss) by the weighted average number of shares of common stock outstanding, plus potential outstanding common stock for the period. Potential outstanding common stock options and RSUs, but only to the extent that their inclusion is dilutive.

The following table presents the calculation of basic and diluted net income (loss) per share of common stock for the three and nine months ended September 30, 2011 and 2010:

	Three Mon	nths Ended	Nine Months Ended			
	September 30, 2011	September 30, 2010	September 30, 2011	September 30, 2010		
(in thousands, except for share amounts)						
Numerator:						
Net income (loss)	\$ (3,074)	\$ 3,184	\$ (4,279)	\$ 4,993		
Denominator:						
Weighted average shares of common stock outstanding, basic	28,107,363	28,003,453	28,104,749	27,872,542		
Stock options and restricted stock units related to the issuance						
of common stock	_	463,079	_	556,681		
Weighted average shares of common stock outstanding, diluted	28,107,363	28,466,532	28,104,749	28,429,223		
Net income (loss) per share:						
Basic	\$ (0.11)	\$ 0.11	\$ (0.15)	\$ 0.18		
Diluted	\$ (0.11)	\$ 0.11	\$ (0.15)	\$ 0.18		
Anti-dilutive securities not included in diluted net income (loss) per share calculation:						
Options to purchase common stock and restricted stock units	4,035,526	3,768,374	3,719,099	3,786,374		

VANDA PHARMACEUTICALS INC.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (unaudited) - (continued)

4. Marketable Securities

The following is a summary of the Company's available-for-sale marketable securities as of September 30, 2011:

	Amortized	Gross Unrealized	Gross Unrealized	Fair Market	
(in thousands)	Cost	Gains	Losses	Value	
Current:					
U.S. Treasury and government agencies	\$ 36,743	\$ 9	\$ (1)	\$ 36,751	
U.S. corporate debt	64,346	35	(10)	64,371	
	\$101,089	\$ 44	\$ (11)	\$ 101,122	
Non-current:					
U.S. Treasury and government agencies	\$ 19,000	\$ 11	\$ —	\$ 19,011	
	\$ 19,000	\$ 11	<u> </u>	\$ 19,011	

The following is a summary of the Company's available-for-sale marketable securities as of December 31, 2010:

(in thousands) Current:	Amortized Cost	Gross Unrealized Gains	Gross Unrealized Losses	Fair Market Value
U.S. Treasury and government agencies	\$ 45,466	\$ —	\$ (11)	\$ 45,455
U.S. corporate debt	110,010	27	(14)	110,023
	\$155,476	\$ 27	\$ (25)	\$ 155,478

5. Prepaid Expenses, Deposits and Other Current Assets

The following is a summary of the Company's prepaid expenses, deposits and other current assets, as of September 30, 2011 and December 31, 2010:

(in thousands)	September 30, 2011	December 31, 2010
Prepaid insurance	\$ 295	\$ 244
Other prepaid expenses and vendor advances	1,289	966
Accrued interest income	289	633
Total prepaid expenses, deposits and other current assets	\$ 1,873	\$ 1,843

6. Property and Equipment, Net

The following is a summary of the Company's property and equipment-at cost, as of September 30, 2011 and December 31, 2010:

(in thousands)	Estimated Useful Life (Years)	September 30, 2011	December 31, 2010
Laboratory equipment	5	\$ 1,273	\$ 1,282
Computer equipment	3	917	764
Furniture and fixtures	7	700	706
Leasehold improvements	10	850	844
		3,740	3,596
Less—accumulated depreciation and amortization		(2,889)	(2,659)
		\$ 851	\$ 937

VANDA PHARMACEUTICALS INC.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (unaudited) - (continued)

Depreciation and amortization expense for the nine months ended September 30, 2011 and 2010 was \$0.3 million and \$0.3 million, respectively.

7. Intangible Asset, Net

The intangible asset consists of the following as of September 30, 2011:

			September 30, 2011	
(in thousands)	Estimated Useful Life <u>(Years)</u>	Gross Carrying Amount	Accumulated Amortization	Net Carrying Amount
Fanapt®	8	\$12,000	\$ 3,596	\$ 8,404
		\$12,000	\$ 3,596	\$ 8,404

The intangible asset consisted of the following as of December 31, 2010:

			December 31, 2010		
(in thousands)	Estimated Useful Life (Years)	Gross Carrying Amount	Accumulated Amortization	Net Carrying Amount	
Fanapt®	8	\$12,000	\$ 2,478	\$ 9,522	
		\$12,000	\$ 2,478	\$ 9,522	

On May 6, 2009, the Company announced that the FDA had approved the NDA for Fanapt®. As a result of the FDA's approval of the NDA for Fanapt®, the Company met a milestone under its original sublicense agreement with Novartis which required the Company to make a payment of \$12.0 million to Novartis. The \$12.0 million is being amortized on a straight line basis over the remaining life of the U.S. patent for Fanapt®, which the Company expects to last until May 15, 2017. This includes the Hatch-Waxman extension that provides patent protection for drug compounds for a period of up to five years to compensate for time spent in development and a six-month pediatric term extension. This term is the Company's best estimate of the life of the patent; if, however, the Hatch-Waxman or pediatric extensions are not granted, the intangible asset will be amortized over a shorter period.

Intangible assets are amortized over their estimated useful economic life using the straight line method. Amortization expense was \$1.1 million for each of the nine months ended September 30, 2011 and 2010. The Company capitalized and began amortizing the asset immediately following the FDA approval of the NDA for Fanapt®.

8. Accrued Liabilities

The following is a summary of accrued liabilities as of September 30, 2011 and December 31, 2010:

(in thousands)	September 30, 2011		ember 31, 2010
Accrued research and development expenses	\$ 2,385	\$	1,061
Accrued consulting and other professional fees	243		201
Employee benefits	782		62
Total accrued liabilities	\$ 3,410	\$	1,324

VANDA PHARMACEUTICALS INC.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (unaudited) - (continued)

9. Revenue Recognition

The Company's revenue activity for the nine months ended September 30, 2011 consisted of the following:

(in thousands)	December 31, 2010 Deferred Revenue				Revenue Recognized	mber 30, 2011 rred Revenue
Revenues:						
Licensing agreement	\$	170,642	\$ 20,037	\$ 150,605		
Royalty revenue		_	2,863	_		
Total revenues	\$	170,642	\$ 22,900	\$ 150,605		

Vanda entered into an amended and restated sublicense agreement with Novartis on October 12, 2009, pursuant to which Novartis has the right to commercialize and develop Fanapt® in the U.S. and Canada. Under the amended and restated sublicense agreement, Vanda received an upfront payment of \$200.0 million in December of 2009. Revenue will be recognized ratably from the date the amended and restated sublicense agreement became effective (November 27, 2009) through the expected life of the U.S. patent for Fanapt® (May 15, 2017). This includes the Hatch-Waxman extension that provides patent protection for drug compounds for a period of up to five years to compensate for time spent in development and a six-month pediatric term extension. This term is the Company's best estimate of the life of the patent. For the nine months ended September 30, 2011, the Company recognized \$20.0 million of revenue for the licensing agreement. Vanda recognized royalty revenue of \$2.9 million for the nine months ended September 30, 2011. Royalty revenue is based on a percentage of the quarterly net sales of Fanapt® sold in the U.S. and Canada by Novartis and is recorded when reliably measurable and earned.

10. Commitments and Contingencies

The following table summarizes our long-term cash commitments as of September 30, 2011:

	Cash payments due by period							
	October to					After		
(in thousands)	Total	Decer	nber 2011	2012	2013	2014	2015	2015
Operating leases	\$15,202	\$	181	\$ 749	\$1,547	\$1,847	\$1,897	\$8,981
Consulting fees	3,600		900	2,700				
Total	\$18,802	\$	1,081	\$3,449	\$1,547	\$1,847	\$1,897	\$8,981

Operating leases

The Company's commitments related to operating leases shown above consist of payments relating to real estate leases for its current headquarters located in Rockville, Maryland, which expires in 2016, and its future headquarters located in Washington, D.C., which expires in 2023. On July 25, 2011, the Company entered into a lease with Square 54 Office Owner LLC (the Landlord) for Vanda's future headquarters, consisting of 21,400 square feet at 2200 Pennsylvania Avenue, N.W. in Washington, D.C. (the Lease). Under the Lease, which will have an 11 year term commencing on April 1, 2012, the Company will pay \$1.6 million in annual rent over the term of the Lease; however, rent will be abated for the first 12 months. The Landlord will provide the Company with an allowance of \$1.9 million for construction of the premises to the Company's specifications. Subject to the prior rights of other tenants in the building, the Company will have the right to renew the Lease for five years following the expiration of its original term. The Company will also have the right to sublease or assign all or a portion of the premises, subject to standard conditions. The Lease may be terminated early by the Company or the Landlord upon certain conditions. The Company paid a security deposit of \$0.5 million upon execution of the Lease. The Company will likely incur costs of between \$1.0 million to \$1.5 million in connection with an early termination of the lease for the Company's current headquarters in Rockville, Maryland. These costs include a termination fee and other lease exit costs. As of September 30, 2011, the Company had not incurred any of these costs. These costs are not included in the table above; however, in the event that the Company exits the lease for its current headquarters in Rockville, Maryland, its contractual cash obligations included in the table above would be reduced by \$2.4 million between 2013 and 2016. In the event that the Company does not sublease its current headquarters, it expects to have completed the ex

Consulting fees

The Company has engaged a regulatory consultant to assist in the Company's efforts to prepare, file and obtain FDA approval of a NDA for tasimelteon. During the initial 15-month term of the engagement, the Company is obligated to pay consulting fees in the aggregate amount of up to \$3.6 million, of which \$0.9 million will be expensed in the fourth quarter of 2011, and the remainder of which will be expensed between January 1, 2012 and December 31, 2012. As part of this engagement, and subject to certain conditions, the Company will be obligated to make milestone payments in the aggregate amount of \$2.8 million upon the achievement of certain milestones, including \$2.0 million in the event that a tasimelteon NDA is approved by the FDA. In addition to these fees and milestone payments, the Company is obligated to reimburse the consultant for its ordinary and necessary business expenses incurred in connection with its engagement. The Company may terminate the engagement at any time upon prior notice; however, subject to certain conditions, the Company will remain obligated to make some or all of the milestone payments if the milestones are achieved following such termination.

VANDA PHARMACEUTICALS INC.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (unaudited) - (continued)

Guarantees and indemnifications

The Company has entered into a number of standard intellectual property indemnification agreements in the ordinary course of its business. Pursuant to these agreements, the Company indemnifies, holds harmless, and agrees to reimburse the indemnified party for losses suffered or incurred by the indemnified party, generally the Company's business partners or customers, in connection with any U.S. patent or any copyright or other intellectual property infringement claim by any third party with respect to the Company's products. The term of these indemnification agreements is generally perpetual from the date of execution of the agreement. The maximum potential amount of future payments the Company could be required to make under these indemnification agreements is unlimited. The Company also indemnifies its officers and directors for certain events or occurrences, subject to certain conditions. Since inception, the Company has not incurred costs to defend lawsuits or settle claims related to these indemnification agreements. The Company believes that the fair value of the indemnification agreements is minimal, and accordingly the Company has not recognized any liabilities relating to these agreements as of September 30, 2011.

License agreements

The Company's rights to develop and commercialize its products are subject to the terms and conditions of licenses granted to the Company by other pharmaceutical companies.

Fanapt®. The Company acquired exclusive worldwide rights to patents and patent applications for Fanapt® (iloperidone), in 2004 through a sublicense agreement with Novartis. A predecessor company of sanofi-aventis, Hoechst Marion Roussel, Inc. (HMRI), discovered Fanapt® and completed early clinical work on the compound. In 1996, following a review of its product portfolio, HMRI licensed its rights to the Fanapt® patents and patent applications to Titan Pharmaceuticals, Inc. (Titan) on an exclusive basis. In 1997, soon after it had acquired its rights, Titan sublicensed its rights to Fanapt® on an exclusive basis to Novartis. In June 2004, the Company acquired exclusive worldwide rights to these patents and patent applications as well as certain Novartis patents and patent applications to develop and commercialize Fanapt® through a sublicense agreement with Novartis. In partial consideration for this sublicense, the Company paid Novartis an initial license fee of \$0.5 million and was obligated to make future milestone payments to Novartis of less than \$100.0 million in the aggregate (the majority of which were tied to sales milestones), as well as royalty payments to Novartis at a rate which, as a percentage of net sales, was in the mid-twenties. In November 2007, the Company met a milestone under this sublicense agreement relating to the acceptance of its filing of the NDA for Fanapt® for the treatment of schizophrenia and made a corresponding payment of \$5.0 million to Novartis. As a result of the FDA's approval of the NDA for Fanapt® in May 2009, the Company met an additional milestone under this sublicense agreement which required the Company to make a payment of \$12.0 million to Novartis.

On October 12, 2009, Vanda entered into an amended and restated sublicense agreement with Novartis which amended and restated the June 2004 sublicense agreement with Novartis. Pursuant to the amended and restated sublicense agreement, Novartis has exclusive commercialization rights to all formulations of Fanapt® in the U.S. and Canada. Novartis began selling Fanapt® in the U.S. during the first quarter of 2010. Novartis is responsible for the further clinical development activities in the U.S. and Canada, including the development of a long-acting injectable (or depot) formulation of Fanapt®. Pursuant to the amended and restated sublicense agreement, Vanda received an upfront payment of \$200.0 million and Vanda is eligible for additional payments totaling up to \$265.0 million upon the achievement of certain commercial and development milestones for Fanapt® in the U.S. and Canada. Vanda also receives royalties, which, as a percentage of net sales, are in the low double-digits, on net sales of Fanapt® in the U.S. and Canada. In addition, Vanda is no longer required to make any future milestone payments with respect to sales of Fanapt® or any future royalty payments with respect to sales of Fanapt® in the U.S. and Canada. Vanda retains exclusive rights to Fanapt® outside the U.S. and Canada and Vanda has exclusive rights to use any of Novartis' data for Fanapt® for developing and commercializing Fanapt® outside the U.S. and Canada and Vanda will enter into good faith discussions with Novartis relating to the co-commercialization of Fanapt® outside of the U.S. and Canada or, alternatively, Novartis will receive a royalty on net sales of Fanapt® outside of the U.S. and Canada. Novartis has chosen not to co-commercialize Fanapt® with Vanda in Europe and certain other countries and will instead receive a royalty on net sales in those countries. These include, but are not limited to, the countries in the European Union as well as Switzerland, Norway, Liechtenstein, and Iceland. Vanda has entered into agreements with th

Country Mexico Argentina Partner
Probiomed S.A. de C.V.
Biotoscana Farma S.A.

VANDA PHARMACEUTICALS INC.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (unaudited) - (continued)

Vanda may lose its rights to develop and commercialize Fanapt® outside the U.S. and Canada if it fails to comply with certain requirements in the amended and restated sublicense agreement regarding its financial condition, or if Vanda fails to comply with certain diligence obligations regarding its development or commercialization activities or if Vanda otherwise breaches the amended and restated sublicense agreement and fails to cure such breach. Vanda's rights to develop and commercialize Fanapt® outside the U.S. and Canada may be impaired if it does not cure breaches by Novartis of similar obligations contained its sublicense agreement with Titan for Fanapt®. Vanda is not aware of any such breach by Novartis. In addition, if Novartis breaches the amended and restated sublicense agreement with respect to its commercialization activities in the U.S. or Canada, Vanda may terminate Novartis' commercialization rights in the applicable country and Vanda would no longer receive royalty payments from Novartis in connection with such country in the event of such termination.

Tasimelteon. In February 2004, the Company entered into a license agreement with Bristol-Myers Squibb (BMS) under which the Company received an exclusive worldwide license under certain patents and patent applications, and other licenses to intellectual property, to develop and commercialize tasimelteon. In partial consideration for the license, the Company paid BMS an initial license fee of \$0.5 million. The Company is also obligated to make future milestone payments to BMS of less than \$40.0 million in the aggregate (the majority of which are tied to sales milestones) as well as royalty payments based on the net sales of tasimelteon at a rate which, as a percentage of net sales, is in the low teens. The Company made a milestone payment to BMS of \$1.0 million under this license agreement in 2006 relating to the initiation of its first Phase III clinical trial for tasimelteon. The Company is also obligated under this agreement to pay BMS a percentage of any sublicense fees, upfront payments and milestone and other payments (excluding royalties) that the Company receives from a third party in connection with any sublicensing arrangement, at a rate which is in the mid-twenties. The Company has agreed with BMS in the license agreement for tasimelteon to use commercially reasonable efforts to develop and commercialize tasimelteon and to meet certain milestones in initiating and completing certain clinical work. The license agreement with BMS was amended on April 15, 2010 to, among other things, extend the deadline by which the Company must enter into a development and commercialization agreement with a third party for tasimelteon until the earliest of: (i) the date mutually agreed upon by the Company and BMS following the provision by the Company to BMS of a full written report of the Phase III report); (ii) the date of the acceptance by a regulatory authority of the filing by the Company for marketing authorization for tasimelteon in a major market country following the provision by the Company to BMS of the Phase

If the Company has not entered into such a development and commercialization agreement with respect to certain major market countries by the foregoing deadline, then BMS will have the option to exclusively develop and commercialize tasimelteon on its own in those countries not covered by such an agreement on pre-determined financial terms, including milestone and royalty payments. In addition to the foregoing, pursuant to the April 15, 2010 amendment, Vanda's deadline for filing a NDA for tasimelteon was extended until June 1, 2013.

Either party may terminate the tasimelteon license agreement under certain circumstances, including a material breach of the agreement by the other. In the event that BMS has not exercised its option to reacquire the rights to tasimelteon and the Company terminates the license, or if BMS terminates the license due to the Company's breach, all rights licensed and developed by the Company under this agreement will revert or otherwise be licensed back to BMS on an exclusive basis.

Future license payments. No amounts were recorded as liabilities nor were any contractual obligations relating to the license agreements included in the condensed consolidated financial statements as of September 30, 2011, since the amounts, timing and likelihood of these future payments are unknown and will depend on the successful outcome of future clinical trials, regulatory filings, favorable FDA regulatory approvals, growth in product sales and other factors.

VANDA PHARMACEUTICALS INC.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (unaudited) - (continued)

Research and development and marketing agreements

In the course of its business the Company regularly enters into agreements with clinical organizations to provide services relating to clinical development and clinical manufacturing activities under fee service arrangements. The Company's current agreements for clinical services may be terminated on no more than 60 days notice without incurring additional charges, other than charges for work completed but not paid for through the effective date of termination and other costs incurred by the Company's contractors in closing out work in progress as of the effective date of termination.

11. Income Taxes

The Company recorded a tax benefit of \$0.2 million for the nine months ended September 30, 2011 and a tax provision of \$3.3 million for the nine months ended September 30, 2010. At September 30, 2011, the Company reflected a net deferred tax asset of \$1.8 million associated with the Company's ability to carryback current taxable losses to recover income taxes accrued in 2010. During the nine months ended September 30, 2010, the Company released \$1.6 million of valuation allowance due to the possibility of offsetting the current year tax provision through the carryback of losses generated by the future reversal of temporary differences. The remaining net deferred tax assets at September 30, 2011 and September 30, 2010 were offset by a valuation allowance since realization of any future benefit from deductible temporary differences and net operating losses could not be sufficiently assured.

12. Fair Value Measurements

FASB guidance establishes a three-tier fair value hierarchy, which prioritizes the inputs used in measuring fair value. These tiers include:

- Level 1 defined as observable inputs such as quoted prices in active markets
- · Level 2 defined as inputs other than quoted prices in active markets that are either directly or indirectly observable
- Level 3 defined as unobservable inputs in which little or no market data exists, therefore requiring an entity to develop its own assumptions

Marketable securities classified in Level 1 and 2 at September 30, 2011 and December 31, 2010 include available-for-sale marketable securities. The valuation of Level 1 instruments is determined using a market approach, and is based upon unadjusted quoted prices for identical assets in active markets. The valuation of investments classified in Level 2 also is determined using a market approach based upon quoted prices for similar assets in active markets, or other inputs that are observable for substantially the full term of the financial instrument. Level 2 securities primarily include commercial paper, corporate notes and government agency notes that use as their basis readily observable market parameters.

As of September 30, 2011, the Company held certain assets that are required to be measured at fair value on a recurring basis.

	Fair Value Measurements at Reporting Date Using								
(in thousands)	Quoted Prices in Active Markets for Significant Other Significa Identical Assets Observable Inputs Unobservable September 30, 2011 (Level 1) (Level 2) (Level 3		Active Markets for Identical Assets			Active Markets for Significant Other Identical Assets Observable Inputs			vable Inputs
Description:									
Available-for-sale securities	\$	120,113	\$	55,762	\$	64,371	\$	_	
Total	\$	120,113	\$	55,762	\$	64,371	\$	_	

As of December 31, 2010, the Company held certain assets that are required to be measured at fair value on a recurring basis.

			Fair Value Measurements at Reporting Date Using					
(in thousands)	December 3	Quoted Prices in Active Markets for Identical Assets (Cevel 1)				Unobser	nificant vable Inputs evel 3)	
Description:		<u> </u>						
Available-for-sale securities	\$ 15	5,478	\$	45,455	\$	110,023	\$	_
Total	\$ 15	5,478	\$	45,455	\$	110,023	\$	_

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations.

Various statements in this report are "forward-looking statements" within the meaning of the Private Securities Litigation Reform Act of 1995. Forward-looking statements may appear throughout this report. Words such as, but not limited to, "believe," "expect," "anticipate," "estimate," "intend," "plan," "targets," "likely," "will," "would," and "could," or the negative of these terms and similar expressions or words, identify forward-looking statements. Forward-looking statements are based upon current expectations that involve risks, changes in circumstances, assumptions and uncertainties. Important factors that could cause actual results to differ materially from those reflected in our forward-looking statements include, among others:

- · the extent and effectiveness of the development, sales and marketing and distribution support Fanapt® receives;
- our ability to successfully commercialize Fanapt® outside of the U.S. and Canada;
- delays in the completion of our clinical trials;
- a failure of our products, product candidates or partnered products to be demonstrably safe and effective;
- · our failure to obtain regulatory approval for our products or product candidates or to comply with ongoing regulatory requirements;
- a lack of acceptance of our products, product candidates or partnered products in the marketplace, or a failure to become or remain profitable;
- · our expectations regarding trends with respect to our costs and expenses;
- our inability to obtain the capital necessary to fund our research and development activities;
- our failure to identify or obtain rights to new products or product candidates;
- our failure to develop or obtain sales, marketing and distribution resources and expertise or to otherwise manage our growth;
- · limitations on our ability to utilize some or all of our prior net operating losses and research and development credits;
- a loss of any of our key scientists or management personnel;
- losses incurred from product liability claims made against us; and
- a loss of rights to develop and commercialize our products or product candidates under our license and sublicense agreements.

All written and verbal forward-looking statements attributable to us or any person acting on our behalf are expressly qualified in their entirety by the cautionary statements contained or referred to in this section. We caution investors not to rely too heavily on the forward-looking statements we make or that are made on our behalf. We undertake no obligation, and specifically decline any obligation, to update or revise publicly any forward-looking statements, whether as a result of new information, future events or otherwise.

We encourage you to read the discussion and analysis of our financial condition and our condensed consolidated financial statements contained in this quarterly report on Form 10-Q. We also encourage you to read Item 1A of Part II of this quarterly report on Form 10-Q entitled "Risk Factors" and Item 1A of Part I of our Annual Report on Form 10-K for the fiscal year ended December 31, 2010 which contain a more complete discussion of the risks and uncertainties associated with our business. In addition to the risks described above and in Item 1A of Part II of this report and Item 1A of Part I of our Annual Report on Form 10-K for the fiscal year ended December 31, 2010, other unknown or unpredictable factors also could affect our results. Therefore, the information in this report should be read together with other reports and documents that we file with the Securities and Exchange Commission (SEC) from time to time, including Forms 10-Q, 8-K and 10-K, which may supplement, modify, supersede or update those risk factors. There can be no assurance that the actual results or developments anticipated by us will be realized or, even if substantially realized, that they will have the expected consequences to, or effects on, us. Therefore, no assurance can be given that the outcomes stated in such forward-looking statements and estimates will be achieved.

Overview

We are a biopharmaceutical company focused on the development and commercialization of products for the treatment of central nervous system disorders. We believe that each of our products and partnered products will address a large market with significant unmet medical needs by offering advantages over currently available therapies. Our product portfolio includes:

Fanapt® (iloperidone) We have developed Fanapt®, and will continue to develop it outside the U.S. and Canada, to treat schizophrenia. On October 12, 2009, we entered into an amended and restated sublicense agreement with Novartis. We had originally entered into a sublicense agreement with Novartis on June 4, 2004 pursuant to which we obtained certain worldwide exclusive licenses from Novartis relating to Fanapt®. Pursuant to the amended and restated sublicense agreement, Novartis has exclusive commercialization rights to all formulations of Fanapt® in the U.S. and Canada. On January 11, 2010, Novartis launched Fanapt® in the U.S. Novartis is responsible for the further clinical development activities in the U.S. and Canada, including the development of a long-acting injectable (or depot) formulation of Fanapt®. Pursuant to the amended and restated sublicense agreement, we received an upfront payment of \$200.0 million and are eligible for additional payments totaling up to \$265.0 million upon the achievement of certain commercial and development milestones for Fanapt® in the U.S. and Canada. We also receive royalties, which, as a percentage of net sales, are in the low doubledigits, on net sales of Fanapt® in the U.S. and Canada. In addition, we are no longer required to make any future milestone payments with respect to sales of Fanapt® or any future royalty payments with respect to sales of Fanapt® in the U.S. and Canada. We retain exclusive rights to Fanapt® outside the U.S. and Canada and we have exclusive rights to use any of Novartis' data for Fanapt® for developing and commercializing Fanapt® outside the U.S. and Canada. At Novartis' option, we will enter into good faith discussions with Novartis relating to the co-commercialization of Fanapt® outside of the U.S. and Canada or, alternatively, Novartis will receive a royalty on net sales of Fanapt® outside of the U.S. and Canada. Novartis has chosen not to co-commercialize Fanapt® with us in Europe and certain other countries and will instead receive a royalty on net sales in those countries. These include, but are not limited to, the countries in the European Union as well as Switzerland, Norway, Liechtenstein, and Iceland. We continue to explore the regulatory path and commercial opportunity for Fanapt® oral formulation outside of the U.S. and Canada. On November 1, 2010, the Therapeutic Goods Administration of Australia's Department of Health and Ageing, accepted for evaluation our application for marketing approval for the Fanapt® oral formulation. On July 22, 2011, the European Medicines Agency (EMA) notified us that it had accepted for evaluation the Marketing Authorization Application (MAA) for oral iloperidone tablets. We have entered into agreements with the following partners for the commercialization of Fanapt® in the countries set forth below:

CountryPartnerMexicoProbiomed S.A. de C.V.ArgentinaBiotoscana Farma S.A.

For the nine months ended September 30, 2011 we incurred \$1.9 million in research and development costs directly attributable to our development of Fanapt®. As a result of the FDA's approval of the new drug application (NDA) for Fanapt® in May 2009, we met a milestone under the original sublicense agreement which required us to make a payment of \$12.0 million to Novartis. The \$12.0 million was capitalized and will be amortized over the remaining life of the U.S. patent for Fanapt®, which we expect to last until May 15, 2017. This includes the Hatch-Waxman extension that extends patent protection for drug compounds for a period of up to five years to compensate for time spent in development and a six-month pediatric term extension.

Tasimelteon. Tasimelteon is an oral compound in development for the treatment of sleep and mood disorders, including Circadian Rhythm Sleep Disorders (CRSD). The compound binds selectively to the brain's melatonin receptors, which are thought to govern the body's natural sleep/wake cycle. Compounds that bind selectively to these receptors are thought to be able to help treat sleep disorders, and additionally are believed to offer potential benefits in mood disorders. We announced positive top-line results from our Phase III trial of tasimelteon in transient insomnia in November 2006. In June 2008, we announced positive top-line results from the Phase III trial of tasimelteon in chronic primary insomnia. The trial was a randomized, double-blind, and placebo-controlled study with 324 patients. The trial measured time to fall asleep and sleep maintenance, as well as next-day performance. On January 19, 2010, the FDA granted orphan drug designation status for tasimelteon in a specific CRSD, Non-24-Hour Sleep-Wake Disorder (N24HSWD) in blind individuals without light perception. The FDA grants orphan drug designation to drugs that may provide significant therapeutic advantage over existing treatments and target conditions affecting 200,000 or fewer U.S. patients per year. Orphan drug designation provides potential financial and regulatory incentives, including study design assistance, tax credits, waiver of FDA user fees, and up to seven years of market exclusivity upon marketing approval. On February 23, 2011, the European Commission (EC) designated tasimelteon as an orphan medicinal product for the same indication. We initiated four clinical trials to pursue FDA approval of tasimelteon for the treatment of N24HSWD in blind individuals without light perception. Two of the clinical trials were initiated in the third quarter of 2010, the third was initiated in the third quarter of 2011 and the fourth was initiated in October 2011. The first clinical trial (3201) is a randomized, double-blind, placebo-controlled study with a planned enrollment of approximately 100 patients with N24HSWD. The trial has a six month treatment period and includes measures of both nighttime and daytime sleep, as well as laboratory measures of the synchronization between the internal body clock and the 24-hour environmental light/dark cycle. The second clinical trial (3202) is a one-year safety study of tasimelteon for the treatment of N24HSWD. This trial is an open-label safety study with a planned enrollment of up to 140 patients with N24HSWD. The third clinical trial (3203) is a placebo-controlled, randomized withdrawal study to examine the maintenance of effect of tasimelteon for the treatment of N24HSWD. Patients will be followed for 8 weeks during which nighttime and daytime sleep as well as synchronization of their internal body clock to the 24-hour day will continue to be evaluated. The fourth clinical trial (3204) is a two year open-label, multicenter, study in blind subjects with N24HSWD to assess the safety of tasimelteon. We plan to conduct these clinical trials over the next one to two years to support the use of tasimelteon as a circadian regulator and the submission of a NDA to the FDA and a marketing authorization application to the EMA. In addition, we initiated a Phase IIb/III clinical trial to study the efficacy of tasimelteon for the treatment of Major Depressive Disorder (MDD) in the third quarter of 2011. The clinical trial is a randomized, double-blind, placebo-controlled study with planned enrollment of approximately 500 patients with MDD. The trial has an eight-week treatment period, followed by an optional one-year open-label extension, and includes measures of depression and anxiety symptoms, nighttime and daytime sleep, as well as laboratory measures of the internal body clock. Given the range of potential indications for tasimelteon, we may pursue one or more partnerships for the development and commercialization of tasimelteon worldwide. For the nine months ended September 30, 2011, we incurred \$15.7 million in direct research and development costs directly attributable to our development of tasimelteon.

Since we began our operations in March 2003, we have devoted substantially all of our resources to the in-licensing and clinical development of our compounds. Our ability to generate revenue and achieve profitability largely depends on Novartis' ability to successfully commercialize Fanapt® in the U.S. and to successfully develop and commercialize Fanapt® in Canada and upon our ability, alone or with others, to complete the development of our products or product candidates, and to obtain the regulatory approvals for and manufacture, market and sell our products and product candidates. The results of our operations will vary significantly from year-to-year and quarter-to-quarter and depend on a number of factors, including risks related to our business, risks related to our industry, and other risks which are detailed in Item 1A of Part I of this quarterly report on Form 10-Q, entitled "Risk Factors" and in Item 1A of Part I of our Annual Report on Form 10-K for the year ended December 31, 2010.

Revenues

Our revenues are derived primarily from our amended and restated sublicense agreement with Novartis and include an up-front payment, product sales and future milestone and royalty payments. Revenue is considered both realizable and earned when each one of the following four conditions is met: (1) persuasive evidence of an arrangement exists, (2) the arrangement fee is fixed or determinable, (3) delivery or performance has occurred and (4) collectability is reasonably assured. Revenue from the \$200.0 million upfront payment will be recognized ratably on a straight-line basis from the date the amended and restated sublicense agreement became effective (November 27, 2009) through the expected life of the U.S. patent for Fanapt®, which we expect to last until May 15, 2017. This includes the Hatch-Waxman extension that extends patent protection for drug compounds for a period of up to five years to compensate for time spent in development and a six-month pediatric term extension. Revenue related to product sales is recognized upon delivery to Novartis. We recognize revenue from Fanapt® royalties and commercial and development milestones from Novartis when realizable and earned.

Research and development expenses

Our research and development expenses consist primarily of fees paid to third-party professional service providers in connection with the services they provide for our clinical trials, costs of contract manufacturing services, milestone license fees, costs of materials used in clinical trials and research and development, costs for regulatory consultants and filings, depreciation of capital resources used to develop our products, all related facilities costs, and salaries, benefits and stock-based compensation expenses related to our research and development personnel. We expense research and development costs as they are incurred for compounds in development stage, including certain payments made under our license agreements prior to FDA approval. Prior to FDA approval, all Fanapt® manufacturing-related and milestone costs were included in research and development expenses. Subsequent to FDA approval of Fanapt®, manufacturing and milestone costs related to this product are being capitalized. Costs related to the acquisition of intellectual property have been expensed as incurred since the underlying technology associated with these acquisitions were made in connection with the Company's research and development efforts and have no alternative future use. Milestone payments are accrued in accordance with the FASB guidance on accounting for contingencies which states that milestone payments be accrued when it is deemed probable that the milestone event will be achieved. We believe that significant investment in product development is a competitive necessity and plan to continue these investments in order to realize the potential of our products and product candidates and pharmacogenetics and pharmacogenomics expertise. For the nine months ended September 30, 2011, we incurred research and development expenses in the aggregate of \$18.4 million, including stock-based compensation expenses of \$1.9 million. We expect our research and development expenses to increase as we continue to develop our products and product

The following table summarizes our product development initiatives for the nine months ended September 30, 2011 and 2010. Included in this table are the research and development expenses recognized in connection with the clinical development of Fanapt® and tasimelteon.

	Three Months Ended			Nine Months Ended			
	ember 30, 2011	September 30, 2010		Sep	September 30, 2011		eptember 30, 2010
(in thousands)	 					_	
Direct project costs(1)							
Fanapt®	\$ 890	\$	992	\$	1,851	\$	2,158
Tasimelteon	6,893		2,723		15,709		5,349
Total direct project costs	7,783		3,715		17,560	_	7,507
Indirect project costs(1)							
Facility	280		150		591		459
Depreciation	78		42		147		144
Other indirect overhead	 33		165		142	_	406
Total indirect project costs	391		357		880		1,009
Total research and development expenses	\$ 8,174	\$	4,072	\$	18,440	\$	8,516

⁽¹⁾ Many of our research and development costs are not attributable to any individual project because we share resources across several development projects. We record direct costs, including personnel costs and related benefits and stock-based compensation, on a project-by-project basis. We record indirect costs that support a number of our research and development activities in the aggregate.

General and administrative expenses

General and administrative expenses consist primarily of salaries and other related costs for personnel, including stock-based compensation, serving executive, finance, accounting, information technology, marketing and human resource functions. Other costs include facility costs not otherwise included in research and development expenses, insurance costs and professional fees for legal, accounting and other professional services. General and administrative expenses also include third party expenses incurred to support business development, marketing and other business activities related to Fanapt®. For the nine months ended September 30, 2011, we incurred general and administrative expenses in the aggregate of \$8.1 million, including stock-based compensation expenses of \$2.3 million.

Interest income

Interest income consists of interest earned on our cash and cash equivalents, marketable securities and restricted cash.

Critical Accounting Policies

The preparation of our condensed consolidated financial statements requires us to make estimates and assumptions that affect the reported amounts of assets and liabilities and the disclosure of contingent assets and liabilities at the date of our financial statements, as well as the reported revenues and expenses during the reported periods. We base our estimates on historical experience and on various other factors that we believe are reasonable under the circumstances, the results of which form the basis for making judgments about the carrying value of assets and liabilities that are not apparent from other sources. Actual results may differ from these estimates under different assumptions or conditions.

Our significant accounting policies are described in the notes to our audited consolidated financial statements for the year ended December 31, 2010 included in our annual report on Form 10-K. However, we believe that the following critical accounting policies are important to understanding and evaluating our reported financial results, and we have accordingly included them in this quarterly report on Form 10-Q.

Accrued expenses

As part of the process of preparing our financial statements, we are required to estimate accrued expenses. The estimation of accrued expenses involves identifying services that have been performed on our behalf, and then estimating the level of service performed and the associated cost incurred for such services as of each balance sheet date in the financial statements. Accrued expenses include professional service fees, such as lawyers and accountants, contract service fees, such as those under contracts with clinical monitors, data management organizations and investigators in conjunction with clinical trials, fees to contract manufacturers in conjunction with the production of clinical materials, and fees for marketing and other commercialization activities. Pursuant to our assessment of the services that have been performed on clinical trials and other contracts, we recognize these expenses as the services are provided. Our assessments include, but are not limited to: (1) an evaluation by the project manager of the work that has been completed during the period, (2) measurement of progress prepared internally and/or provided by the third-party service provider, (3) analyses of data that justify the progress and (4) management's judgment. In the event that we do not identify certain costs that have begun to be incurred or we under- or over-estimate the level of services performed or the costs of such services, our reported expenses for such period would be too low or too high.

Revenue Recognition

Our revenues are derived primarily from our amended and restated sublicense agreement with Novartis and include an up-front payment, product revenue and future milestone and royalty revenues. Revenue will be recognized ratably from the date the amended and restated sublicense agreement became effective (November 27, 2009) through the expected life of the U.S. patent for Fanapt®, which we expect to last until May 15, 2017. This includes the Hatch-Waxman extension that extends patent protection for drug compounds for a period of up to five years to compensate for time spent in development and a six-month pediatric term extension. Revenue related to product sales is recognized upon delivery to Novartis. We recognize revenue from Fanapt® royalties and commercial and development milestones from Novartis when realizable and earned.

Stock-based compensation

We currently use the Black-Scholes-Merton option pricing model to determine the fair value of stock options. The determination of the fair value of stock options on the date of grant using an option pricing model is affected by our stock price as well as assumptions regarding a number of complex and subjective variables. These variables include the expected stock price volatility over the expected term of the awards, actual and projected employee stock option exercise behaviors, risk-free interest rate and expected dividends. Expected volatility rates are based on our historical volatility of our publicly traded common stock blended with the historical volatility of the common stock of comparable entities. The expected term of options granted is based on the transition approach provided by FASB guidance as the options meet the "plain vanilla" criteria required by this method. The risk-free interest rates are based on the U.S. Treasury yield for a period consistent with the expected term of the option in effect at the time of the grant. We have not paid cash dividends to our stockholders since our inception and do not plan to pay any dividends in the foreseeable future. The stock-based compensation expense for a period is also affected by expected forfeiture rate for the respective option grants. If our estimates of the fair value of these equity instruments or expected forfeitures are too high or too low, it would have the effect of overstating or understating expenses.

Total employee stock-based compensation expense related to all of our stock-based awards during the nine months ended September 30, 2011 and 2010 was comprised of the following:

	 Three Months Ended				Nine Months Ended			
			mber 30, 010	September 30, 2011		80, Septen 20		
(in thousands)	 	_						
Research and development	\$ 558	\$	429	\$	1,896	\$	1,983	
General and administrative	 704		368		2,278		1,547	
Stock-based compensation expense	\$ 1,262	\$	797	\$	4,174	\$	3,530	

Income taxes

On a periodic basis, we evaluate the realizability of our deferred tax assets and liabilities and will adjust such amounts in light of changing facts and circumstances, including but not limited to future projections of taxable income, the reversal of deferred tax liabilities, tax legislation, rulings by relevant tax authorities and tax planning strategies. Settlement of filing positions that may be challenged by tax authorities could impact our income taxes in the year of resolution.

In assessing the realizability of deferred tax assets, we consider whether it is more likely than not that some portion or all of the deferred tax assets will not be realized. The ultimate realization of deferred tax assets is dependent upon the generation of future taxable income during the period in which those temporary differences becomes deductible or the NOLs and credit carryforwards can be utilized. When considering the reversal of the valuation allowance, we consider the level of past and future taxable income, the reversal of deferred tax liabilities, the utilization of the carryforwards and other factors. Revisions to the estimated net realizable value of the deferred tax asset could cause our provision for income taxes to vary significantly from period to period.

Recent Accounting Pronouncements

In June 2011, the FASB issued an Accounting Standards Update which eliminates the option to report other comprehensive income and its components in the statement of changes in stockholders' equity. It requires an entity to present total comprehensive income, which includes the components of net income and the components of other comprehensive income either in a single continuous statement or in two separate but consecutive statements. This pronouncement is effective for financial statements issued for annual and interim periods within the first annual period beginning after December 15, 2011. We believe the adoption of this pronouncement will not have a material impact on our financial position or results of operations.

Results of Operations

We have a limited history of operations. We anticipate that our results of operations will fluctuate for the foreseeable future due to several factors, including any possible payments made or received pursuant to licensing or collaboration agreements, progress of our research and development efforts, the timing and outcome of clinical trials and related possible regulatory approvals and our and our partners' ability to successfully commercialize our products, product candidates and partnered products. Our limited operating history makes predictions of future operations difficult or impossible. Since our inception, we have incurred significant losses. As of September 30, 2011, we had a deficit accumulated of \$257.9 million.

Three months ended September 30, 2011 compared to three months ended September 30, 2010

Revenues. Revenues were \$8.0 million for the three months ended September 30, 2011, compared to revenues of \$7.2 for the three months ended September 30, 2010. Revenues for the three months ended September 30, 2011 included \$6.8 million recognized from Novartis related to straight-line recognition of up-front license fees and \$1.2 million in royalty revenue based on third quarter 2011 sales of Fanapt®. Novartis launched Fanapt® commercially in the U.S. in January 2010.

Intangible asset amortization. Intangible asset amortization was \$0.4 million for both the three months ended September 30, 2011 and the three months ended September 30, 2010. Intangible amortization relates to the capitalized intangible asset related to the \$12.0 million payment to Novartis in May 2009.

Research and development expenses. Research and development expenses increased by \$4.1 million, or 100.7%, to \$8.2 million for the three months ended September 30, 2011 compared to \$4.1 million for the three months ended September 30, 2010.

The following table discloses the components of research and development expenses reflecting all of our project expenses for the three months ended September 30, 2011 and 2010:

		Three Months Ended		
	Sep	tember 30, 2011	Sept	ember 30, 2010
(in thousands)				
Direct project costs:				
Clinical trials	\$	4,698	\$	1,210
Contract research and development, consulting, materials and other				
direct costs		1,460		1,343
Salaries, benefits and related costs		1,067		733
Stock-based compensation		558		429
Total direct costs		7,783		3,715
Indirect project costs		391		357
Total	\$	8,174	\$	4,072

Direct costs increased \$4.1 million for the three months ended September 30, 2011 compared to the three months ended September 30, 2010 as a result of increases in clinical trial costs, contract research and development, consulting, materials and other direct costs, salaries, benefits and related costs and stock based compensation. Clinical trials costs increased by \$3.5 million for the three months ended September 30, 2011 relative to the three months ended September 30, 2010, primarily due to costs related to the tasimelteon trials for the treatment of N24HSWD in blind individuals without light perception and the tasimelteon trial for the treatment of MDD. Salaries, benefits and related costs increased by \$0.3 million for the three months ended September 30, 2011 relative to the three months ended September 30, 2010, as a result of employee hiring to support the tasimelteon trials for the treatment of N24HSWD in blind individuals without light perception and the tasimelteon trial for the treatment of MDD.

General and administrative expenses. General and administrative expenses increased by \$0.7 million, or 32.0%, to \$2.7 million for the three months ended September 30, 2011 from \$2.1 million for the three months ended September 30, 2010.

The following table discloses the components of our general and administrative expenses for the three months ended September 30, 2011 and 2010:

	Three Months Ended			
(in thousands)		September 30, 2011		ember 30, 2010
Salaries, benefits and related costs	\$	433	\$	334
Stock-based compensation		704		368
Marketing, legal, accounting and other professional expenses		918		732
Other expenses		656		620
Total	\$	2,711	\$	2,054

Stock-based compensation expense increased by \$0.3 million for the three months ended September 30, 2011 compared to the three months ended September 30, 2010, as a result of the cancellation of unvested options due to an executive departure during the three months ending September 30, 2010. Marketing, legal, accounting and other professional costs increased by \$0.2 million for the three months ended September 30, 2011 compared to the three months ended September 30, 2010 due to a increased legal expenses associated with the commercialization of Fanapt® outside the U.S. and Canada.

Interest income. Interest income decreased by \$0.1 million to \$0.1 million for the three months ended September 30, 2011 from \$0.2 million for the three months ended September 30, 2010.

Nine months ended September 30, 2011 compared to nine months ended September 30, 2010

Revenues. Revenues were \$22.9 million for the nine months ended September 30, 2011, compared to revenues of \$28.0 for the nine months ended September 30, 2010. Revenues for the nine months ended September 30, 2011 included \$20.0 million recognized from Novartis related to straight-line recognition of up-front license fees and \$2.9 million in royalty revenue based on sales of Fanapt® in the nine months ended September 30, 2011. Revenues for the nine months ended September 30, 2010 included \$20.0 million recognized from Novartis related to the straight-line recognition of up-front license fees, \$2.6 million in royalty revenue based on sales of Fanapt® in the nine months ended September 30, 2010 and \$5.3 million for Fanapt® product sales to Novartis. Novartis launched Fanapt® commercially in the U.S. in January 2010.

Intangible asset amortization. Intangible asset amortization was \$1.1 million for both the nine months ended September 30, 2011 and the nine months ended September 30, 2010. Intangible amortization relates to the capitalized intangible asset related to the \$12.0 million payment to Novartis in May 2009.

Research and development expenses. Research and development expenses increased by \$9.9 million, or 116.5%, to \$18.4 million for the nine months ended September 30, 2011 compared to \$8.5 million for the nine months ended September 30, 2010.

The following table discloses the components of research and development expenses reflecting all of our project expenses for the nine months ended September 30, 2011 and 2010:

	Nine Months Ended			
(in thousands)	September 30, 2011			ember 30, 2010
Direct project costs:				
Clinical trials	\$	9,086	\$	1,496
Contract research and development, consulting, materials and other				
direct costs		3,635		1,916
Salaries, benefits and related costs		2,943		2,112
Stock-based compensation		1,896		1,983
Total direct costs		17,560		7,507
Indirect project costs		880		1,009
Total	\$	18,440	\$	8,516

Direct costs increased \$10.1 million for the nine months ended September 30, 2011 compared to the nine months ended September 30, 2010 as a result of increases in clinical trial costs, contract research and development, consulting, materials and other direct costs and salaries, benefits and related costs partially offset by lower stock based compensation. Clinical trials costs increased by \$7.6 million for the nine months ended September 30, 2011 relative to the nine months ended September 30, 2010, primarily due to costs related to the tasimelteon trials for the treatment of N24HSWD in blind individuals without light perception and the tasimelteon trial for the treatment of MDD. Contract research and development, consulting, materials and other direct costs increased \$1.7 million for the nine months ended September 30, 2011 relative to the nine months ended September 30, 2010, primarily due to costs related to those same tasimelteon trials.

General and administrative expenses. General and administrative expenses increased by \$0.8 million, or 10.2%, to \$8.1 million for the nine months ended September 30, 2011 from \$7.4 million for the nine months ended September 30, 2010.

The following table discloses the components of our general and administrative expenses for the nine months ended September 30, 2011 and 2010:

	Nine Months Ended				
(in thousands)	September 30, 2011	September 30, 2010			
Salaries, benefits and related costs	\$ 1,436	\$ 1,346			
Stock-based compensation	2,278	1,547			
Marketing, legal, accounting and other professional expenses	2,530	2,616			
Other expenses	1,897	1,876			
Total	\$ 8,141	\$ 7,385			

Stock-based compensation expense increased by \$0.7 million for the nine months ended September 30, 2011 compared to the nine months ended September 30, 2010, as a result of the cancellation of unvested options due to executive departures during the nine months ending September 30, 2010.

Interest income. Interest income increased by \$0.1 million to \$0.4 million for the nine months ended September 30, 2011 from \$0.3 million for the nine months ended September 30, 2010.

Liquidity and Capital Resources

As of September 30, 2011, our total cash and cash equivalents and marketable securities were \$180.5 million compared to \$198.0 million at December 31, 2010. Our cash and cash equivalents are deposits in operating accounts and highly liquid investments with an original maturity of 90 days or less at date of purchase and consist of time deposits, investments in money market funds with commercial banks and financial institutions, and commercial paper of high-quality corporate issuers. Our marketable securities consist of investments in government sponsored enterprises and commercial paper. As of September 30, 2011, we also held a non-current deposit of \$0.4 million that is used to collateralize a letter of credit issued for our office lease in Rockville, Maryland which expires in 2016, a non-current deposit of \$0.1 million related to a letter of credit issued for the Maryland Board of Pharmacy, and a non-current deposit of \$0.5 million that is used to collateralize a letter of credit issued for our office lease in Washington, DC which expires in 2023.

As of September 30, 2011, we maintained all of our cash and cash equivalents in three financial institutions. Deposits held with these institutions may exceed the amount of insurance provided on such deposits, but we do not anticipate any losses with respect to such deposits.

We entered into an amended and restated sublicense agreement in 2009 with Novartis to commercialize Fanapt® in the U.S. and Canada. Novartis is responsible for the further clinical development activities in the U.S. and Canada, including the development of a long-acting injectable (or depot) formulation of Fanapt®. Pursuant to the amended and restated sublicense agreement, we received an upfront payment of \$200.0 million, and are eligible for additional payments totaling up to \$265.0 million upon the achievement of certain commercial and development milestones for Fanapt® in the U.S. and Canada. We will recognize the \$200.0 million upfront payment ratably from the date the amended and restated sublicense agreement became effective (November 27, 2009) through the expected life of the U.S. patent for Fanapt®, which we expect to last until May 15, 2017. This includes the Hatch-Waxman extension that provides patent protection for drug compounds for a period of up to five years to compensate for time spent in development and a six-month pediatric term extension. We also receive royalties, which, as a percentage of net sales, are in the low double digits, on net sales of Fanapt® in the U.S. and Canada. During the nine months ended September 30, 2011, we recorded \$20.0 million in licensing revenue. Since the launch of Fanapt®, we have recognized product revenue of \$5.3 million from product sold to Novartis and \$6.0 million in royalty revenue. We recognize product revenue on the sale of the existing Fanapt® product to Novartis upon delivery to Novartis and royalty revenue when realizable and earned. Other than participation in the Joint Steering Committee established following the effective date of the amended and restated sublicense agreement with Novartis, we have no control over the progress of Novartis' commercial plans. We cannot forecast with any degree of certainty the achievement of milestones and royalties under this agreement.

We expect to continue to incur substantial expenses relating to our research and development efforts, as we focus on clinical trials and manufacturing required for the development of our active product candidates. We initiated four clinical trials to pursue FDA approval of tasimelteon for the treatment of N24HSWD in blind individuals without light perception. Two of the clinical trials were initiated in the third quarter of 2010, the third was initiated in the third quarter of 2011 and the fourth was initiated in October 2011. In addition, we initiated a Phase IIb/III clinical trial to study the efficacy of tasimelteon for the treatment of MDD in the third quarter of 2011. The duration and cost of clinical trials are a function of numerous factors such as the number of patients to be enrolled in the trial, the amount of time it takes to enroll them, the length of time they must be treated and observed, and the number of clinical sites and countries for the trial. In addition, orphan clinical trials create an additional challenge due to the limited number of available patients afflicted with the disease.

We must receive regulatory approval to launch any of our products commercially. In order to receive such approval, the appropriate regulatory agency must conclude that our clinical data establish safety and efficacy and that our products and the manufacturing facilities meet all applicable regulatory requirements. We cannot be certain that we will establish sufficient safety and efficacy data to receive regulatory approval for any of our drugs or that our drugs and the manufacturing facilities will meet all applicable regulatory requirements.

Because of the uncertainties discussed above, the costs to advance our research and development projects are difficult to estimate and may vary significantly. We expect that our existing funds, primarily consisting of the upfront payment received under the Novartis contract and investment income will be sufficient to fund our planned operations. Our future capital requirements and the adequacy of our available funds will depend on many factors, primarily including the scope and costs of our clinical development programs, the scope and costs of our manufacturing and process development activities, the magnitude of our discovery and preclinical development programs and the level of our pre-commercial launch activities. There can be no assurance that any additional financing required in the future will be available on acceptable terms, if at all.

Cash Flow

The following table summarizes our cash flows for the nine months ended September 30, 2011 and 2010:

	Nine Months Ended		
(in thousands)	September 30, 2011	September 30, 2010	
Net cash provided by (used in)			
Operating activities	\$ (16,053)	\$ (5,802)	
Investing activities	33,815	(99,462)	
Financing activities	5	2,437	
Net change in cash and cash equivalents	\$ 17,767	\$ (102,827)	

Net cash used in operations was \$16.1 million and \$5.8 million for the nine months ended September 30, 2011 and 2010. The increase is a result of additional research and development expenses for tasimelteon. For the nine months ended September 30, 2011, adjustments to reconcile the net loss to net cash used in operating activities included non-cash charges for depreciation and amortization of \$2.2 million and stock-based compensation of \$4.2 million, increases in prepaid expenses and other assets and accounts receivable, accounts payable, accrued liabilities and other liabilities of \$2.1 million, decreases in accrued income taxes of \$0.2 million, and decreases in deferred revenue of \$20.0 million. Net cash provided by investing activities for the nine months ended September 30, 2011 was \$33.8 million and consisted of net maturities of marketable securities of \$34.6 million, purchases of property and equipment of \$0.2 million and an increase in restricted cash of \$0.6 million.

Effects of Inflation

Inflation does not have a material impact on our results of operations.

Off-balance sheet arrangements

We have no off-balance sheet arrangements, as defined in Item 303(a)(4) of the Securities and Exchange Commission's Regulation S-K.

Contractual Obligations and Commitments

The following table summarizes our long-term contractual cash obligations as of September 30, 2011:

		Cash payments due by period							
		Octo	ober to					After	
(in thousands)	Total	Decem	ber 2011	2012	2013	2014	2015	2015	
Operating leases	\$15,202	\$	181	\$ 749	\$1,547	\$1,847	\$1,897	\$8,981	
Consulting fees	3,600	-	900	2,700					
Total	\$18,802	\$	1,081	\$3,449	\$1,547	\$1,847	\$1,897	\$8,981	

Operating leases

Our commitments related to operating leases shown above consist of payments relating to real estate leases for our current headquarters located in Rockville, Maryland, which expires in 2016, and our future headquarters located in Washington, D.C., which expires in 2023. On July 25, 2011, we entered into a lease with Square 54 Office Owner LLC (the Landlord) for our future headquarters, consisting of 21,400 square feet at 2200 Pennsylvania Avenue, N.W. in Washington, D.C. (the Lease). Under the Lease, which will have an 11 year term commencing on April 1, 2012, we will pay \$1.6 million in annual rent over the term of the Lease; however, rent will be abated for the first 12 months. The Landlord will provide us with an allowance of \$1.9 million for construction of the premises to our specifications. Subject to the prior rights of other tenants in the building, we will have the right to renew the Lease for five years following the expiration of its original term. We will also have the right to sublease or assign all or a portion of the premises, subject to standard conditions. The Lease may be terminated early by us or the Landlord upon certain conditions. We paid a security deposit of \$0.5 million upon execution of the Lease. We will likely incur costs of between \$1.0 million to \$1.5 million in connection with an early termination of the lease for our current headquarters in Rockville, Maryland. These costs include a termination fee and other lease exit costs. As of September 30, 2011, we had not incurred any of these costs. These costs are not included in the table above; however, in the event that we exit the lease for our current headquarters in Rockville, Maryland, our contractual cash obligations included in the table above would be reduced by \$2.4 million between 2013 and 2016. In the event that we do not sublease our current headquarters, we expect to have completed the exit of the lease by June 30, 2013.

Consulting fees

We have engaged a regulatory consultant to assist in our efforts to prepare, file and obtain FDA approval of a NDA for tasimelteon. During the initial 15-month term of the engagement, we are obligated to pay consulting fees in the aggregate amount of up to \$3.6 million, of which \$0.9 million will be expensed in the fourth quarter of 2011, and the remainder of which will be expensed between January 1, 2012 and December 31, 2012. As part of this engagement, and subject to certain conditions, we will be obligated to make milestone payments in the aggregate amount of \$2.8 million upon the achievement of certain milestones, including \$2.0 million in the event that a tasimelteon NDA is approved by the FDA. In addition to these fees and milestone payments, we are obligated to reimburse the consultant for its ordinary and necessary business expenses incurred in connection with its engagement. We may terminate the engagement at any time upon prior notice; however, subject to certain conditions, we will remain obligated to make some or all of the milestone payments if the milestones are achieved following such termination.

Clinical research organization contracts and other contracts

Other contracts. We have entered into agreements for tasimelteon with clinical supply manufacturing organizations and other outside contractors who will be responsible for additional services supporting our ongoing clinical development processes. These contractual obligations are not reflected in the table above because we may terminate them on no more than 60 days notice without incurring additional charges (other than charges for work completed but not paid for through the effective date of termination and other costs incurred by our contractors in closing out work in progress as of the effective date of termination).

License agreements. In February 2004 and June 2004, we entered into separate licensing agreements with BMS and Novartis, respectively, for the exclusive rights to develop and commercialize tasimelteon and Fanapt®. On October 12, 2009, we entered into an amended and restated sublicense agreement with Novartis. We are obligated to make (in the case of tasimelteon and, in the case of Fanapt® in the U.S. and Canada, are entitled to receive certain royalties) payments under the conditions in the agreements upon the achievement of specified clinical, regulatory and commercial milestones. If the products are successfully commercialized we will be required to pay certain royalties (and in the case of Fanapt® in the U.S. and Canada, will be entitled to receive) based on net sales for each of the licensed products. See the notes to the consolidated financial statements included with this report for a more detailed description of these license agreements.

As a result of the successful commencement of the Phase III clinical study of tasimelteon in March 2006, we met the first milestone specified in our licensing agreement with BMS and subsequently paid a license fee of \$1.0 million.

As a result of the acceptance by the FDA of the NDA for Fanapt® in October 2007, we met a milestone under our original sublicense agreement with Novartis and subsequently paid a \$5.0 million fee. As a result of the FDA's approval of the NDA for Fanapt® in May 2009, we met an additional milestone under the original sublicense agreement with Novartis which required us to make a payment of \$12.0 million to Novartis. The \$12.0 million was capitalized and will be amortized over the remaining life of the U.S. patent for Fanapt®, which we expect to last until May 15, 2017. This includes the Hatch-Waxman extension that provides patent protection for drug compounds for a period of up to five years to compensate for time spent in development and a six-month pediatric term extension. This term is the Company's best estimate of the life of the patent; if, however, the Hatch-Waxman or pediatric extensions are not granted, the intangible asset will be amortized over a shorter period. No amounts were recorded as liabilities relating to the license agreements included in the consolidated financial statements as of September 30, 2011, since the amounts, timing and likelihood of these payments are unknown and will depend on the successful outcome of future clinical trials, regulatory filings, favorable regulatory approvals, growth in product sales and other factors.

Pursuant to the amended and restated sublicense agreement, Novartis has exclusive commercialization rights to all formulations of Fanapt® in the U.S. and Canada. Novartis is responsible for the further clinical development activities in the U.S. and Canada, including the development of a long-acting injectable (or depot) formulation of Fanapt®. Pursuant to the amended and restated sublicense agreement, we received an upfront payment of \$200.0 million and are eligible for additional payments totaling up to \$265.0 million upon the achievement of certain commercial and development milestones for Fanapt® in the U.S. and Canada. We also receive royalties, which, as a percentage of net sales, are in the low double-digits, on net sales of Fanapt® in the U.S. and Canada. In addition, we are no longer required to make any future milestone payments with respect to sales of Fanapt® or any royalty payments with respect to sales of Fanapt® in the U.S. and Canada. We retain exclusive rights to Fanapt® outside the U.S. and Canada and have exclusive rights to use any of Novartis' data for Fanapt® for developing and commercializing Fanapt® outside the U.S. and Canada. At Novartis' option, we will enter into good faith discussions with Novartis relating to the co-commercialization of Fanapt® outside of the U.S. and Canada or, alternatively, Novartis will receive a royalty on net sales of Fanapt® outside of the U.S. and Canada. Novartis has chosen not to co-commercialize Fanapt® with us in Europe and certain other countries and will instead receive a royalty on net sales in those countries. These include, but are not limited to, the countries in the European Union as well as Switzerland, Norway, Liechtenstein, and Iceland. We have entered into agreements with the following partners for the commercialization of Fanapt® in the countries set forth below:

Country Mexico Argentina Probiomed S.A. de C.V. Biotoscana Farma S.A.

Item 3. Quantitative and Qualitative Disclosures about Market Risk.

Interest Rates

Our exposure to market risk is currently confined to our cash and cash equivalents, marketable securities and restricted cash. We currently do not hedge interest rate exposure. We have not used derivative financial instruments for speculation or trading purposes. Because of the short-term maturities of our cash and cash equivalents and marketable securities, we do not believe that an increase in market rates would have any significant impact on the realized value of our investments.

Marketable Securities

We deposit our cash with financial institutions that we consider to be of high credit quality and purchase marketable securities which are generally investment grade, liquid, short-term fixed income securities and money-market instruments denominated in U.S. dollars.

Item 4. Controls and Procedures.

Conclusion Regarding the Effectiveness of Disclosure Controls and Procedures

Under the supervision and with the participation of our management, including our Chief Executive Officer and Chief Financial Officer, we evaluated the effectiveness of the design and operation of our disclosure controls and procedures (as defined in Rules 13a-15(e) and 15d-15(e) under the Exchange Act) as of September 30, 2011. Based upon that evaluation, our Chief Executive Officer and Chief Financial Officer concluded that our disclosure controls and procedures are effective as of September 30, 2011, the end of the period covered by this quarterly report, to ensure that the information required to be disclosed by us in the reports that we file or submit under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in the SEC's rules and forms, and that such information is accumulated and communicated to our management, including our Chief Executive Officer and Chief Financial Officer, as appropriate to allow timely decisions regarding required disclosures.

Changes in Internal Control over Financial Reporting

There has been no change in our internal control over financial reporting (as defined in Rules 13a-15(f) and 15d-15(f) of the Exchange Act) during the third quarter of 2011 that has materially affected, or is reasonably likely to materially affect, our internal control over financial reporting.

PART II — OTHER INFORMATION

Item 1. Legal Proceedings.

None.

Item 1A. Risk Factors.

In our Annual Report on Form 10-K for the fiscal year ended December 31, 2010, filed with the SEC on March 10, 2011, we identify under Item 1A important factors which could affect our business, financial condition, results of operations and future operations and could cause our actual results for future periods to differ materially from our anticipated results or other expectations, including those expressed in any forward-looking statements made in this Form 10-Q. There have been no material changes in our risk factors subsequent to the filing of our Form 10-K for the fiscal year ended December 31, 2010.

Item 2. Unregistered Sales of Equity Securities and Use of Proceeds.

None.

Item 3. Defaults Upon Senior Securities.

None

Item 4. Removed and Reserved.

Item 5. Other Information.

None.

Item 6. Exhibits.

Exhibit <u>Number</u>	<u>Description</u>
10.42	Lease between Square 54 Office Owner LLC (as Landlord) and the registrant (as Tenant) dated July 25, 2011.
31.1	Certification of the Chief Executive Officer, as required by Section 302 of the Sarbanes-Oxley Act of 2002.
31.2	Certification of the Chief Financial Officer, as required by Section 302 of the Sarbanes-Oxley Act of 2002.
32.1	Certification of the Chief Executive Officer (Principal Executive Officer) and Chief Financial Officer (Principal Financial Officer and Principal Accounting Officer), as required by Section 906 of the Sarbanes-Oxley Act of 2002.
101	The following financial information from this Quarterly Report on Form 10-Q for the fiscal quarter ended September 30, 2011, formatted in XBRL (eXtensible Business Reporting Language) and furnished electronically herewith: (i) Condensed Consolidated Balance Sheets as of September 30, 2011 and December 31, 2010; (ii) Condensed Consolidated Statements of Operations for the three and nine months ended September 30, 2011 and 2010; (iii) Condensed Consolidated Statement of Changes in Stockholders' Equity for the nine months ended September 30, 2011; (iv) Condensed Consolidated Statements of Cash Flows for the nine months ended September 30, 2011 and 2010; and (v) Notes to Condensed Consolidated Financial Statements.

The certification attached as Exhibit 32.1 that accompanies this Quarterly Report on Form 10-Q is not deemed filed with the Securities and Exchange Commission and is not to be incorporated by reference into any filing of Vanda Pharmaceuticals Inc. under the Securities Act of 1933, as amended, or the Securities Exchange Act of 1934, as amended, whether made before or after the date of this Quarterly Report on Form 10-Q, irrespective of any general incorporation language contained in such filing.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

Vanda Pharmaceuticals Inc. November 7, 2011 Mihael H. Polymeropoulos, M.D. Mihael H. Polymeropoulos, M.D. President and Chief Executive Officer (Principal Executive Officer) November 7, 2011 /s/ James P. Kelly

James P. Kelly Senior Vice President and Chief Financial Officer (Principal Financial Officer and Principal Accounting Officer)

VANDA PHARMACEUTICALS INC.

EXHIBIT INDEX

Exhibit Number	<u>Description</u>
10.42	Lease between Square 54 Office Owner LLC (as Landlord) and the registrant (as Tenant) dated July 25, 2011.
31.1	Certification of the Chief Executive Officer, as required by Section 302 of the Sarbanes-Oxley Act of 2002.
31.2	Certification of the Chief Financial Officer, as required by Section 302 of the Sarbanes-Oxley Act of 2002.
32.1	Certification of the Chief Executive Officer (Principal Executive Officer) and Chief Financial Officer (Principal Financial Officer and Principal Accounting Officer), as required by Section 906 of the Sarbanes-Oxley Act of 2002.
101	The following financial information from this Quarterly Report on Form 10-Q for the fiscal quarter ended September 30, 2011, formatted in XBRL (eXtensible Business Reporting Language) and furnished electronically herewith: (i) Condensed Consolidated Balance Sheets as of September 30, 2011 and December 31, 2010; (ii) Condensed Consolidated Statements of Operations for the three and nine months ended September 30, 2011 and 2010; (iii) Condensed Consolidated Statement of Changes in Stockholders' Equity for the nine months ended September 30, 2011; (iv) Condensed Consolidated Statements of Cash Flows for the nine months ended September 30, 2011 and 2010; and (v) Notes to Condensed Consolidated Financial Statements.

The certification attached as Exhibit 32.1 that accompanies this Quarterly Report on Form 10-Q is not deemed filed with the Securities and Exchange Commission and is not to be incorporated by reference into any filing of Vanda Pharmaceuticals Inc. under the Securities Act of 1933, as amended, or the Securities Exchange Act of 1934, as amended, whether made before or after the date of this Quarterly Report on Form 10-Q, irrespective of any general incorporation language contained in such filing.

LEASE

BETWEEN

SQUARE 54 OFFICE OWNER LLC

(as Landlord)

AND

VANDA PHARMACEUTICALS INC.

(as Tenant)

2200 Pennsylvania Avenue, N.W. Washington, D.C.

2200 Pennsylvania Ave Vanda Pharmaceuticals Inc.

TABLE OF CONTENTS

ARTICLE I THE PREMISES	2
ARTICLE II TERM	3
ARTICLE III BASE RENT	4
ARTICLE IV ADDITIONAL RENT	6
ARTICLE V SECURITY DEPOSIT	15
ARTICLE VI USE OF PREMISES	19
ARTICLE VII ASSIGNMENT AND SUBLETTING	22
ARTICLE VIII TENANT'S MAINTENANCE AND REPAIRS	28
ARTICLE IX TENANT ALTERATIONS	29
ARTICLE X SIGNS AND FURNISHINGS	32
ARTICLE XI TENANT'S EQUIPMENT	33
ARTICLE XII ENTRY AND INSPECTION BY LANDLORD	34
ARTICLE XIII TENANT'S INDEMNITY AND INSURANCE	34
ARTICLE XIV SERVICES AND UTILITIES	42
ARTICLE XV LIABILITY OF LANDLORD	46
ARTICLE XVI RULES AND REGULATIONS	47
ARTICLE XVII DAMAGE OR DESTRUCTION	48
ARTICLE XVIII CONDEMNATION	49
ARTICLE XIX DEFAULT BY TENANT	51
ARTICLE XX BANKRUPTCY	55
ARTICLE XXI SUBORDINATION	56
ARTICLE XXII HOLDING OVER	58
ARTICLE XXIII COVENANTS OF LANDLORD	59
ARTICLE XXIV PARKING	59
ARTICLE XXV GENERAL PROVISIONS	61
ARTICLE XXVI COMMUNICATIONS AND ACCESS; BUILDING RISERS	66

i

2200 Pennsylvania Avenue

Vanda Pharmaceuticals Inc.

RIDER NO. 1 Renewal

Diagram of Premises EXHIBIT A

EXHIBIT A-1 Office Parking Area, Residential Parking Area, GWU Parking Area and Garage Common Area

EXHIBIT A-2 Description of Land **EXHIBIT B**

Work Agreement

Schedule I Base Building Office Shell Definition

Schedule II List of Building Plans and Specifications

Schedule III [Intentionally Omitted] Schedule IV Rules for Contractors Schedule V Close-Out Requirements

EXHIBIT C Rules and Regulations **EXHIBIT D** Form of Declaration EXHIBIT E Janitorial Specifications

EXHIBIT F Form of Acceptable Letter of Credit EXHIBIT G Current List of Additional Insureds

EXHIBIT H Acceptable Forms of Certificates of Insurance

EXHIBIT I List of Environmental Reports

EXHIBIT J Form of Current Ground Lessor's Nondisturbance Agreement

LEASE

THIS LEASE (this "Lease") is made as of the 25th day of July, 2011 (the "Effective Date"), by and between SQUARE 54 OFFICE OWNER LLC, a Delaware limited liability company (hereinafter referred to as "Landlord"), and VANDA PHARMACEUTICALS INC., a Delaware corporation (hereinafter referred to as "Tenant").

RECITALS:

- A. The George Washington University, a federally chartered corporation ("**GWU**"), owns fee simple title to the property known for assessment and taxation purposes as Lots 841, 842, 7000, 7001, 7002, 7003, 7004, 7005, 7006, 7007, 7008, 7009, 7010, 7011 and 7012 in Square 54 in the District of Columbia in the subdivision made by The George Washington University in said Square 54 (the "**GW Property**").
- B. The GW Property has been developed as a mixed use development consisting of (collectively, the "**Project**"): (i) a Class A office building (as more particularly described below), (ii) a high-end luxury residential building (with two towers consisting in the aggregate of approximately 272,000 rentable square feet of residential space), including affordable housing units (collectively, the "**Residential Building**"), (iii) retail space to be located within the office building and the residential building of approximately 72,000 rentable square feet (the "**Retail Space**"), (iv) a parking garage (as more particularly described below) and (v) a common courtyard.
- C. Pursuant to that certain Lease dated February 4, 2008 and effective as of February 1, 2008, between GWU, as ground lessor, and Square 54 Residential Owner LLC ("Residential Owner"), as ground lessee (as amended by instrument recorded on May 10, 2011, and as the same may be further amended from time to time, the "Residential Ground Lease"), GWU leases a portion of the GW Property to Residential Owner for the construction of the Residential Building. The Residential Building includes certain below grade areas located within the Garage (but expressly part of the gross area of the Residential Building and not part of the gross area of the Garage) that solely serve the Residential Building, including without limitation, the residential shuttle elevator lobby vestibules, residential mechanical room, residential telephone room, residential switchgear room, residential standby power room, residential domestic water room, residential fire pump room, and residential support services room (collectively, the "Residential MEP Rooms")
- D. Pursuant to that certain Lease dated February 4, 2008 and effective as of February 1, 2008, between GWU, as ground lessor, and Landlord, as ground lessee (as amended by instrument recorded on May 10, 2011, and as the same may be amended from time to time, the "Office Ground Lease"), GWU leases a portion of the GW Property as described on <u>Exhibit A-2</u> attached hereto (the "Land") to Landlord. Landlord has constructed an office building (the "Building") on the Land.

- E. Landlord and the Residential Owner have constructed a parking garage on the GW Property consisting of five (5) levels below grade (the "Garage"), a portion of which will serve the Building (the "Office Parking Area"), and a loading dock (the "Loading Dock"). In addition to the Office Parking Area within the Garage, a portion of the Garage serves the Residential Building (the "Residential Parking Area"), a portion of the Garage serves GWU (the "GWU Parking Area"), and the remaining portion of the Garage will constitute common area (the "Garage Common Area"). The Office Parking Area, Residential Parking Area, GWU Parking Area and Garage Common Area all are more particularly shown on Exhibit A-1 attached hereto and made a part hereof. Landlord also has constructed on the Land a common courtyard (the "Common Courtyard"). GWU, Landlord and the Residential Owner have entered into a Declaration of Cross-Easements and Operating, Parking and Common Area Agreement (as the same may be amended from time to time, the "REA") addressing the operation, maintenance and repair of the Garage, the Loading Dock and the Common Courtyard.
- F. The Building is located at 2200 Pennsylvania Avenue, N.W., Washington, D.C., and consists of ten (10) stories at and above grade, comprised of an "East Tower" and a "West Tower," and certain below grade areas located within the Garage (but expressly part of the gross Building area and not part of the gross Garage area) that solely serve the Building, including without limitation, the office shuttle elevator lobby vestibules, office chiller room, office telephone room, office gas room, office security room, office engineering shop, office switchgear room, office standby power room, office domestic water room, and office fire pump room (collectively, the "Office MEP Rooms"), such Building totaling approximately 460,000 total rentable square feet, consisting of approximately 440,000 square feet of rentable area of office space, sometimes hereinafter referred to as the "Office Space" and approximately 20,000 square feet of rentable area of retail space, sometimes hereinafter referred to as the "Office Building Retail Space."
- G. Tenant desires to lease space in the Building from Landlord, and Landlord is willing to lease space in the Building to Tenant, upon the terms, conditions, covenants and agreements set forth herein.

NOW, THEREFORE, the parties hereto, intending legally to be bound, hereby covenant and agree as set forth below:

ARTICLE I THE PREMISES

1.1 Landlord hereby demises and leases to Tenant and Tenant hereby leases from Landlord, for the term and upon the terms, conditions, covenants and agreements herein provided, twenty-one thousand four hundred (21,400) square feet of rentable area, located on, and comprising the entire rentable area of, the third (3rd) floor of the East Tower of the Building ("Premises"), such amount of rentable area having been conclusively determined and agreed-upon by the parties, it being expressly understood and agreed that Tenant shall have no right of remeasurement with respect to the Premises and/or Building (or any portion thereof). The location and configuration of the Premises are outlined on Exhibit A attached hereto and made a part hereof.

- 1.2 The lease of the Premises includes the right, together with other tenants of the Building and members of the public, to use the common and public areas within the Building, but includes no other rights not specifically set forth herein. The lease of the Premises also is subject to the Office Ground Lease and any covenants, conditions and restrictions of record.
- 1.3 The rentable area in the Premises and the Building have been calculated in accordance with the American National Standards Institute, Inc./Building Owners and Managers Association standard method of measuring floor area, ANSI/BOMA Z65.1-1996 ("BOMA"). Notwithstanding anything in this Lease to the contrary, both parties acknowledge and agree that the rentable square footage of the fitness facility referenced in Section 14.7 below and the property management office have been included in calculating the "core area factor" for the Building.

ARTICLE II TERM

- 2.1 All of the provisions of this Lease shall be in full force and effect from and after the Effective Date. The term of this Lease ("Lease Term") shall be for one hundred thirty-two (132) full calendar months, commencing on the Lease Commencement Date, as determined pursuant to Section 2.2 hereof, and continuing for a period of one hundred thirty-two (132) full calendar months thereafter, unless such Lease Term shall be terminated earlier in accordance with the provisions hereof or shall be extended in accordance with the provisions of Rider No. 1 to this Lease. Notwithstanding the foregoing, if the Lease Commencement Date shall occur on a day other than the first day of a month, the Lease Term shall commence on such date and continue for the balance of such month and for a period of one hundred thirty-two (132) full calendar months thereafter. The term "Lease Term" shall include any and all renewals and extensions of the term of this Lease.
- 2.2 The Premises will be delivered to Tenant in Ready for Buildout Condition (as defined in Exhibit B) promptly following the date on which this Lease has been fully executed and delivered (the "Premises Delivery"), and the "Lease Commencement Date" shall be the date that is the earlier to occur of (a) the date on which Tenant commences beneficial use of the Premises for the conduct of its business and (b) April 1, 2012. Tenant's taking possession of the Premises shall constitute Tenant's acknowledgement that the Premises is in Ready for Buildout Condition. Tenant shall be deemed to have commenced beneficial use of the Premises when Tenant commences business operations in the Premises. In the event Premises Delivery is delayed, regardless of the reasons or causes of such delay, this Lease shall not be rendered void or voidable as a result of such delay, the Term of this Lease shall commence on the Lease Commencement Date as determined in accordance with the foregoing, and except as expressly provided herein, Landlord shall not have any liability whatsoever to Tenant on account of any such delay. Notwithstanding the foregoing, if Premises Delivery does not occur on or before the date that is five (5) business

days following the date on which this Lease has been fully executed and delivered by Landlord and Tenant, including Tenant's delivery to Landlord of the Advanced Rent and Security Deposit required hereunder (each such day beyond such five business day period to constitute "Premises Delivery Delay") and/or in the event of any Landlord Delay (as defined in Paragraph 12 of Exhibit B hereto), then, as Tenant's sole and exclusive remedy in connection therewith, the April 1, 2012 date set forth in this Section 2.2(b) above shall be extended by one (1) day for each such day of Premises Delivery Delay or Landlord Delay, as applicable.

- 2.3 Promptly after the Lease Commencement Date has occurred, Landlord and Tenant shall execute a written declaration setting forth the Lease Commencement Date, the date upon which the initial term of this Lease will expire, and the other information set forth therein. The form of such declaration is attached hereto as Exhibit D, and is made a part hereof. Any failure of the parties to execute such declaration shall not affect the validity of the Lease Commencement Date as determined in accordance with this Article.
- 2.4 For purposes of this Lease, the term "Lease Year" shall mean a period of twelve (12) consecutive calendar months, commencing on the Lease Commencement Date and each successive twelve (12) month period thereafter, except that if the Lease Commencement Date shall occur on a date other than the first day of a month, then the first Lease Year shall also include the period from the Lease Commencement Date to the first day of the following month.

ARTICLE III BASE RENT

3.1 (a) During the Lease Term, Tenant shall pay to Landlord as annual base rent (used interchangeably as "Base Rent" or "base rent") for the Premises, without set off, deduction or demand, an amount equal to the product of Forty-Seven and 00/100 Dollars (\$47.00), multiplied by the total number of square feet of rentable area in the Premises as set forth in Section 1.1, which amount shall be increased as provided in Section 3.2 below. The annual base rent payable hereunder during each Lease Year shall be divided into equal monthly installments and such monthly installments shall be due and payable in advance on the first day of each month during such Lease Year. Concurrently with the signing of this Lease, Tenant shall pay to Landlord the sum of Eighty-Three Thousand Eight Hundred Sixteen and 67/100 Dollars (\$83,816.67) ("Advanced Rent"), which sum shall be credited by Landlord toward the monthly installment of annual base rent due on the first (1st) day of the first calendar month falling after the month in which the Lease Commencement Date occurs (subject to any abatement to which Tenant is entitled pursuant to Section 3.1(b) below). In addition, if the Lease Term begins on a date other than on the first day of a month, rent from such date until the first day of the following month shall be prorated on a per diem basis at the base rate payable during the first month, and such prorated rent shall be payable in advance on the day immediately following the last day of the Abatement Period (as hereinafter defined).

- (b) Notwithstanding anything to the contrary contained in this Article III and provided no Event of Default by Tenant has occurred, Landlord hereby agrees to grant Tenant an abatement of the annual base rent payable hereunder (and the Operating Expenses payable pursuant to Section 4.1(a) below) for a period of twelve (12) full calendar months from the Lease Commencement Date, as defined in Section 2.2 above (the "Abatement Period"). Thereafter, commencing on the first day of the second (2nd) Lease Year, Tenant shall pay the full amount of annual base rent due in accordance with the provisions of this Article III (and the full amount of Tenant's proportionate share of Operating Expenses due in accordance with the provisions of Article IV). Notwithstanding anything to the contrary in this Section 3.1(b), the rent escalation, as required by Section 3.2 below, shall be based on the full and unabated amount of rent payable for the first (1st) Lease Year as set forth in Section 3.1(a) above.
- 3.2 (a) Commencing on the first (1st) day of the second (2nd) Lease Year and on the first day of each and every Lease Year thereafter during the Lease Term, the annual base rent shall be increased by two and fifty hundredths percent (2.50%) of the amount of annual base rent payable for the preceding Lease Year.
- (b) Based on the foregoing, the Annual Base Rent and Monthly Base Rent payable for the Premises during the initial Lease Term shall be as follows (subject to Section 3.1(b) above):

Lease Year	Base Rate/RSF		Annual Base Rent	Monthly Base Rent	
1	\$	47.00	\$ 1,005,800.04	\$	83,816.67
2	\$	48.18	\$ 1,031,052.00	\$	85,921.00
3	\$	49.38	\$ 1,056,732.00	\$	88,061.00
4	\$	50.61	\$ 1,083,054.00	\$	90,254.50
5	\$	51.88	\$ 1,110,231.96	\$	92,519.33
6	\$	53.18	\$ 1,138,052.04	\$	94,837.67
7	\$	54.51	\$ 1,166,514.00	\$	97,209.50
8	\$	55.87	\$ 1,195,617.96	\$	99,634.83
9	\$	57.26	\$ 1,225,364.04	\$	102,113.67
10	\$	58.70	\$ 1,256,180.04	\$	104,681.67
11	\$	60.16	\$ 1,287,423.96	\$	107,285.33

3.3 All rent shall be paid to Landlord in legal tender of the United States at c/o Boston Properties, P.O. Box 3557, Boston, MA 02241-3557, or to such other address as Landlord may designate from time to time by written notice to Tenant. If Landlord shall at any time accept rent after it shall become due and payable, such acceptance shall not excuse a delay upon subsequent occasions, or constitute or be construed as a waiver of any of Landlord's rights hereunder. If any sum payable by Tenant under this Lease is paid by check which is returned due to insufficient funds, stop payment order, or otherwise, then: (a) such event shall be treated as a failure to pay such sum when due; and (b) in addition to

all other rights and remedies of Landlord hereunder, Landlord shall be entitled (i) to impose a returned check charge of Fifty Dollars (\$50.00) to cover Landlord's administrative expenses and overhead for processing, and (ii) to require that all future payments be remitted by wire transfer, money order, or cashier's or certified check.

3.4 Landlord and Tenant agree that no rental or other payment for the use or occupancy of the Premises is or shall be based in whole or in part on the net income or profits derived by any person or entity from the Building or the Premises. Tenant further agrees that it will not enter into any sublease, license, concession or other agreement for any use or occupancy of the Premises which provides for a rental or other payment for such use or occupancy based in whole or in part on the net income or profits derived by any person or entity from the Premises so leased, used or occupied. Nothing in the foregoing sentence, however, shall be construed as permitting or constituting Landlord's approval of any sublease, license, concession, or other use or occupancy agreement not otherwise approved by Landlord in accordance with the provisions of Article VII.

ARTICLE IV ADDITIONAL RENT

4.1 Operating Expenses and Real Estate Taxes.

(a) Commencing on the Lease Commencement Date (subject to any abatement to which Tenant is entitled, as described below in this sentence) and continuing with each calendar year thereafter during the Lease Term, Tenant shall pay Landlord, as additional rent for the Premises, Tenant's proportionate share of the operating expenses incurred by Landlord in connection with the management, operation and ownership of the Building including the portion of the Garage and the Loading Dock and the Common Courtyard serving the Building, and the Garage Common Area ("Operating Expenses") during any calendar year falling entirely or partly within the Lease Term; provided, however, that Tenant is hereby granted an abatement of the foregoing additional rent for the Abatement Period, subject to the terms of Section 3.1(b) above. For purposes of this Article IV Tenant's proportionate share of such Operating Expenses shall be that percentage which is equal to a fraction, the numerator of which is the number of square feet of rentable area in the Premises from time to time and the denominator of which is the total number of square feet of rentable area in the Building from time to time, excluding the number of square feet devoted to storage space and parking. It is understood that the number comprising such denominator is subject to change because of changes in the use or configuration of space in the Building or the addition of space to the Building or the deletion of space from the Building or in the amount of space leased by tenants who pay by separate meter for their electrical and/or janitorial, cleaning, or other utilities or services so that Tenant actually pays its fair share of Operating Expenses. The denominator with respect to Real Estate Taxes (as defined in Section 4.1(c) below) shall be calculated based on the total number of square feet of rentable area in the Building, including portions of the Building occupied by retail tenants but exclusive of the Garage and storage areas. Space leased by retail tenants is exclude

rubbish removal, water, electricity and janitorial service exclusively provided to their premises (and expenses for such services to their premises are excluded from Operating Expenses), and (ii) other costs and expenses determined by Landlord to have been incurred in connection with services related to the office portion of the Building. However, space leased by retail tenants is included in the denominator with respect to common area water, electricity and janitorial, and all other categories of expenses included in Operating Expenses. Tenant acknowledges and understands that with respect to certain of the Operating Expenses set forth herein (e.g., costs relating to the Garage Common Area, the Loading Dock and the Common Courtyard), Tenant will be paying its proportionate share of Operating Expenses which are attributable to the Building's proportionate share of such expenses relative to the Project, with appropriate adjustments to the extent such expenses are not attributable to circumstances or conditions present in the Building or otherwise applicable to the Project as a whole. The specific obligations of Tenant with respect to such expenses shall be governed by the remaining sections of this Article IV. Tenant's proportionate share shall increase in the event Tenant expands the Premises.

- (b) Operating Expenses shall include, without limitation, the costs and expenses described in Subsection (1) below, but shall not include the costs and expenses described in Subsection (2) below.
 - (1) Included costs and expenses (which shall in all cases be net of any discounts, credits, reimbursements and rebates received by Landlord):
- (i) Gas, water, sewer, electricity and other utility charges (including surcharges) of every type and nature (except to the extent separately metered to individual tenants and payable by such tenants directly to the applicable utility, or otherwise reimbursed to Landlord by tenants of the Building).
 - (ii) Insurance premiums paid by Landlord.
- (iii) Personnel costs of the Building, including, but not limited to, salaries, wages, fringe benefits and other direct and indirect costs of engineers, superintendents, watchmen, porters, property accountants and any other personnel related to the management, maintenance, repair and operation of the Building ("Personnel").
- (iv) Costs of service and maintenance contracts, including, but not limited to, chillers, boilers, controls, elevators, mail chute, windows, access control service, landscaping, snow and ice removal, management fees in an amount not to exceed for any calendar year during the initial Lease Term more than 5% of the annual gross revenues for the Building, and air and water quality testing.
- (v) All other maintenance and repair expenses and supplies which are deducted by Landlord in computing its Federal income tax liability.
- (vi) Amortization over the Approved Period (as defined below), with interest at Landlord's cost of financing, or, if the improvement is not financed, at the prime

rate reported by the Bank of America on the date of such expenditure, for capital expenditures made by Landlord (A) to reduce operating expenses if and to the extent the annual reduction in Operating Expenses will be equal to or will exceed the annual amortization and financing costs therefor, or (B) to comply with all present and future laws, ordinances (including zoning ordinances and land use requirements), regulations and orders of the District of Columbia, the United States of America and any other governmental or quasi-governmental agency having jurisdiction over the Premises (collectively, "Legal Requirements"), which Legal Requirements are first applicable to the Building after the Lease Commencement Date; the "Approved Period" shall mean the time period equal to the longest allowable useful life of the improvement permitted under generally accepted accounting principles, except that with respect to an improvement made for the purpose of reducing Operating Expenses, Landlord may reduce such time period to the number of years that it will take to fully amortize the cost of the capital expenditure if the yearly amortization amount (including interest as aforesaid) is equal to the projected annual savings as reasonably estimated by Landlord.

(vii) Any other costs and expenses reasonably incurred by Landlord in maintaining or operating the Building (including major repairs for maintenance purposes, but excluding capital improvements, except as permitted pursuant to subsection 4.1(b)(1)(vi) above), except as provided in (2) below.

(viii) Real Estate Taxes (as hereinafter defined).

- (ix) The costs of any additional services not provided to the Building at the Lease Commencement Date but thereafter provided by Landlord in the prudent management of the Building.
- (x) Charges for concierge (if any), access control, janitorial, and cleaning services (including supplies) for operation and maintenance of the Building (including the loading dock serving the Building), the fitness facility and the roof deck to the extent available for use by all office tenants of the Building.
- (xi) Personnel costs of the regional property manager and regional engineer, even if such persons work off-site, so long as such persons are not part of the corporate office and only if such person's time is allocable to the Building, consistent with the portion of such person's time allocated to the Building.
- (xii) Costs of maintaining management or engineering offices serving the Building, including, without limitation, the costs of telephone services, office equipment, including upgrades and replacements thereof, and office supplies, but excluding any cost for the initial furnishing of such offices.
- (xiii) Accounting expenses reasonably incurred by Landlord in calculating Operating Expenses and legal fees and expenses reasonably incurred by Landlord in connection with proceedings undertaken to reduce Operating Expenses.

(xiv) Project Common Expenses, hereinafter defined.

(xv) Any costs or expenses incurred by Landlord with respect to the imposition of taxes, charges or fees levied, assessed or imposed against the Building or Landlord or any of Landlord's constituent members in connection with the development, financing, construction, operation, maintenance and/or use of any major league baseball stadium and/or other sports complex in the District of Columbia.

(2) Excluded costs and expenses:

- (i) Principal, interest or other amounts payable on indebtedness, debt amortization or ground rent paid by Landlord in connection with any mortgages, deeds of trust or other financing encumbrances, or ground leases of the Building.
 - (ii) Capital improvements to the Building other than those permitted in subsection 4.1(b)(1)(vi) above.
- (iii) Legal, auditing, consulting and professional fees and other costs paid or incurred in connection with financings, refinancings or sales of any interest in Landlord or of Landlord's interest in the Building, or in connection with any ground lease (including, without limitation, recording costs, mortgage recording taxes, title insurance premiums and other similar costs, but excluding those legal, auditing, consulting and professional fees and other costs incurred in connection with the normal and routine maintenance and operation of the Building).
- (iv) Legal fees, space planner's fees, architect's fees, leasing and brokerage commissions, advertising and promotional expenditures and any other advertising and marketing expenses incurred in connection with the leasing of space in the Building (including new leases, lease amendments, lease terminations and lease renewals).
- (v) The cost of any items to the extent to which such cost is reimbursed to Landlord by tenants of the Building (other than pursuant to this Section 4.1), other third parties, or is covered by a warranty to the extent of reimbursement for such coverage.
- (vi) Expenditures for any leasehold improvements which are made in connection with the preparation of any portion of the Building for occupancy by any tenant or which are not made generally to or for the benefit of the Building.
- (vii) The cost of performing work or furnishing service to or for any tenant other than Tenant, at Landlord's expense, to the extent such work or service is in excess of any work or service Landlord is obligated to provide to Tenant or generally to other tenants in the Building at Landlord's expense.
- (viii) The cost of repairs or replacements incurred by reason of fire or other casualty, or condemnation (other than costs not in excess of the deductible on any insurance maintained by Landlord which provides a recovery for such repair or replacement), to the extent Landlord actually receives proceeds of property and casualty insurance policies or condemnation awards or would have received such proceeds had Landlord maintained the insurance required to be maintained by Landlord pursuant to this Lease.

(ix) The cost of acquiring (but not of maintaining) sculptures, paintings or other objects of fine art in the Building.

(x) Reserves for bad debt loss or rent loss, or any other reserves, including without limitation, reserves for tenant improvements and leasing commissions.

- (xi) Unfunded contributions to operating expense reserves by other tenants.
- (xii) Contributions to charitable or political organizations.
- (xiii) Damage and repairs necessitated by the gross negligence or willful misconduct of Landlord Parties.
- (xiv) Fees, costs and expenses incurred by Landlord in connection with or relating to claims against or disputes with tenants of the

Building.

(xv) Interest, fines or penalties for late payment or violations of Legal Requirements by Landlord, if any, except to the extent incurring such expense is either (a) a reasonable business expense under the circumstances, or (b) caused by a corresponding late payment or violation of a Legal Requirement by Tenant, in which event Tenant shall be responsible for the full amount of such expense.

(xvi) The cost of remediation and removal of "Hazardous Materials" (as defined in Section 6.3) in the Building required by "Environmental Law" (as defined in Section 6.3), provided, however, that the provisions of this clause xvi shall not preclude the inclusion of costs with respect to materials (whether existing at the Building as of the Lease Commencement Date or subsequently introduced to the Building) which are not as of the Lease Commencement Date (or as of the date of introduction) deemed to be Hazardous Materials under applicable Hazardous Materials Laws but which are subsequently deemed to be Hazardous Materials under applicable Hazardous Materials Laws (it being understood and agreed that Tenant shall nonetheless be responsible under Article VI for all costs of remediation and removal of Hazardous Materials to the extent caused by Tenant Parties).

(xvii) Costs of replacements, alterations or improvements necessary to make the Building comply with Legal Requirements in effect and applicable to the Building prior to the Lease Commencement Date, provided, however, that the provisions of this clause xvii shall not preclude the inclusion of costs of compliance with Legal Requirements enacted prior to the Effective Date if such compliance is required for the first time by reason of any amendment, modification or reinterpretation of a Legal Requirement which is imposed after the Effective Date.

(xviii) Costs for the original construction and development of the Building (including any "tap fees" or one-time lump sum sewer or water connection fees) and nonrecurring costs for the repair or replacement of any structural portion of the Building made necessary as a result of defects in the original design, workmanship or materials.

(xix) Costs and expenses incurred for the administration of the entity which constitutes Landlord, as the same are distinguished from the costs of operation, management, maintenance and repair of the Building, including, without limitation, entity accounting and legal matters.

(xx) Salaries and all other compensation (including fringe benefits) of partners, officers and executives above the grade of regional property manager or regional engineer.

(xxi) The wages and benefits of any employee who does not devote substantially all of his or her employed time to the Building unless such wages and benefits are reasonably allocated to the Building.

(xxii) Except as may be otherwise expressly provided in this Lease with respect to specific items, the cost of any services or materials provided by any party related to Landlord, to the extent such cost exceeds the reasonable cost for such services or materials in Class A office buildings in the Market Area absent such relationship.

(xxiii) Depreciation for the Building and personal property contained therein.

(xxiv) Except as may be otherwise expressly provided in this Lease with respect to specific items, the cost of any services or materials provided by any party related to Landlord, to the extent such cost exceeds the reasonable cost for such services or materials in Class A office buildings in the Market Area absent such relationship.

(xxv) Compensation paid to clerks, attendants or other persons in commercial concessions operated by Landlord or by the operator thereof (i.e., newsstands).

(xxvi) Any entertainment expenses of Landlord's employees, and any travel expenses not related to the operation or management of the Building.

(c) "Real Estate Taxes" shall mean (i) all real estate taxes and other impositions, including general and special assessments, property owner association fees, business improvement district taxes, arena taxes, and other similar taxes and assessments if any, which are imposed upon Landlord or assessed against the Building or the Land upon which the Building is situated; (ii) any other present or future taxes or governmental charges that are imposed upon Landlord or assessed against the Building or the Land, including, but not limited to, any tax levied on or measured by the rents payable by tenants of the Building, which are in the nature of, or in substitution for, real estate taxes; (iii) all taxes which are imposed upon

Landlord, and which are assessed against the value of any improvements to the Premises made by Tenant or any machinery, equipment, fixtures or other personal property of Tenant used therein; (iv) expenses (including attorneys' fees) incurred in reviewing, protesting, negotiating or seeking (whether formally or informally) a reduction or abatement of Real Estate Taxes; (v) any rental or other charges or fees imposed upon Landlord in connection with the lease or use of any vault space(s); (vi) any taxes or other charges levied pursuant to the Business Improvement Districts Act of 1996 or any amendments thereto; and (vii) any taxes, charges or fees imposed upon Landlord in connection with the development, financing, construction, operation, maintenance and/or use of any major league baseball stadium and/or other sports complex in the District of Columbia, but only if and to the extent such taxes, charges or fees are includable as Operating Expenses pursuant to this Article IV (including without limitation, Section 4(b)(1)(xv) hereof). Notwithstanding the foregoing, except as expressly included above, Real Estate Taxes shall not include any income taxes, excess profits taxes, excise taxes, franchise taxes, estate taxes, succession taxes and transfer taxes, except to the extent any of such taxes are in the nature of or are in substitution for or recharacterization or replacement of Real Estate Taxes. If Landlord contests the Real Estate Taxes for any calendar year, and such contest results in an increase in Real Estate Taxes for such calendar year, then Landlord shall have the right to bill Tenant for prior underpayments of Real Estate Taxes thereby resulting. If Landlord receives a refund of any portion of Real Estate Taxes that were included in the Real Estate Taxes paid by Tenant, then Landlord shall credit against the next estimated payment or payments due under this Article IV an amount equal to Tenant's pro rata share of the refunded taxes, less any expenses that Landlord incurred to obtain the refund or if th

(d) "Project Common Expenses" shall mean those Operating Expenses incurred by Landlord in owning, operating and/or managing the Office Parking Area, Residential Parking Area, GWU Parking Area, and Garage Common Area (as opposed to those Operating Expenses related exclusively to the Building or the Office Parking Area), as well as those Operating Expenses incurred in owning, operating and/or managing the Loading Dock and the Common Courtyard components of the Project, but only a portion of such expenses equal to the pro rata share attributable to the Building relative to the Project or the applicable portion of the Project. For purposes hereof, (i) the pro rata share of the Office Parking Area, Residential Parking Area, GWU Parking Area and Garage Common Area costs attributable to the Building shall be determined using the methodology set forth on Exhibit A-1 attached hereto as applied to the actual gross square footage of the applicable elements of the Project (which pro rata share is equal to thirty-nine and sixteen hundredths percent (39.16%) as of the date hereof based on the gross square footage of the applicable elements of the Project set forth on Exhibit A-1), and (ii) the pro rata share of the Loading Dock costs and the Common Courtyard costs attributable to the Building each shall be determined using the methodology set forth on Exhibit A-1 as applied to the actual gross square footage of the Building (including the Retail Space to be located in the Building) and the Residential Building (including the Retail Space to be located in the Residential Building and the gross square footage of the Building and the gross square footage of the Residential Building and the gross square footage of the Residential Building and the gross square footage of the Residential Building and the gross square footage of the Residential Building and the gross square footage of the Residential Building and the gross square footage of the Residential Building and the gross square footage of the Residential Building and the gr

expenditures related to the Project that would have been excluded from Operating Expenses if related solely to the Building, similarly shall be excluded from Operating Expenses. It is further understood and agreed that except with respect to certain expenses incurred in operating and/or managing the Office Parking Area, Residential Parking Area, GWU Parking Area, and Garage Common Area, as well as the expenses incurred in operating and/or maintaining the Loading Dock and Common Courtyard components of the Project, operating expenses and real estate taxes shall be separately accounted for with respect to the Building, the Residential Building and the GWU Parking Area, and all operating expenses and real estate taxes arising from the Residential Building (including the Residential MEP Rooms) shall be excluded from Operating Expenses.

4.2 In the event the average occupancy rate for the entire Building shall be less than one hundred percent (100%) or if any tenant is paying separately for electricity or other utilities or services for any calendar year, for purposes of calculating the additional rent payable by Tenant pursuant to this Article IV for each calendar year, the Operating Expenses for such calendar year shall be increased by the amount of additional costs and expenses that Landlord reasonably estimates would have been incurred if the average occupancy rate for the entire Building had been one hundred percent (100%) and as if no tenants had separately paid for electricity or other utilities and services for such calendar year. It is the intent of this provision to permit Landlord to recover from Tenant its proportionate share of Operating Expenses attributable to occupied space in the Building even though the aggregate of such expenses shall have been reduced as a result of vacancies in the Building. This Section 4.2 shall not be construed to permit Landlord to recover from Tenant additional rent pursuant to Article IV for any calendar year which, when added to the total amount of additional rent payable (whether actually paid, payable but unpaid, or that would have been payable except that it has been abated in accordance with the terms of the applicable tenant's lease) by all tenants of the Building on account of Operating Expenses for such year, will exceed the actual amount of Operating Expenses incurred by Landlord for such year.

4.3 On or about the Lease Commencement Date and at the beginning of each calendar year thereafter during the Lease Term, Landlord shall submit to Tenant a statement setting forth Landlord's reasonable estimate of (a) the amount of the Operating Expenses that are expected to be incurred during such calendar year, and (b) the computation of Tenant's proportionate share of such anticipated Operating Expenses. Except as otherwise provided herein, Tenant shall pay to Landlord on the first day of each month following receipt of such statement during such calendar year an amount equal to Tenant's proportionate share of the anticipated Operating Expenses multiplied by a fraction, the numerator of which is 1, and the denominator of which is the number of months during such calendar year which fall entirely or partly within the Lease Term and follow the date of the foregoing statement. Within approximately one hundred twenty (120) days after the expiration of each calendar year falling entirely or partly within the Lease Term, Landlord shall submit to Tenant a statement showing (i) the actual amount of Operating Expenses paid or incurred by Landlord during the immediately preceding calendar year, with reasonable detail, (ii) a computation of Tenant's proportionate share of the Operating Expenses actually incurred during the preceding calendar year, and (iii) the

aggregate amount of the estimated payments made by Tenant on account thereof. If the aggregate amount of such estimated payments exceeds Tenant's actual liability for such Operating Expenses, then Tenant shall deduct the net overpayment from its next estimated payment or payments due under this Article IV for the then current year, or, in the case of the reconciliation for the calendar year in which the Lease Term expires, Landlord shall pay Tenant the net overpayment (after deducting therefrom any amounts then due from Tenant to Landlord). If Tenant's actual liability for such amounts exceeds the estimated payments made by Tenant on account thereof, then Tenant shall pay to Landlord the total amount of such deficiency as additional rent due hereunder in accordance with Section 25.16 below.

- 4.4 In the event Tenant's obligation for the payment of Operating Expenses begins or expires on a day other than the first and last day of a calendar year, respectively, then Tenant's obligation for Operating Expenses for such partial calendar year shall be an amount equal to the product of (i) Tenant's Proportionate Share of the Operating Expenses for the full calendar year, multiplied by (ii) a fraction, the numerator of which is the number of days during such calendar year for which Tenant is obligated for the payment of Operating Expenses, and the denominator of which is three hundred sixty-five (365).
- 4.5 All payments required to be made by Tenant pursuant to this Article IV shall be paid to Landlord, without setoff or deduction, in the same manner as annual base rent is payable pursuant to Article III hereof.
- 4.6 Tenant's liability for its proportionate share of Operating Expenses described in Section 4.1 hereof for the last calendar year falling entirely or partly within the Lease Term shall survive the expiration of the Lease Term. Similarly, Landlord's obligation to refund to Tenant the excess, if any, of the amount of Tenant's estimated payments on account of such Operating Expenses for such last calendar year over Tenant's actual liability therefor shall survive the expiration of the Lease Term.
- 4.7 In the event that Tenant, in good faith, believes that the amounts paid by Tenant to Landlord relating to Operating Expenses during any calendar year falling within the Lease Term exceeded the amounts to which Landlord was entitled to hereunder and Tenant details the alleged discrepancy in writing to Landlord, then, a regular employee of Tenant or an Acceptable CPA (as defined below) shall have the right, during regular business hours and after giving not less than ten (10) business days' advance written notice to Landlord, to inspect and complete an audit of Landlord's books and records relating to such Operating Expenses for a period of one hundred eighty (180) days following receipt by Tenant of the statement required to be delivered by Landlord to Tenant pursuant to Section 4.3 hereof for such calendar year. As used herein, an "Acceptable CPA" shall mean an independent, certified public accountant who is employed by a nationally recognized accounting firm and retained by Tenant on a non-contingency basis. If such audit shows that the amounts paid by Tenant to Landlord on account of such Operating Expenses exceeded the amounts to which Landlord was entitled hereunder, or that Tenant is entitled to a credit with respect to any such Operating Expenses, Landlord shall promptly refund to Tenant the amount of such excess or the amount of such credit, as the

case may be. Similarly, if it is determined that the amounts paid by Tenant to Landlord on account of Operating Expenses were less than the amounts to which Landlord was entitled hereunder, then Tenant shall promptly pay to Landlord, as additional rent hereunder, the amount of such deficiency. Tenant shall (and shall cause its agents to) keep the results of such audit or audited statement strictly confidential and shall execute Landlord's standard form of confidentiality agreement prior to commencing the audit. All costs and expenses of any such audit shall be paid by Tenant, except that if such audit shows that the amount of Operating Expenses was overstated by more than five percent (5%) in the aggregate, Landlord shall reimburse Tenant for the reasonable out-of-pocket costs and expenses incurred by Tenant in such audit, up to a maximum of the lesser of (a) Five Thousand Dollars (\$5,000) and (b) the amount of the overstatement of Tenant's proportionate share of Operating Expenses. Notwithstanding the foregoing, if Tenant does not notify Landlord in writing of any objection to the statement required of Landlord pursuant to Section 4.3 hereof within said one hundred eighty (180) day period, then Tenant shall be deemed to have waived any such objection and any potential claim against Landlord for any refunds with respect to such Operating Expenses.

ARTICLE V SECURITY DEPOSIT

5.1 (a) Simultaneously with Tenant's execution of this Lease, Tenant shall post a Letter of Credit (as defined below) in an amount equal to Five Hundred Thousand and 00/100 Dollars (\$500,000.00), as a security deposit (hereinafter referred to as "security deposit" or "Security Deposit"), which sum shall be in addition to the Advanced Rent paid by Tenant to Landlord pursuant to Section 3.1 hereof. Among other things, Landlord has assigned (or intends to assign) to the holder of the mortgage now or hereafter encumbering the Building, all of Landlord's interest in this Lease, including, without limitation, the Security Deposit.

(b) The Security Deposit shall be security for the performance by Tenant of all of Tenant's obligations, covenants, conditions and agreements under this Lease. Within ninety (90) days after the expiration of the Lease Term, provided Tenant has vacated the Premises, Landlord shall return the Security Deposit to Tenant, less such portion thereof as Landlord shall have appropriated to satisfy any Event of Default under this Lease and such portion as Landlord reasonably believes will be payable by Tenant in connection with the reconciliation of Operating Expenses for the calendar year in which the Lease Term expires. In the event of any Event of Default by Tenant under this Lease, Landlord shall have the right, but not the obligation, to use, apply or retain all or any portion of the Security Deposit for (i) the payment of any annual base rent or additional rent or any other sum due under this Lease, (ii) the payment of any amount Landlord may spend or become obligated to spend as a result of the Event of Default, (iii) for the compensation of Landlord for any losses incurred by reason of the Event of Default, or (iv) any damage or deficiency arising in connection with the reletting of the Premises. If any portion of the Security Deposit is so used or applied, within five (5) business days after written notice to Tenant of such use or application, Tenant shall restore the Security Deposit by providing a replacement Letter of Credit, an additional

Letter of Credit or an amendment to the initial Letter of Credit, as the case may be, such that Landlord is holding one or more Letters of Credit in the aggregate amount of the Security Deposit required hereunder. Tenant's failure to restore the Letter of Credit shall constitute an Event of Default under this Lease. Tenant hereby authorizes Landlord to deposit the Security Deposit with the holder of any mortgage now or hereafter encumbering the Building if and to the extent required by said holder; provided, however, that such holder shall hold the Security Deposit subject to Tenant's rights with respect to the Security Deposit set forth herein.

(c) The Security Deposit shall be in the form of one or more unconditional, irrevocable letters of credit (each, a "Letter of Credit"), subject to the terms and conditions set forth herein. Such Letter of Credit shall (i) be in the form attached hereto as Exhibit F or otherwise in form and substance reasonably satisfactory to Landlord; (ii) be at all times in the amount of the Security Deposit (subject to any reduction to which Tenant is entitled pursuant to Section 5.1(d) below), (iii) permit multiple draws; (iv) be issued by a commercial bank reasonably acceptable to Landlord from time to time; (v) be made payable to, and expressly transferable and assignable at no charge by, the owner from time to time of the Building or, at Landlord's option, the holder of any mortgage (which transfer/assignment shall be conditioned only upon the execution of a written document in connection therewith; provided, however, that in the event the issuing bank of the Letter of Credit charges a fee for a transfer and/or assignment, any and all such fees shall be payable by Tenant); (vi) be payable at sight or by facsimile upon presentation of a simple sight draft to the issuing bank of the Letter of Credit; (vii) have a term not less than one (1) year; and (viii) be at least thirty (30) days prior to the then-current expiration date of such Letter of Credit, automatically renewed (or automatically and unconditionally extended) from time to time through the ninetieth (90th) day after the expiration of the Lease Term. Notwithstanding anything in this Lease to the contrary, any cure or grace periods set forth in this Lease shall not apply to any of the foregoing, and, specifically, if Tenant fails to timely comply with the requirements of Subsection (v) and/or (viii) above, then Landlord shall have the right to immediately draw upon the Letter of Credit without notice to Tenant and apply the proceeds to the Security Deposit. Each Letter of Credit shall be issued by and drawn on a bank reasonably approved by Landlord and at a minimum having a long-term issuer credit rating from Standard & Poor's Professional Rating Service of A or a comparable rating from Moody's Professional Rating Service, and the Letter of Credit shall be otherwise acceptable to Landlord in its sole and absolute discretion. If the issuer's credit rating is reduced below A (or equivalent) by Standard & Poor's Corporation or by Moody's Professional Rating Service, or if the financial condition of such issuer changes in any other materially adverse way, then Landlord shall have the right to require that Tenant obtain from a different issuer a substitute letter of credit that complies in all respects with the requirements of this Article, and Tenant's failure to obtain such substitute letter of credit within ten (10) days following Landlord's written demand therefor (with no other notice or cure or grace period being applicable thereto, notwithstanding anything in this Lease to the contrary) shall entitle Landlord to immediately draw upon the then existing Letter of Credit in whole or in part, without notice to Tenant. In the event the issuer of any Letter of Credit held by Landlord is placed into receivership or conservatorship by the Federal Deposit Insurance Corporation or any successor or similar entity, then, effective as of the date such receivership or conservatorship occurs, said Letter of Credit shall be deemed to not meet the requirements of this Section, and, within ten (10) days

thereof, Tenant shall replace such Letter of Credit with other collateral acceptable to Landlord in its sole and absolute discretion (and Tenant's failure to do so shall, notwithstanding anything in this Lease to the contrary, constitute an Event of Default for which there shall be no notice or grace or cure periods being applicable thereto other than the aforesaid ten (10) day period). Tenant shall be responsible for the payment of any and all costs incurred by Landlord in connection with the replacement Letter of Credit (including without limitation Landlord's reasonable attorneys' fees), which replacement is required pursuant to this Section or is otherwise requested by Tenant. Any failure or refusal of the issuer to honor the Letter of Credit shall be at Tenant's sole risk and shall not relieve Tenant of its obligations hereunder with respect to the Security Deposit.

- (d) (i) As required by Section 25.4(b) below, Tenant shall deliver to Landlord Tenant's financial statements for Tenant's most recently completed fiscal year, audited by a certified public accountant ("Financial Statements"). Tenant's Financial Statements provided to Landlord hereunder shall state, among other things, Tenant's Liquidity ("Liquidity" is equal to the sum of cash, market securities and credit line availability, if any) and Cash Flow from Operations ("Cash Flow from Operations" definition is consistent with that found in the statement of cash flows from Tenant's Financial Statements). Tenant shall make its chief financial officer reasonably available to answer any questions Landlord may have concerning such Financial Statements and shall deliver any additional information reasonably requested by Landlord to clarify or verify the data shown on the Financial Statements provided pursuant hereto.
- (ii) Provided that, as of the applicable Reduction Date (as defined below) (x) no Event of Default shall then be in existence under this Lease, (y) Tenant's then-current Liquidity is equal to or greater than the sum of Tenant's remaining obligations under this Lease, and (z) if Cash Flow from Operations is negative, Tenant's then current Liquidity also is equal to or greater than the product of (I) the annual Cash Flow from Operations multiplied by (II) the number of years remaining in the Lease Term (i.e., the cash burn), Tenant shall have the right with respect to each Reduction Date to reduce the Security Deposit to the amount of the Security Deposit set forth below as of each Reduction Date. For purposes of example only, assume for the particular fiscal year of Tenant that Tenant's Financial Statements reflect liquidity equal to \$200,000,000 (based on \$155,000,000 in market securities and \$45,000,000 in cash) and cash burn from operations equal to \$10,000,000, and there remain ten (10) years in the Lease Term. In such event, (A) the total cash burn would equal \$100,000,000 (i.e., \$10M cash burn from operations x 10 remaining years in Lease Term), and (B) Tenant's then current Liquidity would exceed the total cash burn by \$100,000,000 (i.e. \$200M Liquidity \$100M cash burn), such that, provided no Event of Default then exists under this Lease, Tenant would be entitled to the applicable reduction in Security Deposit. The following chart reflects the potential dates on which a reduction in the amount the Security Deposit may occur, and the resulting required amount of the Security Deposit in the event of the applicable reduction:

Reduction Date,
being later of (A) 30 days after Audited Financial Statements are delivered to Landlord for fiscal year ending immediately prior to start of applicable Lease Year below or (B) first day of applicable Lease Year below:

Fourth Lease Year

Sixth Lease Year

\$ 400,000

Sixth Lease Year

\$ 300,000

Eighth Lease Year

\$ 200,000

If all of the aforesaid conditions are met, within ten (10) business days after Landlord's receipt of Tenant's written request certifying that all conditions to the applicable reduction have been met, Landlord shall notify the issuer of the Letter of Credit that the Letter of Credit shall be reduced in the amount of the reduction so authorized and the security deposit shall be so reduced in accordance with this Section 5.1(d); provided, however, that in no event shall the security deposit be reduced to an amount that is less than Two Hundred Thousand Dollars (\$200,000). Such reduction shall occur by means of delivery by Tenant to Landlord of an amendment to the Letter of Credit reducing the amount thereof as directed by Landlord, or a substitute Letter of Credit in such amount and in strict conformity with the terms of this Article V, in which latter event, the original Letter of Credit will be promptly returned to Tenant. Substitutions of the Letter of Credit shall be made in a manner such that at all times one of the Letters of Credit in the required amount of the security deposit is in full force and effect and may be drawn upon (as permitted hereby). If Tenant does not qualify for a reduction as of any applicable Reduction Date due to Tenant's failure to satisfy either of the conditions in clauses (x) and (y) of this Section 5.1(d) as of such Reduction Date, but Tenant subsequently does qualify for such reduction, then such reduction shall be deferred to the first day of the Lease Year following the Lease Year in which Tenant so qualifies, and all further scheduled reductions shall be deferred by one year $(\underline{e},\underline{q},\underline{q})$ if Tenant does not qualify for a reduction on the first day of the sixth (6^{th}) Lease Year, but Tenant does so qualify upon its delivery of financial statements during the seventh (7th) Lease Year, then the reduction in the amount of the Security Deposit that otherwise would have occurred during the sixth (6th) Lease Year shall occur instead on the first day of the eighth (8th) Lease Year, and the reduction in the amount of the Security Deposit that was scheduled for the first day of the eighth (8th) Lease Year shall be deferred until the first day of the ninth (9th) Lease Year (subject to satisfaction of the terms and conditions herein). In no event shall the amount of the Letter of Credit be reduced unless the issuing bank receives prior written notice from Landlord, authorizing a reduction by a certain amount (it being understood that in no event shall the reduction exceed the amount so authorized by Landlord). Notwithstanding anything in this Article to the contrary, if two (2) or more monetary Events of Default have occurred (whether or not the same are later cured), then there shall occur no further reduction in the security deposit.

5.2 In the event of the sale or transfer of Landlord's interest in the Building, Landlord shall have the right to transfer the Security Deposit to the purchaser or assignee.

If Landlord transfers the Security Deposit to a purchaser or assignee, Tenant shall look only to such purchaser or assignee for the return of the Security Deposit, and Landlord shall thereupon be released from all liability to Tenant for the return of the Security Deposit. Tenant shall, at Landlord's sole expense, within ten (10) days after Landlord's request therefor, have the Letter of Credit amended or reissued by the issuing bank to indicate the new beneficiary.

5.3 Tenant hereby acknowledges that Tenant will not look to the holder of any mortgage now or hereafter encumbering the Building for return of the Security Deposit if such holder, or its successors or assigns, shall succeed to the ownership of the Building, whether by foreclosure or deed in lieu thereof, except if and to the extent the Security Deposit is actually transferred to such holder.

ARTICLE VI USE OF PREMISES

6.1 (a) Tenant shall use and occupy the Premises solely for general office use and for no other use or purpose. Tenant shall not use or occupy the Premises for any unlawful purpose or in any manner that will constitute waste, nuisance or unreasonable annoyance to Landlord or other tenants of the Building. Tenant shall comply with all Legal Requirements concerning the use, occupancy or condition of the Premises and all machinery, equipment and furnishings therein, including, but not limited to applicable Environmental Law (as defined in Section 6.3), the Americans with Disabilities Act and regulations promulgated from time to time thereunder. Tenant's compliance with Legal Requirements shall include, but not be limited to, permitting employees, agents or contractors of any governmental or quasi-governmental agency access to the Premises in connection with public safety issues or any Legal Requirement. If any Legal Requirements require an occupancy or use permit, license or other authorization for the Premises or the operation of the business conducted therein, then Tenant shall obtain and keep current such permit, license or authorization at Tenant's expense and shall promptly deliver a copy thereof to Landlord. It is expressly understood that if any change in the use of the Premises by Tenant, or any alterations to the Premises by Tenant, or any future Legal Requirements require a new or additional permit from, or approval by, any governmental agency having jurisdiction over the Building, such permit or approval shall be obtained by Tenant on its behalf and at its sole expense. Further, Tenant shall comply with all Legal Requirements which shall impose a duty on Landlord or Tenant relating to or as a result of the use or occupancy of the Premises. Tenant shall pay all fines, penalties and damages that may arise out of or be imposed on Landlord or Tenant because of Tenant's or an Invitee's (as defined in Section 8.2) failure to comply with applicable Legal Requirements or the provisions of this Lease.

(b) Subject to Tenant's obligations under Article IV of this Lease, Landlord shall comply in all material respects with all Legal Requirements, including without limitation, fire/life safety regulations and the Americans with Disabilities Act ("ADA"), that are applicable to the structural components of the Building (including such

structural components located within the Premises), the Base Building systems, the machinery and equipment provided by Landlord in the operation of the Building, and the operation of the common and public areas of the Building which are within Landlord's sole and exclusive control, including but not limited to, Building standard restrooms on any floor of the Building leased entirely by Tenant, but expressly excluding the elevator lobby on any floor of the Building leased entirely by Tenant; provided, however, that if Tenant makes any change to the Base Building, including without limitation, the Base Building restrooms (whether as part of the initial build out or as a subsequent Alteration), then compliance with Legal Requirements with respect to such changes shall be the sole responsibility and at the sole expense of Tenant. Notwithstanding anything to the contrary in this Section 6.1, all additions, replacements or alterations to the Building (other than the Premises) which are required due to the enactment of any Legal Requirements on or after the Effective Date shall be performed by Landlord and the cost thereof shall be an Operating Expense (if and to the extent permitted in accordance with Article IV of this Lease) unless such addition, replacement or alteration is necessitated by Tenant's particular use, design or layout of the Premises (but not to the extent solely arising out of Tenant's use for general office use) or caused by Tenant or any of its employees, agents, contractors or subtenants, in which case (i) Tenant shall pay its proportionate share of the costs of performing such addition, replacement or alteration if and to the extent Landlord is permitted to pass through such costs to Tenant as an Operating Expense pursuant to Article IV above, and (ii) Tenant shall bear the entire cost of performing such addition, replacement or alteration if and to the extent Landlord is not permitted to pass through such costs to Tenant as an Operating Expense pursuant to Article IV above.

6.2 Tenant shall pay any business, rent or other taxes that are now or hereafter levied upon Tenant's use or occupancy of the Premises, the conduct of Tenant's business at the Premises, or Tenant's equipment, fixtures or personal property. In the event that any such taxes are enacted, changed or altered so that any of such taxes are levied against Landlord or the mode of collection of such taxes is changed so that Landlord is responsible for collection or payment of such taxes, Tenant shall pay any and all such taxes to Landlord upon written demand from Landlord.

6.3 Tenant shall not cause or permit any Hazardous Materials (as defined below) to be generated, used, released, stored, disposed, or abandoned in, on, under or about the Premises, Building, or the Land provided that Tenant may use and store in the Premises such quantities of standard cleaning and office materials as may be reasonably necessary for Tenant to conduct normal general office use operations in the Premises, but only to the extent that such materials are used, stored, and disposed by Tenant in compliance with Environmental Law. At the expiration or earlier termination of this Lease, Tenant shall surrender the Premises to Landlord free of Hazardous Materials and in compliance with all Environmental Laws. "Hazardous Materials" means any of the following and any substance or material that contains any of the following: (a) asbestos, asbestos containing materials, and presumed asbestos containing materials; (b) oils, petroleum, petroleum products and by-products, drilling fluids, produced waters, and other wastes associated with the exploration, development, or production of crude oil, natural gas, or geothermal resources; (c) polychlorinated biphenyls, urea formaldehyde, radon gas, radioactive

materials (including any source, special nuclear, or byproduct material), medical waste, chlorofluorocarbons, lead or lead-based products, and any other substance whose presence could be detrimental to the Premises, Building, or the Land or to health or the environment and (d) any substance that is then defined or listed in, or otherwise classified pursuant to, any Environmental Law or any other applicable Legal Requirements as a "hazardous substance," "hazardous material," "hazardous waste," infectious waste," "toxic substance," "toxic pollutant," or any other formulation intended to define, list, or classify substances by reason of deleterious properties such as ignitability, corrosivity, reactivity, carcinogenicity, toxicity, and reproductive toxicity. "Environmental Law" means any present and future law and any amendments thereto (whether common law, statute, rule, order, regulation or otherwise), permits, directives, and other requirements of governmental authorities applicable to the Premises, the Building or the Land and relating to the environment, environmental conditions, health, safety, or to any Hazardous Material, including, without limitation, the Comprehensive Environmental Response, Compensation and Liability Act, 42 U.S.C. § 9601 et seq., the Resource Conservation and Recovery Act of 1976, 42 U.S.C. § 6901 et seq., the Hazardous Materials Transportation Act, 49 U.S.C. § 5101 et seq., the Federal Water Pollution Control Act, 33 U.S.C. § 1251 et seq., the Clean Air Act, 42 U.S.C. § 7401 et seq., the Toxic Substances Control Act, 15 U.S.C. § 2601 et seq., the Safe Drinking Water Act, 42 U.S.C. § 300f et seq., the Emergency Planning and Community Right-To-Know Act, 42 U.S.C. § 11001 et seq., the Occupational Safety and Health Act, 29 U.S.C. § 651 et seq., any so-called "Super Fund" or "Super Lien" law, any Legal Requirements requiring the filing of reports or notices relating to Hazardous Materials, and any similar state and local laws, all amendments thereto, and all regulations, orders, decisions

6.4 Notwithstanding the expiration or early termination of this Lease, Tenant shall release, indemnify and hold harmless Landlord, its affiliates, employees, agents and contractors, in accordance with the applicable terms of Section 13.1 below, from and against any damage, injury, loss, liability, violation, charge, demand or claim (including reasonable attorneys' fees, consultants' fees, and any costs of litigation) based on, arising out of, or related to: (a) the actual or alleged release, presence, removal, or failure to remove, of Hazardous Materials generated, used, released, stored, disposed, or abandoned by Tenant or its employees, agents or contractors, in, on, under or about the Premises, the Building, or the Land whether before or after the Effective Date, (b) any violation of Environmental Law by Tenant or its employees, agents or contractors, or (c) any investigation, assessment, removal, cleanup, abatement, or other corrective action taken with respect to the use or occupancy of the Premises by Tenant or its employees, agents or contractors. In addition, Tenant shall give Landlord immediate verbal and follow-up written notice of any actual Environmental Default, or any threatened Environmental Default of which Tenant has knowledge, which Environmental Default in any event Tenant shall cure in compliance with all Environmental Law and to the satisfaction of Landlord. An "Environmental Default" means any of the following by Tenant or its employees, agents or contractors relating to the Premises, the Building or the Land: (x) a violation of Environmental Law; (y) a release, spill or discharge of Hazardous Materials whether or not required to be reported under Environmental Law; or (z) an environmental condition

requiring responsive action, whether or not the condition presents an emergency. Upon any Environmental Default, in addition to all other rights available to Landlord under this Lease, at law or in equity, Landlord shall have the right but not the obligation to immediately enter the Premises, to supervise and direct actions taken by Tenant to address the Environmental Default, and, if Tenant fails to promptly address same to Landlord's satisfaction, to perform, at Tenant's sole cost and expense, any action Landlord deems necessary to address same. If any lender or governmental agency shall require an assessment, testing or other action to ascertain whether an Environmental Default is pending or threatened, then Tenant shall pay the reasonable costs therefor as additional rent. Promptly upon request, if reasonably requested by Landlord or its mortgagee, Tenant shall execute commercially reasonable forms, affidavits, representations and similar documents concerning Tenant's best knowledge and belief regarding compliance with Environmental Law and the presence, use, storage, and disposal of Hazardous Materials in, on, under or from the Premises, the Building and the Land by Tenant.

6.5 Landlord represents that, as of the Effective Date, Landlord has provided to Tenant all of the environmental reports in its possession or control regarding the Building and the Land ("Environmental Reports"). A list of such reports is attached hereto as Exhibit I. Landlord shall deliver the Premises to Tenant in compliance with all applicable Environmental Law. Landlord shall not knowingly permit the use of any Hazardous Materials in violation of any Environmental Law in the construction or development of the Building or the Project. From and after the Effective Date and continuing throughout the Lease Term, in the event Landlord is advised, or it shall come to Landlord's attention, that Hazardous Materials exist in the Building or in, on, or about the Land in violation of Environmental Law, then Landlord shall take all reasonable steps necessary to abate, encapsulate, manage, or remove, at Landlord's expense, all such Hazardous Materials to the extent mandated by Environmental Law, and in doing so, Landlord shall use its commercially reasonable efforts not to materially interfere with the conduct of Tenant's business; provided, however, that Landlord shall be permitted (but not required) to remove, at Tenant's expense, any Hazardous Materials from the Premises, the Building or the Land which Tenant, its employees, agents, subcontractors or subtenants shall have introduced or otherwise brought in, on or about the Premises, the Building or the Land that result in an Environmental Default. Notwithstanding anything herein to the contrary, no holder of any mortgage (nor any person or entity claiming by, through or under any such holder) shall have any liability under this Section 6.5 or any responsibility under this Section 6.5 to perform any of Landlord's obligations set forth in this Section 6.5 (other than to deliver to Tenant the Premises in compliance with applicable Environmental Law).

ARTICLE VII ASSIGNMENT AND SUBLETTING

7.1 Assignments.

(a) Tenant shall not have the right to assign (in whole or in part) this Lease or its interest in this Lease without the prior written consent of Landlord, which

consent shall not be unreasonably withheld, conditioned or delayed, except as otherwise specifically provided in this Article. Without limiting any other instances in which it may not be unreasonable for Landlord to withhold its consent to an assignment, it shall not be unreasonable for Landlord to withhold its consent to an assignment for any of the reasons set forth in Section 7.2 below. The following transactions will be deemed assignments for purposes of this Section 7.1 and will require Landlord's prior written consent in accordance with and subject to the provisions of this Section 7.1 and the other applicable provisions of this Article VII: (i) an assignment by operation of law; (ii) an imposition (whether or not consensual) of a lien, mortgage, or other encumbrance upon Tenant's interest in this Lease; (iii) if Tenant is a partnership or a limited liability company, a withdrawal or change, whether voluntary, involuntary or by operation of law, of partners or members owning, individually or collectively, a controlling interest in Tenant (occurring in one transaction or in a series of related transactions); (iv) if Tenant is a corporation, any dissolution, merger, consolidation or other reorganization of Tenant, or the sale or transfer of a controlling interest of the capital stock of Tenant (occurring in one transaction or in a series of related transactions), except that this clause will not apply to corporations, the stock of which is traded through a national or regional stock exchange; and (v) if Tenant is a general partnership, conversion of Tenant from a general partnership to a limited liability partnership. Any attempted assignment of this Lease or Tenant's interest in this Lease without Landlord's prior written consent shall, at the option of Landlord, terminate this Lease; however, in the event of such termination, Tenant shall remain liable for all rent and other sums due under this Lease and all damages suffered by Landlord on account of such breach by Tenant.

(b) In the event of any assignment of this Lease, Tenant shall remain fully liable as a primary obligor and principal for Tenant's obligations under this Lease, including, without limitation, the payment of all rent and other sums or charges required hereunder. The limitations in this Section 7.1 shall be deemed to apply to any subtenant(s), assignee(s) and guarantors(s) of this Lease.

7.2 Subleases.

(a) Tenant shall not have the right to sublease (which term, as used herein, shall include any type of subrental arrangement and any type of license to occupy) all or any part of the Premises without the prior written consent of Landlord, which consent shall not be unreasonably withheld, conditioned or delayed. Without limiting the other instances in which it may not be unreasonable for Landlord to withhold its consent to a sublease, it shall not be unreasonable for Landlord to withhold its consent in any one of the following instances: (i) there exists a monetary default or non-monetary Event of Default by Tenant under this Lease as of the effective date of the sublease; (ii) in the case of subletting of less than the entire Premises, if the subletting would result in the division of the Premises into more than three (3) subparcels, would require access be provided through space leased or held for lease to another tenant, or improvements be made outside of the Premises; (iii) the sublease is prohibited by Landlord's lender (except for any Affiliate Transaction permitted pursuant to Section 7.4 below); (iv) Landlord determines, in its reasonable discretion, that the character, reputation or business of the proposed

subtenant would adversely affect the other tenants of the Building or would impair the reputation of the Building as a Class A office and retail building; (v) [intentionally omitted]; (vi) the financial capacity or credit rating of the proposed subtenant is unacceptable to Landlord in its reasonable discretion, based on the obligations of Tenant under this Lease for the remainder of the Lease Term; (vii) the proposed sublease may have an adverse effect on the real estate investment trust qualification tests applicable to Landlord and its affiliates; (viii) the proposed sublease raises unrelated business taxable income concerns for the holder of the mortgage on the Building; (ix) the use of the Premises by the proposed subtenant will violate any provisions or restrictions contained in this Lease, including but not limited to, any relating to the use or occupancy of the Premises; or (x) the business to be conducted or the proposed use of the Premises by the proposed subtenant is likely to increase Operating Expenses beyond that which Landlord incurs prior to such proposed subletting, or is likely to increase the burden on elevators or other Building systems or equipment over the burden prior to such proposed subletting; provided, however, that such determination shall not be based solely on an increase in headcount that is within the permitted occupancy load for the portion of the Premises to be sublet, and in any event, such proposed sublease shall not be prohibited on the basis of this clause (ix) if Tenant agrees to be responsible for such increase, whether financial or otherwise. Any attempted subletting by Tenant of any portion of the Premises without Landlord's prior written consent shall, at the option of Landlord, terminate this Lease; however, in the event of such termination, Tenant shall remain liable for all rent and other sums due under this Lease and all damages suffered by Landlord on account of such breach by Tenant. Furthermore, Tenant shall not have the right to sublease all or any portion of th

(b) In the event of any sublease, Tenant shall remain fully liable as a primary obligor and principal for Tenant's obligations under this Lease, including, without limitation, the payment of all rent and other sums required hereunder.

7.3 (a) Tenant shall give Landlord written notice of its desire to sublease all or a portion of the Premises ("Tenant's Sublease Notice"). Tenant's Sublease Notice shall specify the portion of the Premises proposed to be sublet ("Proposed Sublease Premises") and the date on which the Proposed Sublease Premises will be made available for subleasing. If (i) the Proposed Sublease Premises is (or, when aggregated with all other space then being subleased by Tenant, will be) more than fifty percent (50%) of the rentable area of the Premises, and/or (ii) the term of the sublease for the Proposed Sublease Premises is for ninety percent (90%) or more of the remaining Lease Term as of the commencement date of the sublease for the Proposed Sublease Premises, then within thirty (30) days after receipt of the Tenant's Sublease Notice, Landlord shall notify Tenant in writing whether or not Landlord will retake possession of all or any portion of the Proposed Sublease Premises and thereby terminate this Lease with respect to such portion Landlord elects to retake. If Landlord elects to retake all or any portion of the Proposed Sublease Premises, then (1) Landlord shall retake possession of such portion on the date specified in the Tenant's Sublease Notice or such other date mutually agreed upon by Landlord and Tenant, (2) Tenant's obligation to pay rent for such portion shall cease on such date, and (3) Landlord and Tenant shall promptly execute an amendment to this Lease setting forth the new square footage of the reduced Premises to be

occupied by Tenant. Thereafter, Tenant shall not have any further rights of any kind, including any rights of renewal, in or to the portion of the Premises so retaken. If the Proposed Sublease Premises constitutes less than the entire Premises, Tenant shall cause to be constructed and installed, at Tenant's sole cost and expense, a demising wall separating the Proposed Sublease Premises from the remaining Premises in accordance with all applicable Legal Requirements, except that, at Landlord's option, Landlord shall cause such demising wall to be constructed and installed at Tenant's cost and expense. If Landlord does not elect to retake all or any portion of the Proposed Sublease Premises within the aforesaid thirty (30) day period, Tenant shall comply with the provisions of Subsections (b) through (e) below with respect to any proposed sublease of such portion of the Premises.

- (b) Subject to the requirements of Section 7.2 hereof, Tenant shall have the right to sublease any portion of the Proposed Sublease Premises that Landlord has not elected to retake pursuant to Subsection 7.3 (a) above ("Eligible Sublease Premises").
- (c) Tenant's right to sublease the Eligible Sublease Premises shall expire one hundred eighty (180) days after the date of the Tenant's Sublease Notice. Thereafter, Tenant shall have no right to sublease the Eligible Sublease Premises unless Tenant shall have again complied with the procedures set forth in this Section 7.3.
- (d) Provided there does not exist an Event of Default by Tenant under this Lease, Tenant shall be entitled to retain fifty percent (50%) of any Profit Derived From Subletting the Premises (hereinafter defined) or any part thereof. "**Profit Derived From Subletting the Premises**" shall mean any and all sums paid to Tenant pursuant to any sublease (other than the fair market value consideration for furniture and equipment) that exceed the base rent and additional rent due under this Lease for such portion of the Premises sublet (but shall not include any period of vacancy), less all reasonable out-of-pocket third-party costs and expenses actually incurred by Tenant in connection with such subletting, including, but not limited to, brokerage commissions, reasonable attorneys' fees, improvements to the Premises and reasonable advertising expenses. For any period during which there exists an Event of Default by Tenant under this Lease, Landlord shall be entitled to one hundred percent (100%) of the rent due from any subtenant of Tenant and Tenant shall provide written notice to each subtenant to pay said rent directly to Landlord. Landlord shall have the right to inspect and audit Tenant's books and records relating to any sublease and expenses incurred by Tenant in connection therewith.
- (e) If Tenant requests Landlord's consent to a sublease to a specific subtenant, Tenant will give Landlord at the time of its request, which must be in writing ("Sublease Consent Request"), reasonably sufficient information about the proposed subtenant to enable Landlord to make the determination called for by Section 7.2 hereof, including, without limitation, the following information (collectively, the "Subtenant Information"): (i) the name and address of the proposed subtenant, (ii) a copy of the proposed sublease, (iii) reasonable information about the nature, business, and business history of the proposed subtenant, and its proposed use of the Premises, and (iv) audited financial statements and/or such other banking, financial or other credit information. Provided Tenant provides the Subtenant Information and such other information reasonably requested by

Landlord, Landlord agrees to advise Tenant of its decision to grant or withhold its consent to such subletting within [thirty (30)] days after Landlord's receipt of the Sublease Consent Request and the Subtenant Information.

7.4 Affiliate Transactions.

(a) Notwithstanding the above restrictions on subletting and assignments, Landlord's prior consent shall not be required with respect to any assignment or subletting to an "Affiliate of Tenant" (as hereinafter defined) or a "Parent of Tenant" (as hereinafter defined), or to any entity resulting from the merger, consolidation, or other corporate reorganization of Tenant, or the sale or transfer of all or substantially all of the assets, capital stock, partnership interests, membership interests or other ownership interests of Tenant (a "Successor to Tenant"), provided that Tenant delivers to Landlord not more than ninety (90) days and not less than thirty (30) days prior to the effective date of such assignment or subletting a written certification (together with reasonable back-up information to support such certification, including, without limitation, certified financial statements) that the following conditions are satisfied: (i) that such Affiliate of Tenant, Parent of Tenant or Successor to Tenant, as applicable, has (x) a net worth (which shall be determined on a pro forma basis using generally accepted accounting principles consistently applied and using the most recent financial statements) at least equal to the lesser of (I) the net worth of Tenant as of the Effective Date and (II) the net worth of Tenant immediately prior to the effective date of such proposed assignment or sublease, and (y) then-current Liquidity equal to or greater than the sum of Tenant's remaining obligations under this Lease; provided, however, that such financial requirements shall not be applicable solely with respect to a sublease to an Affiliate of Tenant; (ii) that such Affiliate of Tenant, Parent of Tenant or Successor to Tenant, as applicable, agrees in writing to be bound by the terms and conditions of this Lease and to assume all of the obligations of Tenant under this Lease; (iii) that such Affiliate of Tenant, Parent of Tenant or Successor to Tenant, as applicable, shall conduct substantially the same business on the Premises as that conducted by Tenant or a related business which is a permitted use pursuant to Article VI of this Lease; (iv) that the character of such Affiliate of Tenant, Parent of Tenant or Successor to Tenant, as applicable, and the nature of its activities in the Premises and in the Building will not adversely affect other tenants in the Building or impair the reputation of the Building as a Class A office and retail building; and (v) that the assignment or sublease is not a so-called "sham" transaction intended by Tenant to circumvent the provisions of this Article VII. Any assignment or subletting that is permitted pursuant to this Section 7.4 shall be referred to herein as an "Affiliate Transaction" and any Affiliate of Tenant, Parent of Tenant or Successor to Tenant in connection with an Affiliate Transaction shall be referred to as a "Permitted Transferee."

(b) In the event of any such assignment or subletting pursuant to this Section 7.4, Tenant shall remain fully liable as a primary obligor and principal for Tenant's obligations under this Lease, including without limitation, the payment of all rent and other sums required hereunder.

(c) For purposes of this Section 7.4, an "Affiliate of Tenant" shall mean any corporation, association, trust, limited liability company or partnership (i) which Controls (as herein defined) Tenant, or (ii) which is under the Control of Tenant through stock ownership or otherwise, or (iii) which is under common Control with Tenant. For the purposes of this Section 7.4, a "Parent of Tenant" shall mean any corporation, association, trust, limited liability company or partnership which Controls Tenant, or which owns more than fifty percent (50%) of the issued and outstanding voting securities or other ownership interests of Tenant. The terms "Control" or "Controls" as used in this Section 7.4 shall mean the power directly or indirectly to influence the direction, management or policies of Tenant or such other entity.

7.5 General Provisions.

- (a) Landlord's consent to an assignment or sublease will not be effective until (i) a fully executed copy of the instrument of assignment or sublease has been delivered to Landlord, and the form and substance of the instrument has been approved by Landlord, which approval shall not be unreasonably withheld, conditioned or delayed; (ii) in the case of an assignment, Landlord has received a written instrument in which the assignee has assumed and agreed to perform all of Tenant's obligations under this Lease; and (iii) Landlord has been reimbursed for the costs pursuant to Section 7.5(f) hereof.
- (b) The consent by Landlord to any assignment or subletting shall not be construed as a waiver or release of Tenant from any and all liability for the performance of all covenants and obligations to be performed by Tenant under this Lease.
- (c) Landlord's collection or acceptance of rent from any assignee, transferee or subtenant shall not constitute a waiver or release of Tenant from any of its obligations under this Lease.
- (d) Notwithstanding the provisions of Section 7.1 or 7.2 hereof to the contrary, if consent to any assignment or subletting is required by the holder of any mortgage on the Building, no assignment of this Lease in whole or in part or sublease of all or any portions of the Premises shall be permitted without the prior written consent of such holder (except for any Affiliate Transaction permitted pursuant to Section 7.4 above).
- (e) Landlord's consent to any one assignment or subletting will not waive the requirement of its consent to any subsequent assignment or subletting.
- (f) Tenant shall reimburse Landlord for all reasonable third-party costs incurred by Landlord in connection with any request by Tenant to sublease all or any portion of the Premises or to assign this Lease or its interest therein, plus an administrative fee of One Thousand and 00/100 Dollars (\$1,000.00) per request (whether or not Landlord's consent thereto is granted).

(g) Tenant shall require its subtenants and other occupants of the Premises to comply with the applicable provisions of Article XIII below, including without limitation, Sections 13.1(d), 13.9 and 13.13 thereof.

ARTICLE VIII TENANT'S MAINTENANCE AND REPAIRS

8.1 Tenant will keep and maintain the Premises and all fixtures and equipment located therein in clean, safe and sanitary condition, will take good care thereof and make all required repairs thereto, and will suffer no waste or injury thereto, all in a manner consistent with a Class A office and retail Building. Tenant acknowledges the importance of maintaining a uniform and attractive appearance in all areas of the Premises that are visible from: (i) common or public areas of the Building; (ii) the lobby areas serving the Building; (iii) other tenant premises; and (iv) the exterior of the Building, and agrees to comply with all rules established from time to time by Landlord in connection therewith. At the expiration or earlier termination of this Lease, Tenant shall surrender the Premises, broom clean, in the same order and condition in which they are in on the Lease Commencement Date, ordinary wear and tear and unavoidable damage by the elements excepted. Landlord shall provide and install (subject to reimbursement in accordance with Article IV) replacement tubes and bulbs for Building standard light fixtures in the Premises, if any; all other bulbs and tubes for the Premises shall be Tenant's responsibility, however, at Tenant's request, Landlord shall stock and install such other bulbs and tubes and Tenant shall reimburse Landlord for its costs and expenses incurred in connection with said stocking and installation.

8.2 Except as otherwise provided in Section 13.13 and Article XVII hereof, all injury, breakage and damage to the Premises and to any other part of the Building or Project caused by any act or omission of Tenant, or of any agent, employee, subtenant, contractor, customer, client, licensee, family member, guest or other invitee of Tenant (each, an "Invitee" or, collectively, "Invitees"), shall be repaired by and at the sole expense of Tenant, except that Landlord shall have the right, at its option, to make such repairs and to charge Tenant for all costs and expenses incurred in connection therewith as additional rent hereunder. The liability of Tenant for such costs and expenses shall be reduced by the amount of any insurance proceeds received by Landlord on account of such injury, breakage or damage. Landlord shall not be deemed to be an Invitee of Tenant.

8.3 Landlord shall keep and maintain the exterior and demising walls, foundations, floor slabs (other than any deflection thereof), exterior pane of window glass, roof and common areas that form a part of the Building, the Office Parking Area, and the Building standard heating, ventilation and air conditioning, mechanical, electrical and plumbing systems, pipes and conduits that are provided by Landlord in the operation of the Building or, on a non-exclusive basis, to the Premises, in clean, safe, sanitary and operating condition in accordance with standards customarily maintained by Class A office and retail buildings in the central business district, west end and east end submarkets of Washington, D.C. ("Market Area") and will make all required repairs thereto. All

common or public areas of the Building and the land upon which it is situated (including without limitation the first floor lobby area and the exterior landscaping) shall be maintained by Landlord in accordance with standards customarily maintained by Class A office and retail buildings in the Market Area. Tenant shall promptly provide Landlord with written notice of any defect or need for repairs in or about the Building of which Tenant is aware, and Landlord shall perform such repair after notice by Tenant, if and to the extent expressly the obligation of Landlord under this Lease. Notwithstanding any of the foregoing to the contrary: (a) maintenance and repair of special tenant areas, facilities, finishes and equipment (including, but not limited to, any special fire protection equipment, telecommunications and computer equipment, kitchen/galley equipment, or internal staircase(s) which may be installed by or at the request of Tenant, supplemental air-conditioning equipment serving the Premises only and all other furniture, furnishings and equipment of Tenant and all Alterations) shall be the sole responsibility of Tenant and shall be deemed not to be a part of the Building structure and systems; and (b) Landlord shall have no obligation to make any repairs brought about by any act or neglect of Tenant or any Invitee.

ARTICLE IX TENANT ALTERATIONS

- 9.1 The initial improvements shall be constructed in the Premises in accordance with <u>Exhibit B</u> attached hereto and made a part hereof. It is understood and agreed that Landlord will not make, and is under no obligation to make, any structural or other alterations, decorations, additions or improvements in or to the Premises, except as provided in this Section 9.1.
- 9.2 (a) Tenant will not make or permit anyone to make any alterations, decorations, additions or improvements (hereinafter referred to collectively as "improvements" or "Alterations") in or to the Premises or the Building, without the prior reasonable written consent of Landlord; provided however Landlord's sole consent shall be required for any Alteration which is deemed to be a Structural Alteration. "Structural Alterations" shall be any Alterations that (i) will or may necessitate any changes, replacements or additions to columns or floors or other structural elements of the Building; (ii) are readily visible to the exterior of the Building, or the common and public areas thereof, or the main lobby of the Building, (iii) adversely affect the base building systems of the Building or the roof of the Building, or (iv) would have a negative impact on any building warranty.

 Notwithstanding the foregoing, provided Tenant gives Landlord reasonable prior written notice thereof, Tenant may install in the Premises, without obtaining Landlord's prior written consent, minor, nonstructural Alterations of a decorative nature ("Cosmetic Alterations") the value of which (as reasonably determined by Landlord) is less than Fifty Thousand Dollars (\$50,000) and which do not require a building permit, for example, the hanging of artwork or the installation of carpeting.
- (b) Any Alterations made by Tenant shall be made: (i) in a good, workmanlike, first-class and prompt manner and otherwise in accordance with Landlord's

rules, including any rules for contractors, that may be established by Landlord from time to time; (ii) using new or sustainable materials only; (iii) by a contractor, on days, at times and under the supervision of an architect approved in writing by Landlord; (iv) after coordinating the work schedule and scope with the Building's property manager to avoid undue interference with the normal operations and use of the Building; (v) in accordance with plans and specifications prepared by an engineer or architect reasonably acceptable to Landlord, which plans and specifications shall be approved in writing by Landlord, and Tenant shall reimburse Landlord for all reasonable, third-party, out-of-pocket costs (if any) incurred by Landlord in connection therewith; (vi) in accordance with all Legal Requirements, Insurance Requirements (as defined below) and the requirements of the Underwriters' Association of the District of Columbia; (vii) after having obtained any required consent of the holder of any mortgage (if any such consent is required); and (viii) after obtaining public liability and worker's compensation insurance policies in accordance with the terms and conditions of Article XIII approved in writing by Landlord, which policies shall cover every person who will perform any work with respect to such Alteration.

(c) Prior to each payment to any contractor, subcontractor, laborer, or material supplier for all work, labor, and services to be performed and materials to be furnished in connection with Alterations, Tenant shall obtain and deliver to Landlord written waivers of mechanics' and materialmen's liens against the Premises and the Building from all contractors, subcontractors, laborers and material suppliers for all work, labor and services performed and materials furnished in connection with Alterations through the date of the then-current requisition, conditioned only on payment of the amount requisitioned. If any lien (or a petition to establish such lien) is filed in connection with any Alteration, such lien (or petition) shall be discharged by Tenant within ten (10) days thereafter, at Tenant's sole cost and expense, by the payment thereof or by the filing of a bond acceptable to Landlord. If Landlord gives its consent to the making of any Alteration, such consent shall not be deemed to be an agreement or consent by Landlord to subject its interest in the Premises or the Building to any liens which may be filed in connection therewith. If Tenant shall fail to discharge any such mechanic's or materialmen's lien, Landlord may, at its option, discharge such lien and treat the cost thereof (including reasonable attorneys' fees incurred in connection therewith) as additional rent payable with the next monthly installment of annual base rent falling due; it being expressly agreed that such discharge by Landlord shall not be deemed to waive or release the default of Tenant in not discharging such lien. It is understood and agreed that any improvements to the Premises shall be conducted on behalf of Tenant and not on behalf of Landlord, and that Tenant shall be deemed the "owner" of such improvements (and not the agent of Landlord) for purposes of the application of District of Columbia lien laws.

(d) All Alterations involving tie-ins to the Building's fire and life safety systems, changes and modifications to the Building's exterior envelope (roof, glass, glazing, etc.), or any other item affecting a warranty shall, at Landlord's election, be performed by Landlord's designated contractor or subcontractor at Tenant's expense.

- (e) Tenant's contractor shall use light sensors in connection with performance of the Alterations.
- (f) Promptly after the completion of an Alteration, Tenant at its expense shall deliver to Landlord one (1) set of accurate as-built drawings and one (1) AutoCAD computer disc showing such Alteration in place.
- (g) When granting its consent, Landlord may impose any reasonable conditions it deems appropriate, including, without limitation, the approval of plans and specifications, approval of the contractor or other persons who will perform the work, and the obtaining of required permits and specified insurance. Portions of the Premises visible to the public shall maintain a uniform appearance with the rest of the Building. Landlord's review and approval of any such plans and specifications and its consent to perform work described therein shall not be deemed an agreement by Landlord that such plans, specifications and work conform with all applicable Legal Requirements and requirements of the insurers of the Building ("Insurance Requirements") nor deemed a waiver of Tenant's obligations under this Lease with respect to all applicable Legal Requirements and Insurance Requirements nor impose any liability or obligation upon Landlord with respect to the completeness, design sufficiency or compliance with all applicable Legal Requirements or Insurance Requirements of such plans, specifications and work.
- (h) Tenant acknowledges and agrees that Landlord shall be the owner of any additions, alterations and improvements in the Premises or the Building to the extent paid for by Landlord.
- 9.3 Tenant shall indemnify and hold Landlord harmless from and against any and all expenses, liens, claims, liabilities and damages based on or arising, directly or indirectly, by reason of the making of any improvements to the Premises by Tenant, or its contractors, agents or employees. If any improvements (other than Cosmetic Alterations) are made without the prior written consent of Landlord, Landlord shall have the right to remove and correct such improvements and restore the Premises to their condition immediately prior thereto, and Tenant shall be liable for all expenses incurred by Landlord in connection therewith. All improvements to the Premises or the Building made by either party shall remain upon and be surrendered with the Premises as a part thereof at the end of the Lease Term, unless Landlord specifies in its approval of the plans and specifications for such improvements that Tenant must remove the improvements upon the expiration or earlier termination of the Lease Term (subject to the terms of the last sentence of this Section 9.3), except that if Tenant is not in default under this Lease, Tenant shall have the right to remove, at Tenant's sole expense, prior to the expiration of the Lease Term, all movable furniture, furnishings and equipment installed in the Premises solely at the expense of Tenant. All damages and injury to the Premises or the Building caused by such removal shall be repaired by Tenant, at Tenant's sole expense. If such property of Tenant is not removed by Tenant prior to the expiration or termination of this Lease, the same shall become the property of Landlord and shall be surrendered with the Premises as a part thereof. Notwithstanding anything to the contrary contained in this Lease, (a) any

removal and restoration that Landlord may require hereunder will be limited to above standard improvements (whether performed as part of the Leasehold Work or as a subsequent Alteration), including without limitation, supplemental HVAC units, LAN rooms, raised flooring and private lavatories, and (b) removal and restoration shall be required in all events for all voice and data cabling if required pursuant to Section 26.1(j) below, which must be bundled and identified at installation and removed in accordance with the terms of Article XXVI below.

ARTICLE X SIGNS AND FURNISHINGS

10.1 No sign, advertisement or notice referring to Tenant shall be inscribed, painted, affixed or otherwise displayed on any part of the exterior or the interior of the Building or the Project except on the directories and doors of offices and such other areas as are designated by Landlord, and then only in such place, number, size, color and style as are approved by Landlord and are in accordance with any applicable state or local building code or zoning regulations. Notwithstanding the foregoing, Tenant shall be permitted, at its sole cost and expense, to install and maintain signage identifying Tenant in the elevator lobby of the third (3rd) floor of the East Tower (provided Tenant is leasing the entire rentable area of such floor), but only in such place, number, size, color and style as are approved by Landlord, which approval shall not be unreasonably withheld, conditioned or delayed provided that such signage is consistent with signage standards in other Class A office buildings in the Market Area. All of Tenant's signs that are approved by Landlord shall, at Landlord's election, be installed by Tenant at Tenant's cost and expense and shall be removed by Tenant at Tenant's sole cost and expense at the end of the Lease Term (and Tenant shall repair any damage to the Building or the Premises caused by such removal). If any sign, advertisement or notice that has not been approved by Landlord is exhibited or installed by Tenant, Landlord shall have the right to remove the same at Tenant's expense. Landlord shall list Tenant's name and the names of its employees who work at the Premises as of the Lease Commencement Date in the Building lobby directory; provided however, that if Tenant requests Landlord to change the names on such lobby directory, then Tenant shall reimburse Landlord for all actual costs incurred by Landlord therefor. Landlord's acceptance of any name for listing on the Building directory will not be deemed, nor will it substitute for, Landlord's consent, as required by this Lease, to any sublease, assignment or other occupancy of the Premises. Landlord shall have the right to prohibit any advertisement of or by Tenant which in its opinion tends to impair the reputation of the Building or its desirability as a Class A office and retail building, and upon notice from Landlord, Tenant shall immediately refrain from and discontinue any such advertisement. Landlord reserves the right to affix, install and display signs, advertisements and notices on any part of the exterior or interior of the Building but not in the Premises except as may be required by law or in emergency situations.

10.2 Landlord shall have the right to prescribe the weight and position of safes and other heavy equipment and fixtures, which, if considered necessary by Landlord, shall

be installed in such manner as Landlord directs in order to distribute their weight adequately. Any additional structural support or upgrading of the floor supports that may be needed to accommodate any of Tenant's equipment that exceeds the floor loading specifications for the Building (i.e., 100 pounds per square foot, comprised of 80 pounds per square foot live load and 20 pounds per square foot partition load) shall be installed at Tenant's sole cost and expense and shall be subject to the prior written approval of Landlord, which approval shall be granted or withheld in Landlord's sole and absolute discretion. Any and all damage or injury to the Premises or the Building caused by moving the property of Tenant into or out of the Premises, or due to the same being in or upon the Premises, shall be repaired at the sole cost of Tenant. No furniture, equipment or other bulky matter of any description will be received into the Building or carried in the elevators except as coordinated in advance with Landlord, and all such furniture, equipment and other bulky matter shall be delivered only through the designated delivery entrance of the Building and the designated freight elevator. All moving of furniture, equipment and other materials shall be under the supervision of Landlord, who shall not, however, be responsible for any damage to or charges for moving the same. Tenant agrees to remove promptly from the sidewalks adjacent to the Building any of Tenant's furniture, equipment or other material there delivered or deposited.

ARTICLE XI TENANT'S EQUIPMENT

- 11.1 The Base Building is designed and will be constructed to, and the Base Building shall provide electrical capacity to the Premises in accordance with, Schedule I to Exhibit B. Any electrically operated equipment or machinery to be installed in the Premises as part of the initial Leasehold Work or subsequent alterations shall conform to the requirements of such Schedule I to Exhibit B and this Article XI. The following installations and operations shall require Landlord's prior written consent, which consent shall not be unreasonably withheld, conditioned or delayed, however Landlord may condition such consent upon the payment by Tenant for the cost of any separate metering or sub-metering, the cost of any additional wiring or apparatus that may be occasioned by the operation of such equipment or machinery, and additional rent in compensation for any excess consumption of electricity or other utilities:
- (a) The installation of any supplemental heating, ventilation and air conditioning equipment for the Premises, excluding the installation of additional VAV boxes without any other supplemental equipment;
- (b) The installation of lighting for the Premises that consumes, in the aggregate, more than 1.5 watts per usable square foot ("USF") of any floor of the Premises in the East Tower or the West Tower;
- (c) The use or installation in the Premises of any electrically operated equipment or machinery that operates on greater than 120/208 volt power, excluding tenant lighting and all VAV boxes; and

(d) The use or installation in the Premises of any electrically operated equipment or machinery that operates on 120/208 volt power and that consumes, in the aggregate, more than 5.0 watts per USF of any floor of the Premises in the East Tower or the West Tower, or that requires the installation of any additional electrical capacity in excess of the 5.0 watts per USF per floor of the East Tower or the West Tower provided by the Base Building. In addition to the foregoing, Tenant shall not install any equipment of any type or nature that will or may necessitate any changes, replacements or additions to, or in the use of, the Base Building water system, heating system, plumbing system, air-conditioning system or electrical system of the Premises or the Base Building, without first obtaining the prior written consent of Landlord and Landlord may require that any additional equipment be sub-metered and any excess consumption be paid for by Tenant. Business machines and mechanical equipment belonging to Tenant which cause noise or vibration that may be transmitted to the structure of the Building or to any space therein to such a degree as to be objectionable to Landlord or to any tenant in the Building shall be installed and maintained by Tenant, at Tenant's expense, on vibration eliminators or other devices sufficient to reduce such noise and vibration to a level satisfactory to Landlord.

ARTICLE XII ENTRY AND INSPECTION BY LANDLORD

12.1 Tenant shall permit Landlord, its agents or representatives, to enter the Premises, without charge therefor to Landlord and without diminution of the rent payable by Tenant, to examine, inspect and protect the Premises and the Building, to make such alterations or repairs (as in the sole judgment of Landlord may be deemed necessary), and to exhibit the same to prospective tenants at any time during the Lease Term. In connection with any such entry, Landlord shall endeavor to (i) minimize the disruption to Tenant's use of the Premises and (ii) provide twenty-four (24) hours' prior verbal notice to Tenant (except in cases of emergency).

ARTICLE XIII TENANT'S INDEMNITY AND INSURANCE

13.1 Tenant's Indemnity

(a) <u>Indemnity</u>. To the fullest extent permitted by law, Tenant waives any right to contribution against the Landlord Parties (as hereinafter defined) and agrees to indemnify and save harmless the Landlord Parties from and against all claims of whatever nature arising from or claimed to have arisen from (i) any act, omission or negligence of the Tenant Parties (as hereinafter defined); (ii) any accident, injury or damage whatsoever caused to any person, or to the property of any person, occurring in or about the Premises from the earlier of (A) the date on which any Tenant Party first enters the Premises for any reason or (B) the Lease Commencement Date, and thereafter throughout and until the end of the Lease Term and after the end of the Lease Term for as long as Tenant or anyone acting by, through or under Tenant is in occupancy

of the Premises or any portion thereof; (iii) any accident, injury or damage whatsoever occurring outside the Premises but within the Building or the parking facility, or on common areas or other areas of the Complex, where such accident, injury or damage results, or is claimed to have resulted, from any act, omission or negligence on the part of any of the Tenant Parties; (iv) any breach of this Lease by Tenant; or (v) Tenant's or its employees' or agent's use of the fitness facility in the Building or the Risers (as defined in Section 26.1). Tenant shall pay such indemnified amounts as they are incurred by the Landlord Parties. This indemnification shall not be construed to deny or reduce any other rights or obligations of indemnity that the Landlord Parties may have under this Lease or the common law. Notwithstanding anything contained herein to the contrary, Tenant shall not be obligated to indemnify a Landlord Party for any claims to the extent such Landlord Party's damages in fact result from such Landlord Party's negligence or willful misconduct.

- (b) <u>Breach</u>. In the event that Tenant breaches any of its indemnity obligations hereunder or under any other contractual or common law indemnity: (i) Tenant shall pay to the Landlord Parties all liabilities, loss, cost, or expense (including attorney's fees) incurred as a result of said breach, and the reasonable value of time expended by the Landlord Parties as a result of said breach; and (ii) the Landlord Parties may deduct and offset from any amounts due to Tenant under this Lease any amounts owed by Tenant pursuant to this section.
- (c) No limitation. The indemnification obligations under this Section shall be limited to actual damages, but shall not otherwise be limited in any way by any limitation on the amount or type of damages, compensation or benefits payable by or for Tenant or any subtenant or other occupant of the Premises under workers' compensation acts, disability benefit acts, or other employee benefit acts. Tenant waives any immunity from or limitation on its indemnity or contribution liability to the Landlord Parties based upon such acts.
- (d) <u>Subtenants and other occupants</u>. Tenant shall require its subtenants and other occupants of the Premises to provide similar indemnities to the Landlord Parties in a form acceptable to Landlord.
 - (e) <u>Survival</u>. The terms of this section shall survive any termination or expiration of this Lease.
- (f) Costs. The foregoing indemnity and hold harmless agreement shall include indemnity for all costs, expenses and liabilities (including, without limitation, attorneys' fees and disbursements) incurred by the Landlord Parties in connection with any such claim or any action or proceeding brought thereon, and the defense thereof. In addition, in the event that any action or proceeding shall be brought against one or more Landlord Parties by reason of any such claim, Tenant, upon request from the Landlord Party, shall resist and defend such action or proceeding on behalf of the Landlord Party by counsel appointed by Tenant's insurer (if such claim is covered by insurance without reservation) or otherwise by counsel reasonably satisfactory to the Landlord Party. The Landlord Parties shall not be bound by any compromise or settlement of any such claim, action or proceeding without the prior written consent of such Landlord Parties.

13.2 Tenant's Risk. Tenant agrees to use and occupy the Premises, and to use such other portions of the Building and the Project as Tenant is given the right to use by this Lease at Tenant's own risk. The Landlord Parties shall not be liable to the Tenant Parties for any damage, injury, loss, compensation, or claim (including, but not limited to, claims for the interruption of or loss to a Tenant Party's business) based on, arising out of or resulting from any cause whatsoever, including, but not limited to, repairs to any portion of the Premises or the Building or the Project, any fire, robbery, theft, mysterious disappearance, or any other crime or casualty, the actions of any other tenants of the Building or of any other person or persons, or any leakage in any part or portion of the Premises or the Building or the Project, or from water, rain or snow that may leak into, or flow from any part of the Premises or the Building or the Project, or from drains, pipes or plumbing fixtures in the Building or the Project. Any goods, property or personal effects stored or placed in or about the Premises shall be at the sole risk of the Tenant Party, and neither the Landlord Parties nor their insurers shall in any manner be held responsible therefor. The Landlord Parties shall not be responsible or liable to a Tenant Party, or to those claiming by, through or under a Tenant Party, for any loss or damage that may be occasioned by or through the acts or omissions of persons occupying adjoining premises or any part of the premises adjacent to or connecting with the Premises or any part of the Building or otherwise. The provisions of this section shall be applicable to the fullest extent permitted by law, and until the expiration or earlier termination of the Lease Term, and during such further period as Tenant may use or be in occupancy of any part of the Premises or of the Building. Notwithstanding anything to the contrary contained in this Lease, the Landlord Parties shall not be released from liability for any injury, loss, damages or liability to the extent arising from the gross negligence or willful misconduct of the Landlord Parties on or about the Premises, Building or Project; provided, however, in no event shall the Landlord Parties have any liability to a Tenant Party based on any loss with respect to or interruption in the operation of Tenant's business (except for any rent abatement to which Tenant otherwise may be expressly entitled pursuant to the terms of this Lease).

13.3 Tenant's Commercial General Liability Insurance. Tenant agrees to maintain in full force on or before the earlier of (i) the date on which any Tenant Party first enters the Premises for any reason or (ii) the Lease Commencement Date, and thereafter throughout and until the end of the Lease Term, and after the end of the Lease Term, for so long after the end of the Lease Term as Tenant or anyone acting by, through or under Tenant is in occupancy of the Premises or any portion thereof, a policy of commercial general liability insurance, on an occurrence basis, issued on a form at least as broad as Insurance Services Office ("ISO") Commercial General Liability Coverage "occurrence" form CG 00 01 10 01 or another ISO Commercial General Liability "occurrence" form providing equivalent coverage. Such insurance shall include broad form contractual liability coverage, specifically covering but not limited to the indemnification obligations undertaken by Tenant in this Lease. The minimum limits of liability of such insurance shall be Five Million Dollars (\$5,000,000.00) per occurrence. In addition, in the event Tenant hosts a function in the Premises, Tenant agrees to obtain, and cause any persons or parties providing services for such function to obtain, the appropriate insurance coverages as determined by Landlord (including liquor liability coverage, if applicable) and provide Landlord with evidence of the same.

13.4 Tenant's Property Insurance. Tenant shall maintain at all times during the Term of the Lease, and during such earlier time as Tenant may be performing work in or to the Premises or have property, fixtures, furniture, equipment, machinery, goods, supplies, wares or merchandise on the Premises, and continuing thereafter so long as Tenant is in occupancy of any part of the Premises, business interruption insurance and insurance against loss or damage covered by the so-called "all risk" type insurance coverage with respect to Tenant's property, fixtures, furniture, equipment, machinery, goods, supplies, wares and merchandise, and all alterations, improvements and other modifications made by or on behalf of the Tenant in the Premises, and other property of Tenant located at the Premises, except to the extent paid for by Landlord (collectively "Tenant's Property"). The business interruption insurance required by this section shall be in minimum amounts typically carried by prudent tenants engaged in similar operations, but in no event shall be in an amount less than the Base Rent then in effect during any Lease Year, plus any Additional Rent (as hereinafter defined) due and payable for the immediately preceding Lease Year. The "all risk" insurance required by this section shall be in an amount at least equal to the full replacement cost of Tenant's Property. In addition, during such time as Tenant is performing work in or to the Premises, Tenant, at Tenant's expense, shall also maintain, or shall cause its contractor(s) to maintain, builder's risk insurance for the full insurable value of such work. Landlord and such additional persons or entities as Landlord may reasonably request shall be named as loss payees, as their interests may appear, on the policy or policies required by this section. In the event of loss or damage covered by the "all risk" insurance required by this section, the responsibilities for repairing or restoring the loss or damage shall be determined in accordance with Article XVII below. To the extent that Landlord is obligated to pay for the repair or restoration of the loss or damage covered by the policy, Landlord shall be paid the proceeds of the "all risk" insurance covering the loss or damage. To the extent Tenant is obligated to pay for the repair or restoration of the loss or damage, covered by the policy, Tenant shall be paid the proceeds of the "all risk" insurance covering the loss or damage. If both Landlord and Tenant are obligated to pay for the repair or restoration of the loss or damage covered by the policy, the insurance proceeds shall be paid to each of them in the pro rata proportion of their obligations to repair or restore the loss or damage. If the loss or damage is not repaired or restored (for example, if the Lease is terminated pursuant to Article XVII), the insurance proceeds shall be paid to Landlord and Tenant in the pro rata proportion of their relative contributions to the cost of the leasehold improvements covered by the policy.

13.5 <u>Tenant's Other Insurance</u>. Throughout the Lease Term, Tenant shall obtain and maintain (1) comprehensive automobile liability insurance (covering any automobiles owned or operated by Tenant) issued on a form at least as broad as ISO Business Auto Coverage form CA 00 01 07 97 or other form providing equivalent coverage; (2) worker's compensation insurance or participation in a monopolistic state workers' compensation fund; and (3) employer's liability insurance or (in a monopolistic state) Stop Gap Liability insurance. Such automobile liability insurance shall be in an amount not less than One

Million Dollars (\$1,000,000) for each accident. Such worker's compensation insurance shall carry minimum limits as defined by the law of the jurisdiction in which the Premises are located (as the same may be amended from time to time). Such employer's liability insurance shall be in an amount not less than One Million Dollars (\$1,000,000) for each accident, One Million Dollars (\$1,000,000) disease-policy limit, and One Million Dollars (\$1,000,000) disease-each employee.

13.6 Requirements For Insurance. All insurance required to be maintained by Tenant pursuant to this Lease shall be maintained with responsible companies that are admitted to do business, and are in good standing, in the jurisdiction in which the Premises are located and that have a rating of at least "A" and are within a financial size category of not less than "Class X" in the most current Best's Key Rating Guide or such similar rating as may be reasonably selected by Landlord. All such insurance shall: (1) be acceptable in form and content to Landlord; (2) be primary and noncontributory; and (3) contain an endorsement prohibiting cancellation, failure to renew, reduction of amount of insurance, or change in coverage without the insurer first giving Landlord thirty (30) days' prior written notice (by certified or registered mail, return receipt requested, or by fax or email) of such proposed action. No such liability policy shall contain any deductible or self-insured retention greater than Twenty-Five Thousand Dollars (\$25,000.00) and no such property damage policy shall contain any deductible or self-insured retention greater than One Hundred Thousand Dollars (\$100,000.00). Such deductibles and self-insured retentions shall be deemed to be "insurance" for purposes of the waiver in Section 13.13 below. Landlord reserves the right from time to time to require Tenant to obtain higher minimum amounts of insurance based on such limits as are customarily carried with respect to similar properties in the area in which the Premises are located. The minimum amounts of insurance required by this Lease shall not be reduced by the payment of claims or for any other reason. In the event Tenant shall fail to obtain or maintain any insurance meeting the requirements of this Article, or to deliver such policies or certificates as required by this Article, Landlord may, at its option, on five (5) days notice to Tenant, procure such policies for the account of Tenant, and the cost thereof shall be paid to Landlord within five

13.7 Additional Insureds. To the fullest extent permitted by law, the commercial general liability and auto insurance carried by Tenant pursuant to this Lease, and any additional liability insurance carried by Tenant pursuant to Section 13.3 of this Lease, shall name Landlord and the other persons and entities set forth on Exhibit G attached hereto and made a part hereof, and such other persons and entities as Landlord may reasonably request from time to time as additional insureds with respect to liability arising out of or related to this Lease or the operations of Tenant (collectively "Additional Insureds"). Such insurance shall provide primary coverage without contribution from any other insurance carried by or for the benefit of Landlord, Landlord's managing agent, or other Additional Insureds. Such insurance shall also waive any right of subrogation against each Additional Insured.

13.8 <u>Certificates of Insurance</u>. On or before the earlier of (i) the date on which any Tenant Party first enters the Premises for any reason or (ii) the Lease Commencement

Date, Tenant shall furnish Landlord with certificates evidencing the insurance coverage required by this Lease, and renewal certificates shall be furnished to Landlord at least annually thereafter, and at least thirty (30) days prior to the expiration date of each policy for which a certificate was furnished. (Acceptable forms of such certificates for liability and property insurance, respectively, are attached as Exhibit H.) In jurisdictions requiring mandatory participation in a monopolistic state workers' compensation fund, the insurance certificate requirements for the coverage required for workers' compensation will be satisfied by a letter from the appropriate state agency confirming participation in accordance with statutory requirements. Such current participation letters required by this Section shall be provided every six (6) months for the duration of this Lease. Failure by the Tenant to provide the certificates or letters required by this Section shall not be deemed to be a waiver of the requirements in this Section. Upon request by Landlord, a true and complete copy of any insurance policy required by this Lease shall be delivered to Landlord within ten (10) days following Landlord's request.

13.9 <u>Subtenants and Other Occupants</u>. Tenant shall require its subtenants and other occupants of the Premises to provide written documentation evidencing the obligation of such subtenant or other occupant to indemnify the Landlord Parties to the same extent that Tenant is required to indemnify the Landlord Parties pursuant to section 13.1 above, and to maintain insurance that meets the requirements of this Article, and otherwise to comply with the requirements of this Article. Tenant shall require all such subtenants and occupants to supply certificates of insurance evidencing that the insurance requirements of this Article have been met and shall forward such certificates to Landlord on or before the earlier of (i) the date on which the subtenant or other occupant or any of their respective direct or indirect partners, officers, shareholders, directors, members, trustees, beneficiaries, servants, employees, principals, contractors, licensees, agents, invitees or representatives first enters the Premises or (ii) the commencement of the sublease. Tenant shall be responsible for identifying and remedying any deficiencies in such certificates or policy provisions.

13.10 No Violation of Building Policies. Tenant shall not commit or permit any violation of the policies of fire, boiler, sprinkler, water damage or other insurance covering the Project and/or the fixtures, equipment and property therein carried by Landlord, or do or permit anything to be done, or keep or permit anything to be kept, in the Premises, which in case of any of the foregoing (i) would result in termination of any such policies, (ii) would adversely affect Landlord's right of recovery under any of such policies, or (iii) would result in reputable and independent insurance companies refusing to insure the Project or the property of Landlord in amounts reasonably satisfactory to Landlord.

13.11 Tenant to Pay Premium Increases. If, because of anything done, caused or permitted to be done, or omitted by Tenant (or its subtenant or other occupants of the Premises), the rates for liability, fire, boiler, sprinkler, water damage or other insurance on the Project or on the property and equipment of Landlord or any other tenant or subtenant in the Building shall be higher than they otherwise would be, Tenant shall reimburse Landlord and/or the other tenants and subtenants in the Building for the additional insurance premiums thereafter paid by Landlord or by any of the other tenants and subtenants in the Building which shall have been charged because of the aforesaid reasons, such reimbursement to be made from time to time on Landlord's demand.

13.12 Landlord's Insurance

- (a) <u>Required insurance</u>. Landlord shall maintain insurance against loss or damage with respect to the Building on an "all risk" type insurance form, with customary exceptions, subject to such deductibles as Landlord may determine, in an amount equal to at least the replacement value of the Building. Landlord shall also maintain such insurance with respect to any improvements, alterations, and fixtures of Tenant located at the Premises to the extent paid for by Landlord. The cost of such insurance shall be treated as a part of Operating Expenses. Such insurance shall be maintained with an insurance company selected by Landlord. Payment for losses thereunder shall be made solely to Landlord.
- (b) <u>Optional insurance</u>. Landlord may maintain such additional insurance with respect to the Building and the Project, including, without limitation, earthquake insurance, terrorism insurance, flood insurance, liability insurance and/or rent insurance, as Landlord may in its sole discretion elect. Landlord may also maintain such other insurance as may from time to time be required by the holder of any mortgage encumbering the Building or the Project. The cost of all such additional insurance shall also be part of the Operating Expenses.
- (c) <u>Blanket and self-insurance</u>. Any or all of Landlord's insurance may be provided by blanket coverage maintained by Landlord or any affiliate of Landlord under its insurance program for its portfolio of properties, or by Landlord or any affiliate of Landlord under a program of self-insurance, and in such event Operating Expenses shall include the portion of the reasonable cost of blanket insurance or self-insurance that is allocated to the Building.
- (d) <u>No obligation</u>. Landlord shall not be obligated to insure, and shall not assume any liability of risk of loss for, Tenant's Property, including any such property or work of tenant's subtenants or occupants. Landlord will also have no obligation to carry insurance against, nor be responsible for, any loss suffered by Tenant, subtenants or other occupants due to interruption of Tenant's or any subtenant's or occupant's business.

13.13 Waiver of Subrogation.

(a) To the fullest extent permitted by law, the parties hereto waive and release any and all rights of recovery against the other, and agree not to seek to recover from the other or to make any claim against the other, and in the case of Landlord, against all Tenant Parties, and in the case of Tenant, against all Landlord Parties, for any loss or damage incurred by the waiving/releasing party to the extent such loss or damage is insured under any insurance policy required by this Lease or which would have been so insured had the party carried the insurance it was required to carry hereunder. Tenant shall obtain from its subtenants and other occupants of the Premises a similar waiver and release of claims against any or all of Tenant or Landlord. In addition, the parties hereto (and in the case of Tenant, its subtenants and other occupants of the

Premises) shall procure an appropriate clause in, or endorsement on, any insurance policy required by this Lease pursuant to which the insurance company waives subrogation. The insurance policies required by this Lease shall contain no provision that would invalidate or restrict the parties' waiver and release of the rights of recovery in this section. The parties hereto covenant that no insurer shall hold any right of subrogation against the parties hereto by virtue of such insurance policy.

(b) The term "Landlord Party" or "Landlord Parties" shall mean Landlord, any affiliate of Landlord, Landlord's managing agents for the Building, the holder of any mortgage encumbering the Building, the Land or the Project, the ground lessor under any ground lease encumbering the Land, and each of their respective direct or indirect partners, officers, shareholders, directors, members, trustees, beneficiaries, servants, employees, principals, contractors, licensees, agents or representatives. For the purposes of this Lease, the term "Tenant Party" or "Tenant Parties" shall mean Tenant, any affiliate of Tenant, any permitted subtenant or any other permitted occupant of the Premises, and each of their respective direct or indirect partners, officers, shareholders, directors, members, trustees, beneficiaries, servants, employees, principals, contractors, licensees, agents, invitees or representatives.

13.14 Tenant's Work. During such times as Tenant is performing work or having work or services performed in or to the Premises, Tenant shall require its contractors, and their subcontractors of all tiers, to obtain and maintain commercial general liability, automobile, workers compensation, employer's liability, builder's risk, and equipment/property insurance in accordance with the terms and conditions of Schedule IV to Exhibit B attached hereto and made a part hereof, as the same may be modified by written notice from Landlord from time to time consistent with such terms as are customarily required of such contractors and subcontractors on similar projects. The amounts and terms of all such insurance are subject to Landlord's written approval, which approval shall not be unreasonably withheld. The commercial general liability and auto insurance carried by Tenant's contractors and their subcontractors of all tiers pursuant to this section shall name Landlord, Landlord's managing agent, and the other Additional Insureds provided by Landlord pursuant to Section 13.7 above. Such insurance shall provide primary coverage without contribution from any other insurance carried by or for the benefit of Landlord, Landlord's managing agent, or other Additional Insureds. Such insurance shall also waive any right of subrogation against each Additional Insured. Tenant shall obtain and submit to Landlord, prior to the earlier of (i) the entry onto the Premises by such contractors or subcontractors or (ii) commencement of the work or services, certificates of insurance evidencing compliance with the requirements of this section and Schedule IV to Exhibit B hereto.

ARTICLE XIV SERVICES AND UTILITIES

14.1 Landlord shall furnish to the Premises year-round HVAC during normal hours of operation of the Building, as hereinafter provided, during the seasons when such utilities are required, as determined in Landlord's reasonable judgment. The base Building HVAC system has been designed to provide heating and cooling in accordance with the specifications included in the Base Building Office Shell Definition described on Schedule I to Exhibit B, subject to the terms and conditions of this Article XIV. Landlord shall also provide reasonably adequate electricity, water, exterior windowcleaning service (at least two times per year), and janitorial service (after 6:00 p.m.) on Monday through Friday only, excluding legal holidays, in accordance with the janitorial specifications outlined in Exhibit E attached hereto. Landlord will also provide elevator service; provided, however, that Landlord shall have the right to remove elevators from service as may be required for moving freight, or for servicing or maintaining the elevators or the Building. At least one elevator cab shall be available for use by Tenant at all times (except in the event of an emergency). The normal hours of operation of the Base Building HVAC system will be 7:00 a.m. to 8:00 p.m. on Monday through Friday (except legal holidays) and, upon request by Tenant provided to Landlord prior to 3:00 p.m. on the preceding Friday, in writing, via telephone, or via email, 9:00 a.m. to 4:00 p.m. on Saturday (except legal holidays). There will be no normal hours of operation of the Building on Sundays or legal holidays, and Landlord shall not be obligated to maintain or operate the Building at such times unless special arrangements are made by Tenant. In the event Tenant requires after-hours HVAC service on Saturdays (after 4:00 p.m.), Sundays or legal holidays, Tenant shall request such service prior to 3:00 p.m. on the preceding Friday (or 3:00 p.m. on the preceding business day in the case of desired service on a legal holiday). In the event Tenant requires after-hours HVAC service on regular business days (i.e. service after 8:00 p.m.), Tenant shall give notice to Landlord on or before 3:00 p.m. of such business day. It is understood and agreed that any such advance notice requirement may be satisfied through the use of Tenant's direct, 24/7 telephone or internet access system to after-hours air conditioning and heat, which shall activate such air conditioning or heat on a per zone basis, without the requirement that such notice be provided to Landlord. The services and utilities required to be furnished by Landlord, other than electricity and water, will be provided only during the normal hours of operation of the Building, except as otherwise specified herein. It is agreed that if Tenant requires HVAC beyond the normal hours of operation set forth herein, Landlord will furnish such HVAC, provided Tenant gives Landlord's agent sufficient advance notice of such requirement (as set forth above) and Tenant agrees to pay for the cost of such overtime HVAC in accordance with Landlord's then current schedule of costs and assessments for such HVAC as charged to other tenants in the Building (which overtime HVAC rate, as of the Effective Date, is \$38.25 per hour per zone, and is subject to change from time to time). Landlord agrees to provide an access-control system in the Building comparable to the system

in Class A office and retail buildings in the Market Area, and Tenant shall have access to the Premises on a 24-hour, seven-days-a-week basis (except in the event of emergency). Landlord shall, at its cost, provide an initial set of access cards to the Building and Garage in an amount equal to the number of initial employees of Tenant who work on a full-time basis at the Premises as of the Lease Commencement Date in an aggregate amount not to exceed one (1) access card for each five hundred (500) square feet of above grade rentable area in the Premises (excluding any storage space leased by Tenant); provided, however, that any replacement or additional cards requested by Tenant after the Lease Commencement Date shall be provided by Landlord and Tenant shall reimburse Landlord for Landlord's cost therefor. Any access control system installed by Tenant for the Premises, whether as part of the initial Leasehold Work or as a subsequent Alteration, shall be subject to Landlord's prior written approval and the other terms and conditions of this Lease, and shall be compatible with the base Building access control system. Upon Tenant's written request, Landlord shall program the Building elevators to require access cards for access to the floor on which the Premises are located, provided Tenant is leasing the entire rentable area on such floor. Landlord shall provide an attendant for the Building on a twenty-four (24)-hour basis, which attendant will be stationed in the main lobby of the Building during the normal hours of operation of the Building, the costs of which will be included in Operating Expenses to the extent permitted pursuant to Article IV above. Landlord also agrees to provide cameras directed at certain means of ingress/egress to the Building for recording purposes only and for no other use or purpose, the costs of which will be included in Operating Expenses to the extent permitted pursuant to Article IV above. Notwithstanding anything contained in this Lease to the contrary, Tenant expressly acknowledges and agrees that (a) in no event shall Landlord or any employee, agent or contractor of Landlord be required to man, monitor or respond to any output of any cameras or other devices recording access to or from the Building, and (b) no liability or obligation whatsoever on the part of Landlord or any employee, agent or contractor of Landlord is imposed or implied by any such cameras or other recording devices that may be installed in the Building.

14.2 It is understood and agreed that Landlord shall not have any liability to Tenant whatsoever as a result of Landlord's failure or inability to furnish any of the utilities or services required to be furnished by Landlord hereunder, whether resulting from breakdown, removal from service for maintenance or repairs, strikes, scarcity of labor or materials, acts of God, governmental requirements, or any other cause whatsoever. It is further agreed that any such failure or inability to furnish the utilities or services required hereunder shall not be considered an eviction, actual or constructive, of Tenant from the Premises, and shall not entitle Tenant to terminate this Lease or to an abatement of any rent payable hereunder.

14.3 Landlord will use its commercially reasonable efforts to cause the restoration of any interrupted utility services; further, should any equipment or machinery in the Building break down so as to render the Premises unusable by Tenant, Landlord shall promptly repair or replace it (subject to delays which result from strikes, unavailability of parts or other materials, or other matters beyond Landlord's reasonable control). Notwithstanding the provisions of Section 14.2 to the contrary, if (a) the services described in Section 14.1 hereof are interrupted for (i) a period of more than five (5) consecutive business days as a result of Landlord's (or its agents' or employees') negligence or willful misconduct, or (ii) a period of more than eight (8) consecutive business days for any other reason (other than Tenant's or its agents' or employees' negligence or willful misconduct), or (b) Tenant's access to the Premises is prevented for a period of more than five (5) consecutive business days as a result of Landlord's (or its agents' or employees') negligence or willful misconduct, (c) such interruption of services or access is not the result of any act of Tenant or its Invitees, and (d) such interruption of services or access renders all or a substantial portion of the Premises untenantable by Tenant and the Premises or such portion thereof are not used or occupied by Tenant, then, as Tenant's sole and exclusive remedy therefor, Tenant shall be entitled to a pro rata abatement of Base Rent and Operating Expenses beginning on the sixth (6th) consecutive business day or the ninth (9th) consecutive business day, whichever is applicable, that the Premises are untenantable (and not used or occupied) and continuing until the Premises or such portion thereof is rendered tenantable, it being agreed that such time periods shall not be extended as a result of a Force Majeure Event (as defined below).

14.4 The parties hereto agree to comply with all energy conservation controls and requirements applicable to office and retail buildings that are imposed or instituted by the Federal, state or local governments, including without limitation, controls on the permitted range of temperature settings in office and retail buildings, and requirements necessitating curtailment of the volume of energy consumption or the hours of operation of the Building. Any terms or conditions of this Lease that conflict or interfere with compliance with such controls or requirements shall be suspended for the duration of such controls or requirements. It is further agreed that compliance with such controls or requirements shall not be considered an eviction, actual or constructive, of Tenant from the Premises and shall not entitle Tenant to terminate this Lease or to an abatement of any rent payable hereunder.

14.5 Tenant shall reimburse Landlord for any excess water usage in the Premises. "Excess water usage" shall mean the excess of Tenant's water usage during any billing period for water services over the estimated average water usage during the same period for all office tenants of the Building (excluding Tenant), as computed by Landlord. If Tenant connects into Landlord's supplemental cooling system currently located (or to be located) on the roof of the Building, then Tenant shall reimburse Landlord for all costs incurred by Landlord therefor, as reasonably determined by Landlord. Landlord may install checkmeters to electrical circuits serving Tenant's equipment to verify that Tenant is not consuming excessive electricity. If such checkmeters indicate that Tenant's electricity consumption is excessive, then Landlord may install at Tenant's expense submeters to ascertain Tenant's actual electricity consumption, and Tenant shall thereafter

pay for such consumption at the then-current price per kilowatt hour charged Landlord by the utility. Tenant's electricity consumption shall be deemed excessive if the electricity consumption in the Premises per square foot of rentable area (including, without limitation, electricity consumed in connection with outlets and lighting use) during any billing period exceeds the average electricity consumption per square foot of rentable area during the same period for typical, similarly situated tenants in the Building, as reasonably calculated by Landlord. Similarly, Tenant shall reimburse Landlord for any excess usage of supplemental condenser water, which excess usage shall be paid for by Tenant in the same manner and subject to similar conditions as apply to excess water usage and electricity usage pursuant to this Section 14.5.

14.6 The Building (including the roof deck and the Garage) is a non-smoking facility. Tenant agrees to adhere to Landlord's rules and regulations pertaining to such policy (as the same may be amended from time to time), as set forth in the Building's Rules and Regulations, a current copy of which are attached hereto as Exhibit C.

14.7 Subject to applicable Legal Requirements and governmental and quasi-governmental prohibitions and/or restrictions, for so long as Tenant is a tenant in the Building, Tenant's employees who work in the Building shall have the nonexclusive right (subject to reasonable rules and regulations and reasonable fees for elective services, if any, but excluding membership fees) to use the fitness facility. The fitness facility shall be available to Tenant's employees who work in the Building on a regular basis on a non-exclusive first-come, first-served basis. Landlord may specifically condition the use of the fitness facility by any person upon such person's execution of a written waiver and release holding Landlord harmless from any and all liability, damage, expense, cause of action, suit, claim, judgment and cost of defense arising from injury to such employee or guest occurring in the fitness facility or resulting from the use thereof. Neither Landlord nor Landlord's agents or partners, shall have any liability to Tenant or its Invitees for any damage, injury, loss, expense, compensation or claim whatsoever arising out of the use of the fitness facility.

14.8 Subject to applicable Legal Requirements, governmental or quasi-governmental prohibitions and/or restrictions (including without limitation any temporary or permanent closure(s) of the roof deck and/or restriction of tenants' use thereof due to concerns about terror or terrorism), the availability of insurance at commercially reasonable rates, and Landlord's reasonable rules with respect thereto that may be established from time to time, the Building shall contain a roof deck located on the top floor of the Building that will be available for Tenant's use on a non-exclusive, first-come, first-served basis. Except in the event of an emergency, Tenant's employees shall have access to the roof deck during the normal hours of operation of the Building, excluding Inauguration Day, Independence Day and other legal holidays, as well as such other days and times as Landlord shall reasonably determine, including such dates and times as Landlord may have granted another Tenant exclusive use of the roof deck. With prior authorization from Landlord and subject to availability and to compliance with rules and regulations established from time to time by Landlord (including Tenant's reimbursing Landlord as additional rent for all costs associated therewith including, but not limited to,

cleaning, landscaping and access control), Tenant shall have the right to hold "events" on the roof deck. Landlord shall not be liable to Tenant or any of its Invitees for injuries received by Tenant or its Invitees while using the roof deck, and Tenant agrees to indemnify and save harmless Landlord from any such liability. Landlord reserves the right to institute a card-key (or similar type of) access system to permit access to the roof deck (the cost of which shall be included as an Operating Expense), and Tenant agrees to reimburse Landlord for the costs of any such cards distributed to Tenant.

14.9 Unless otherwise expressly provided in this Lease, costs for the services and utilities required to be furnished and/or performed by Landlord that are described in this Article XIV shall be passed through to Tenant as an Operating Expense to the extent permitted pursuant to Article IV hereof.

ARTICLE XV LIABILITY OF LANDLORD

15.1 Landlord shall not be liable to Tenant or to its Invitees for any damage, injury, loss, compensation or claim, including but not limited to claims for the interruption of or loss to Tenant's business, based on, arising out of, or resulting from any cause whatsoever, including but not limited to the following: repairs to any portion of the Premises or the Building; interruption in the use of the Premises; any accident or damage resulting from the use or operation (by Landlord, Tenant or any other person or persons) of elevators, or of the heating, cooling, electrical or plumbing equipment or apparatus; the termination of this Lease by reason of the destruction of the Premises or the Building; any fire, robbery, theft, mysterious disappearance or any other casualty; the actions of any other tenants of the Building or of any other person or persons; and any leakage in any part or portion of the Premises or the Building, or from water, rain or snow that may leak into, or flow from, any part of the Premises or the Building, or from drains, pipes or plumbing fixtures in the Building. Any goods, property or personal effects stored or placed by Tenant or its employees in or about the Premises or the Building shall be at the sole risk of Tenant, and Landlord shall not in any manner be held responsible therefor. It is understood that the employees of Landlord are prohibited from receiving any packages or other articles delivered to the Building for Tenant, and if any such employee receives any such package or articles, such employee shall be acting as the agent of Tenant for such purposes and not as the agent of Landlord. Notwithstanding the foregoing provisions of this Section 15.1 to the contrary, Landlord shall not be released from liability to Tenant for damage or injury caused by the negligence or willful misconduct of Landlord or its employees; provided, however, in no event shall Landlord have any liability to Tenant for any claims based on the interruption of or loss to Tenant's business (except for any rent

15.2 In the event that at any time Landlord shall sell or transfer title to the Building, provided the purchaser or transferee assumes the obligations of Landlord

hereunder arising from and after the date of the transfer, Landlord named herein shall not be liable to Tenant for any obligations or liabilities based on or arising out of events or conditions occurring on or after the date of such sale or transfer. Furthermore, Tenant agrees to attorn to any such purchaser or transferee upon all the terms and conditions of this Lease.

- 15.3 In the event that Tenant shall have a claim against Landlord, Tenant shall not have the right to deduct the amount allegedly owed to Tenant from any rent or other sums payable to Landlord hereunder, it being understood that Tenant's sole remedy for recovering upon such claim shall be to institute an independent action against Landlord.
- 15.4 Tenant agrees that in the event Tenant is awarded a money judgment against Landlord, Tenant's sole recourse for satisfaction of such judgment shall be limited to execution against the estate and interest of Landlord in the Building. In no event shall any other assets of Landlord, any partner of Landlord, the holder of any mortgage (or anyone claiming by through or under such holder) or any other person or entity be available to satisfy, or be subject to, such judgment, nor shall any partner of Landlord or any such other person or entity be held to have any personal liability for satisfaction of any claims or judgments that Tenant may have against Landlord or any partner of Landlord in such partner's capacity as a partner of Landlord.

ARTICLE XVI RULES AND REGULATIONS

16.1 Tenant and its Invitees shall at all times abide by and observe the Rules and Regulations attached hereto as Exhibit C. In addition, Tenant and its Invitees shall abide by and observe all other rules or regulations that Landlord may promulgate from time to time for the operation and maintenance of the Building, provided that notice thereof is given to Tenant and such rules and regulations are commercially reasonable and are not inconsistent with the provisions of this Lease. Nothing contained in this Lease shall be construed as imposing upon Landlord any duty or obligation to enforce such rules and regulations, or the terms, conditions or covenants contained in any other lease, as against any other tenant, and Landlord shall not be liable to Tenant or its Invitees for the violation of such rules or regulations by any other tenant or such other tenant's employees, agents, invitees, licensees, customers, subtenants, contractors, clients, family members or guests. Landlord shall use reasonable efforts to enforce all such rules and regulations, although it is understood that Landlord may grant exceptions to such rules and regulations in circumstances in which it reasonably determines such exceptions are warranted. If there is any inconsistency between this Lease and the Rules and Regulations set forth in Exhibit C, this Lease shall govern.

ARTICLE XVII DAMAGE OR DESTRUCTION

- 17.1 (a) Subject to subparagraphs (c) and (d) below, if the Premises or the Building are totally or partially damaged or destroyed from any cause, thereby rendering the Premises totally or partially inaccessible or untenantable, Landlord shall diligently (taking into account the time necessary to effectuate a satisfactory settlement with any insurance company involved) restore and repair the Premises and/or the Building to substantially the same condition they were in prior to such damage or destruction.
- (b) Within forty-five (45) days after the occurrence of such damage or destruction (the "**Determination Period**"), Landlord will provide Tenant, in writing (the "**Restoration Notice**"), with a good faith estimate of the date by which the repairs and restoration will be completed, including the time needed for removal of debris, preparation of plans, bidding of contracts, and issuance of all required governmental permits.
- (c) If, in the sole judgment of Landlord, the repairs and restoration cannot be completed within one hundred eighty (180) days after the occurrence of such damage or destruction, including the time needed for removal of debris, preparation of plans, bidding of contracts, and issuance of all required governmental permits, Landlord shall have the right, at its sole option, to terminate this Lease by giving written notice to Tenant at any time prior to the expiration of the Determination Period.
- (d) Additionally, if, in the sole judgment of Landlord, exercised in good faith, the Building is damaged or destroyed from any cause to such an extent that the costs of repairing and restoring the Building would exceed fifty percent (50%) of the replacement value of the Building at the time of such damage or destruction, whether or not the Premises are damaged or destroyed, then Landlord shall have the right, at its sole option, to terminate this Lease by giving written notice of termination to Tenant at any time prior to the expiration of the Determination Period.
- 17.2 If the Restoration Notice provides that the repairs and restoration cannot be substantially completed within one hundred and eighty (180) days after the occurrence of such damage or destruction, then Tenant shall have the right to terminate this Lease by providing written notice to Landlord within thirty (30) days after the date of the Restoration Notice. Notwithstanding the foregoing, Tenant shall not have the right to terminate this Lease if the act or omission of Tenant, or any Tenant Party, shall have caused the damage or destruction.
- 17.3 If this Lease is terminated pursuant to this Article XVII, all rent payable hereunder shall be apportioned and paid to the date of the occurrence of such damage or destruction and Tenant shall have no further rights or remedies against Landlord pursuant to this Lease, or otherwise. If this Lease is not terminated pursuant to the terms of this Article XVII, and provided that such damage or destruction was not caused by the act or omission to act of Tenant, or any Tenant Party, until the repair and restoration of the Premises is completed, Tenant shall be required to pay annual base rent and additional rent only for that portion of the Premises that Tenant is able to use while repairs are being made, based on the ratio that the amount of rentable area in the usable portion of the Premises bears to the total rentable area of the Premises.

17.4 If this Lease is not terminated as provided in this Article XVII, Landlord shall proceed to diligently repair and restore the Premises (including the Leasehold Work) and/or the Building using (i) the proceeds of Landlord's insurance (covering damage to the Building and to the Leasehold Work up to the amount of the Improvement Allowance, as defined in Exhibit B attached hereto), calculated on a per rentable square foot basis and adjusted to reflect a commercially reasonable rate of inflation, and (ii) the proceeds of Tenant's insurance (covering the Leasehold Work in excess of the amount of the Improvement Allowance, calculated on a per rentable square foot basis and adjusted in accordance with the foregoing). Tenant shall be required to repair and restore at its sole expense all decorations, trade fixtures, furnishings, equipment and personal property installed by or belonging to Tenant. In connection with any restoration of the Leasehold Work, Landlord shall perform the Leasehold Work, and Landlord shall be obligated to pay for the cost of the Leasehold Work only up to the amount of the Improvement Allowance. Tenant shall reimburse Landlord (within thirty (30) days of demand therefor) for the cost of any Leasehold Work above the amount of the Improvement Allowance.

17.5 Notwithstanding anything provided herein to the contrary, Landlord shall not be obligated to restore the Premises and/or the Building if (i) the damage or destruction was not caused by an insurable event, or (ii) the estimated cost of such repair or restoration, as determined by Landlord in its sole judgment, exceeds the amount of insurance proceeds available to Landlord for such repair or restoration. This right of termination shall be in addition to any other right of termination provided in this Lease.

ARTICLE XVIII CONDEMNATION

18.1 (a) If the whole or a substantial part (as hereinafter defined) of the Building or the Premises, or the use or occupancy of a substantial part of the Premises, shall be taken or condemned by any governmental or quasi-governmental authority for any public or quasi-public use or purpose (including a sale thereof under threat of such a taking), then this Lease shall terminate on the date title thereto vests in such governmental or quasi-governmental authority, and all rent payable hereunder shall be apportioned as of such date. If less than a substantial part of the Premises, or the use or occupancy thereof, is taken or condemned by any governmental or quasi-governmental authority for any public or quasi-public use or purpose (including a sale thereof under threat of such a taking), this Lease shall continue in full force and effect, but the annual base rent and additional rent thereafter payable hereunder shall be equitably adjusted (on the basis of the ratio of the number of square feet of rentable area taken to the total rentable area of the Premises prior to such taking) as of the date title vests in the governmental or quasi-governmental authority. For purposes of this Section 18.1(a), a "substantial part" of the Building or the Premises shall be considered to have been taken if, as a result of any permanent taking, the remainder of the Premises is not reasonably satisfactory for the conduct of Tenant's business operations in the ordinary course therein, including without limitation, as a result of Tenant's inability to access such remainder of the Premises.

(b) In addition to the foregoing, if (i) there occurs a taking of such portion of the Garage or Tenant's parking rights therein so as to deprive Tenant of the use of more than fifty percent (50%) of the number of parking permits to which Tenant is entitled pursuant to Article XXIV below for a period of more than one hundred eighty (180) consecutive days, and (ii) Landlord is unable to provide Tenant with (A) an equal number of substitute parking permits within a six (6)-block radius of the Building (in which event, in lieu of paying Landlord the charge for the parking permits for which Tenant is deprived of the use as a result of such taking, Tenant will contract directly for such substitute permits and Landlord will reimburse Tenant therefor in an amount not to exceed the product of the then current rate for unreserved parking permits at the Building multiplied by the number of parking permits for which Tenant is deprived of the use under this Lease) or (B) other reasonable parking accommodations, then Tenant shall have the right to terminate this Lease by providing Landlord, within thirty (30) days after the end of such one hundred eighty (180) day period, with sixty (60) days prior written notice thereof. If Tenant fails to provide such termination notice within such thirty (30)-day period, Deannt shall have no right to terminate this Lease pursuant to this Section 18.1(b). If Tenant provides such termination notice within such thirty (30)-day period, but Landlord, within sixty (60) days after the date of Tenant's termination notice, provides Tenant with alternative parking arrangements permitted hereunder or otherwise makes available in the Garage the permits to which Tenant is entitled hereunder, Tenant's termination notice automatically shall be void and without force or effect; otherwise, Tenant's termination notice shall be effective as of the date that is sixty (60) days from the date of such notice.

18.2 All awards, damages and other compensation paid by the condemning authority on account of such taking or condemnation (or sale under threat of such taking) shall belong to Landlord, and Tenant hereby assigns to Landlord all rights to such awards, damages and compensation. Tenant agrees not to make any claim against Landlord or the condemning authority for any portion of such award or compensation attributable to damage to the Premises, the value of the unexpired term of this Lease, the loss of profits or goodwill, leasehold improvements or severance damages. Nothing contained herein, however, shall prevent Tenant from pursuing a separate claim against the condemning authority for the value of furnishings, equipment and trade fixtures installed in the Premises at Tenant's expense and for relocation expenses, provided that such claim does not in any way diminish the award or compensation payable to or recoverable by Landlord in connection with such taking or condemnation.

ARTICLE XIX DEFAULT BY TENANT

- 19.1 In addition to those events or occurrences described in this Lease as an Event of Default, the occurrence of any of the following shall constitute an "Event of Default" by Tenant under this Lease:
- (a) If Tenant shall fail to pay any installment of base rent, additional rent or any other sums required by this Lease when due and such failure shall remain uncured for a period of five (5) days after Landlord notifies Tenant in writing of such failure; provided however, that Landlord shall not be required to give Tenant more than one (1) such written notice in any twelve (12) month period, nor more than five (5) such written notices over the Lease Term (after which time nonpayment on the date due shall constitute an Event of Default).
- (b) If Tenant shall violate or fail to perform any other term, condition, covenant or agreement to be performed or observed by Tenant under this Lease and such violation or failure shall continue uncured for a period of thirty (30) days after Landlord notifies Tenant of such violation or failure. If such violation or failure is not capable of being cured within such thirty (30) day period, then provided Tenant commences curative action within such thirty (30) day period and proceeds diligently and in good faith thereafter to cure such violation or failure, such cure period shall be extended for a reasonable time not to exceed ninety (90) days.
- (c) If Tenant shall assign its interest in this Lease or sublet any portion of the Premises in violation of the requirements of Article VII of this Lease.
 - (d) [Intentionally Omitted].
 - (e) If Tenant permits any liens to continue on the Premises, or any part thereof, beyond the periods set forth herein.
 - (f) If an Event of Bankruptcy, as defined in Section 20.1 of this Lease, shall occur.
 - (g) If an Environmental Default, as defined in Section 6.4 of this Lease, shall occur.
- 19.2 (a) If there shall be an Event of Default, then the provisions of this Section 19.2 shall apply. Landlord shall have the right, at its sole option, to terminate this Lease. In addition, with or without terminating this Lease, Landlord may reenter, terminate Tenant's right of possession and take possession of the Premises. The provisions of this Article shall operate as a notice to quit, and Tenant hereby waives any other notice to quit or notice of Landlord's intention to reenter the Premises or terminate this Lease. If necessary, Landlord may proceed to recover possession of the Premises under the applicable Legal Requirements, or by such other proceedings, including reentry and possession, as may be applicable. If Landlord elects to terminate this Lease and/or elects to terminate Tenant's right of possession, everything contained in this Lease required to be done or performed by Landlord shall cease, without prejudice, however, with regard to Tenant's liability for all base rent, additional rent and other sums due under this Lease. Whether or not this Lease and/or Tenant's right of possession is terminated, Landlord shall have the right, at its sole option, to terminate any renewal or expansion right

contained in this Lease and to grant or withhold any consent or approval pursuant to this Lease in its sole and absolute discretion. Landlord may relet the Premises or any part thereof, alone or together with other premises, for such term(s) (which may extend beyond the date on which the Lease Term would have expired but for Tenant's default) and on such terms and conditions (which may include any concessions or allowances granted by Landlord) as Landlord, in its sole and absolute discretion, may determine, but Landlord shall not be liable for, nor shall Tenant's obligations hereunder be diminished by reason of, any failure by Landlord to relet all or any portion of the Premises or to collect any rent due upon such reletting. If there has occurred an Event of Default under this Lease, and Landlord has either terminated this Lease or Tenant's right of possession hereunder, and Tenant has vacated the Premises, then Landlord shall thereafter use reasonable efforts to remarket the Premises and consummate market leasing transactions. Notwithstanding anything to the contrary in this Section 19.2, Tenant expressly acknowledges that Landlord's agreement to use reasonable efforts to relet the Premises in accordance with the terms and conditions herein specified shall in no event limit, restrict or prejudice Landlord's right to lease all other vacant or to be vacated space in the Building prior to reletting the Premises.

(b) Whether or not this Lease and/or Tenant's right of possession is terminated or any suit is instituted, Tenant shall be liable for any base rent, additional rent, damages or other sum which may be due or sustained prior to such Event of Default, and for all costs, fees and expenses (including, but not limited to, reasonable attorneys' fees and costs, brokerage fees, expenses incurred in enforcing any of Tenant's obligations under this Lease or in placing the Premises in Class A rentable condition, advertising expenses, and any concessions or allowances granted by Landlord) incurred by Landlord in pursuit of its remedies hereunder and/or in recovering possession of the Premises and renting the Premises to others from time to time plus other damages suffered or incurred by Landlord on account of such Event of Default (including, but not limited to late fees or other charges incurred by Landlord under any mortgage). Tenant also shall be liable for additional damages which at Landlord's election shall be either one or any combination of the following:

(i) an amount equal to the Base Rent and additional rent due or which would have become due from the date of such Event of Default through the remainder of the Lease Term, plus all expenses (including broker and attorneys' fees) incurred in connection with the reletting of the Premises, less the amount of rental income, if any, which Landlord actually receives during such period from others to whom Landlord may (but is not required to) relet the Premises, other than any additional rent received by Landlord as a result of any failure of such other person to perform any of its obligations to Landlord, which amount shall be computed and payable in monthly installments, in advance, on the first day of each calendar month following such Event of Default and continuing until the date on which the Lease Term would have expired but for such Event of Default, it being understood that separate suits may be brought, at Landlord's discretion, from time to time to collect any such damages for any month(s) (and any such separate suit shall not in any manner prejudice the right of Landlord to collect any damages for any subsequent month(s)), or Landlord may defer initiating any such suit

until after the expiration of the Lease Term (in which event such deferral shall not be construed as a waiver of Landlord's rights as set forth herein and Landlord's cause of action shall be deemed for limitations purposes not to have accrued until the expiration of the Lease Term), and it being further understood that if Landlord elects to bring suits from time to time prior to reletting the Premises, Landlord shall be entitled to its full damages through the date of the award of damages without regard to any Base Rent, additional rent or other sums that are or may be projected to be received by Landlord upon reletting of the Premises; or

(ii) an amount equal to the sum of (a) all base rent, additional rent and other sums due or which would be due and payable under this Lease as of the date of such Event of Default through the end of the scheduled Lease Term, plus (b) all expenses (including broker and attorneys' fees) incurred in connection with the reletting of the Premises, minus (c) any base rent, additional rent and other sums which Tenant proves by a preponderance of the evidence would be received by Landlord upon reletting of the Premises from the end of the vacancy period projected by Landlord through the expiration of the scheduled Lease Term.

The damage amounts calculated under option (ii) shall be accelerated and discounted using a discount factor equal to the yield of the Treasury Note or Bill, as appropriate, having a maturity period approximately commensurate to the remainder of the Term, and such resulting amount shall be payable to Landlord in a lump sum on demand, and Landlord may bring suit to collect any such damages at any time after an Event of Default if Tenant does not make such payment on demand, it being understood that upon payment of such liquidated and agreed final damages, Tenant shall be released from further liability under this Lease with respect to the period after the date of such payment.

(c) In the event Landlord relets the Premises together with other premises or for a term extending beyond the scheduled expiration of the Lease Term, it is understood that Tenant will not be entitled to apply any base rent, additional rent or other sums generated or projected to be generated by either such other premises or in the period extending beyond the scheduled expiration of the Lease Term (collectively, "Extra Rent") against Landlord's damages. Similarly in proving the amount that would be received by Landlord upon a reletting of the Premises as set forth in subparagraph (b) above, Tenant shall not take into account the Extra Rent. The provisions contained in this Section shall be in addition to, and shall not prevent the enforcement of, any claim Landlord may have against Tenant for anticipatory breach of this Lease. Nothing herein shall be construed to affect or prejudice Landlord's right to prove, and claim in full, unpaid rent accrued prior to termination of this Lease. If Landlord is entitled, or Tenant is required, pursuant to any provision hereof to take any action upon the termination of the Lease Term, then Landlord shall be entitled, and Tenant shall be required, to take such action also upon the termination of Tenant's right of possession.

19.3 (a) Tenant hereby expressly waives, for itself and all persons claiming by, through or under it, any right of redemption, reentry or restoration of the operation of this Lease under any present or future Legal Requirements, including without limitation any such right which Tenant would otherwise have in case Tenant shall be dispossessed for any cause, or in case Landlord shall obtain possession of the Premises as herein provided.

(b) All rights and remedies of Landlord set forth herein are in addition to all other rights and remedies available to Landlord hereunder or at law or in equity. All rights and remedies available to Landlord hereunder or at law or in equity are expressly declared to be cumulative. The exercise by Landlord of any such right or remedy shall not prevent the concurrent or subsequent exercise of any other right or remedy. No delay in the enforcement or exercise of any such right or remedy shall constitute a waiver of any default by Tenant hereunder or of any of Landlord's rights or remedies in connection therewith. Landlord shall not be deemed to have waived any default by Tenant hereunder unless such waiver is set forth in a written instrument signed by Landlord. If Landlord waives in writing any default by Tenant, such waiver shall not be construed as a waiver of any covenant, condition or agreement set forth in this Lease except as to the specific circumstances described in such written waiver.

19.4 If Landlord shall institute proceedings against Tenant and a compromise or settlement thereof shall be made, the same shall not constitute a waiver of default or of any other covenant, condition or agreement set forth herein, nor of any of Landlord's rights hereunder. Neither the payment by Tenant of a lesser amount than the installments of annual base rent, additional rent or of any sums due hereunder nor any endorsement or statement on a check or letter accompanying a check for payment of rent or other sums payable hereunder shall be deemed an accord and satisfaction, and Landlord may accept such check or payment without prejudice to Landlord's right to recover the balance of such rent or other sums or to pursue any other remedy available to Landlord. No re-entry by Landlord, and no acceptance by Landlord of keys from Tenant, shall be considered an acceptance of a surrender of this Lease.

19.5 If Tenant defaults in the making of any payment or in the doing of any act herein required to be made or done by Tenant, then Landlord may, but shall not be required to, make such payment or do such act. If Landlord elects to make such payment or do such act, all costs and expenses incurred by Landlord, plus interest thereon at the rate per annum ("Default Rate") which is two percent (2%) higher than the publicly announced "prime rate" then being reported by the Bank of America, from the date paid by Landlord to the date of payment thereof by Tenant, shall be immediately paid by Tenant to Landlord; provided, however, that nothing contained herein shall be construed as permitting Landlord to charge or receive interest in excess of the maximum legal rate then allowed by law. The taking of such action by Landlord shall not be considered as a cure of such default by Tenant or prevent Landlord from pursuing any remedy it is otherwise entitled to in connection with such default.

19.6 If Tenant fails to make any payment of base rent or of additional rent on or before the date such payment is due and payable, Tenant shall pay to Landlord a late

charge of five percent (5%) of the amount of such payment. In addition, such payment shall bear interest at the Default Rate from the date such payment became due to the date of payment thereof by Tenant; provided, however, that nothing contained herein shall be construed as permitting Landlord to charge or receive interest in excess of the maximum legal rate then allowed by law. Such late charge and interest shall constitute additional rent due and payable hereunder with the next installment of annual base rent due hereunder.

19.7 [Intentionally Omitted].

19.8 In the event either Landlord or Tenant shall employ an attorney to enforce the other party's covenants and obligations under this Lease, whether or not Landlord proceeds to recover possession or Landlord or Tenant commence any other proceeding against the other party, the non-prevailing party shall be liable for all costs and expenses sustained by the prevailing party in the enforcement of such covenants and obligations, including but not limited to attorneys' fees and expenses, costs of collection and court costs.

ARTICLE XX BANKRUPTCY

- 20.1 Each of the following shall be an "Event of Bankruptcy" under this Lease:
- (a) Tenant's or any guarantor's (i) becoming insolvent, as that term is defined in Title 11 of the United States Code ("Bankruptcy Code") or under the insolvency laws of any state, district, commonwealth or territory of the United States ("Insolvency Laws"); (ii) generally not paying its debts as they become due unless such debts are the subject of a bona fide dispute, or (iii) inability to pay its debts as they become due;
- (b) The appointment of a receiver, trustee, custodian or any similar responsible party or representative for any or all of Tenant's or any guarantor's property or assets, or the institution of a foreclosure, replevin, forfeiture, seizure, attachment, garnishment, or any similar action, proceeding or process upon or against any of Tenant's or any guarantor's real, personal or other property;
 - (c) The filing by Tenant or any guarantor of a voluntary petition under the provisions of the Bankruptcy Code or Insolvency Laws;
- (d) The filing of an involuntary petition against Tenant or any guarantor as the subject debtor under the Bankruptcy Code or Insolvency Laws, which either (i) is not dismissed within sixty (60) days of filing, or (ii) results in the issuance of an order for relief against the debtor; or
- (e) Tenant's or any guarantor's making or consenting to an assignment for the benefit of creditors or a common law composition of creditors, or otherwise consenting to the default rights or remedies of Tenant's or any guarantor's other creditors.

ARTICLE XXI SUBORDINATION

- 21.1 (a) This Lease and Tenant's rights under this Lease are subject to and subordinate to the Office Ground Lease, and any other ground lease or underlying lease, first mortgage, first deed of trust, or other first lien encumbrance or indenture, together with any renewals, extensions, modifications, consolidations, and replacements of them (each a "Superior Lien") that now or hereafter affects the Premises or any interest of Landlord in the Premises or Landlord's interest in this Lease and the estate created by this Lease (except to the extent that the recorded instrument evidencing the Superior Lien expressly provides that this Lease is superior to the Superior Lien). The holder of any Superior Lien shall be referred to herein as a "Superior Lien Holder." This Lease shall also be subject and subordinate to the lien of any second or junior mortgages that may hereafter encumber the Building, provided the holder of the first mortgage consents to such subordination. At any time after the execution of this Lease, a Superior Lien Holder shall have the right to declare this Lease to be superior to the lien of its Superior Lien, and Tenant agrees to execute all documents required by such holder in confirmation thereof.
- (b) There are no mortgages or ground leases (other than the Office Ground Lease) encumbering the Building as of the Effective Date. Landlord shall obtain from GWU, concurrently with the execution and delivery of this Lease, a nondisturbance agreement on the form attached hereto as **Exhibit J** for the benefit of Tenant ("**GW NDA**"). In addition, Landlord shall, at no cost to Landlord, obtain from any future holder of any mortgage or deed of trust on the Building a subordination, non-disturbance and attornment agreement ("**SNDA**") on such holder's standard form. Notwithstanding anything to the contrary contained herein, subordination of this Lease to any such mortgage or deed of trust hereafter placed on the Building is conditioned upon receipt of an SNDA as described herein. In connection with each SNDA obtained in favor of Tenant, Tenant shall reimburse Landlord, as additional rent, for the out-of-pocket costs and expenses incurred by Landlord in connection therewith, up to an amount equal to One Thousand Dollars (\$1,000.00) per SNDA.
- 21.2 In confirmation of the foregoing subordination, Tenant shall, at Landlord's request, promptly execute, acknowledge and deliver any requisite or appropriate certificate or other document. If Tenant fails to execute, acknowledge and deliver such certificate or other document within ten (10) days after Landlord's written request, and such failure continues for two (2) business days after a second (2nd) written notice from Landlord, Landlord and its successors and assigns will be entitled to execute, acknowledge and deliver any such certificate or other document on behalf of Tenant as Tenant's attorney-in-fact. Tenant hereby constitutes and appoints Landlord, its successors and assigns, as Tenant's attorney-in-fact to execute, acknowledge and deliver any such certificate or other document on behalf of Tenant. Tenant agrees that in the event any ground lease encumbering the Land is terminated, (a) Tenant shall attorn to the ground lessor and shall recognize the ground lessor as the Landlord under this Lease, (b) Tenant shall execute and deliver, upon the reasonable request of the ground lessor, an instrument evidencing its agreement to attorn to the ground lessor, and (c) Tenant hereby waives

the provisions of any statute or rule of law which may give Tenant any right of election to terminate this Lease or to surrender possession of the Premises in the event of a termination of the ground lease, and this Lease shall not be affected in any way whatsoever by any such termination of the ground lease. Tenant agrees that upon such attornment, (I) the ground lessor shall be required to provide only those services that (x) are generally and customarily provided in buildings comparable to the Building in downtown Washington, D.C., but not in the event of any force majeure affecting the Building, or (y) are expressly approved by the ground lessor in a non-disturbance agreement with Tenant, (II) the ground lessor shall not be obligated to construct or pay for any improvements required under this Lease, or pay any allowances, concessions or other amounts that may be provided for in this Lease, (III) the ground lessor shall not be bound by any payment of rent under this Lease for more than one (1) month prior to its due date, (IV) the ground lessor shall not be liable for damages for any breach, act or omission of Landlord or any prior landlord under this Lease, or subject to any offsets or defenses which Tenant may have against Landlord or any prior landlord under this Lease, (V) the ground lessor shall not be responsible for the return of any security deposit furnished to Landlord or any prior landlord under this Lease that has not been received by the ground lessor, and (VI) the ground lessor shall not be obligated to recognize the right to possession granted to Tenant under this Lease if Tenant is in default under this Lease beyond the expiration of any cure period provided for herein, and in no event shall the ground lessor be obligated to recognize any right to possession beyond the expiration date of the ground lease, unless specifically agreed to in a separate document executed by an authorized officer of the ground lessor. Tenant further agrees that in the event any proceedings are brought for the foreclosure of any mortgage encumbering the Building, Tenant shall attorn to the purchaser at such foreclosure sale, if requested to do so by such purchaser, and shall recognize such purchaser as the landlord under this Lease, and Tenant waives the provisions of any statute or rule of law, now or hereafter in effect, which may give or purport to give Tenant any right to terminate or otherwise adversely affect this Lease and the obligations of Tenant hereunder in the event any such foreclosure proceeding is prosecuted or completed or any deed in lieu obtained. Tenant agrees that upon such attornment, such purchaser shall not (i) be bound by any payment of annual base rent or additional rent for more than one (1) month in advance, except prepayments in the nature of security for the performance by Tenant of its obligations under this Lease, but only to the extent such prepayments have been delivered to such purchaser; (ii) be bound by any amendment of this Lease made without the consent of any lender providing construction or permanent financing for the Building; (iii) be liable for damages for any act or omission of any prior landlord; (iv) be subject to any offsets or defenses which Tenant might have against any prior landlord (expressly excluding any offset to which Tenant is entitled pursuant to Paragraph 6(d) of Exhibit B hereto); (v) be obligated for construction of any improvements otherwise to be constructed by Landlord under this Lease; or (vi) be obligated under any provision of this Lease setting forth terms of indemnification by Landlord of Tenant. After succeeding to Landlord's interest under this Lease, such purchaser shall perform in accordance with the terms of this Lease all obligations of Landlord arising after the date such purchaser acquires title to the Building. Upon request by such purchaser, Tenant shall execute and deliver an instrument or instruments confirming its attornment.

21.3 (a) After receiving notice from any person, firm or other entity that it holds Superior Lien, no notice from Tenant to Landlord alleging any default by Landlord shall be effective unless and until a copy of the same is given to such Superior Lien Holder; provided,

however, that Tenant shall have been furnished with the name and address of such Superior Lien Holder. The curing of any of Landlord's defaults by such Superior Lien Holder shall be treated as performance by Landlord.

- (b) In addition to the time afforded Landlord for the curing of any default, any such Superior Lien Holder shall have such additional time as may be necessary given the nature and extent of the default (including such time as may be necessary in order to foreclose the mortgage, deed of trust or other similar security instrument, or obtain a deed in lieu therefor or otherwise obtain possession of the Land and Building) after the expiration of the period allowed to Landlord for the cure of any such default within which to cure such default so long as any such holder, trustee or ground lessor acts with reasonable diligence.
- (c) In the event that any lender providing construction or permanent financing or any refinancing for the Building requires, as a condition of such financing, that modifications to this Lease be obtained, and provided that such modifications (i) are reasonable, (ii) do not adversely affect in a material manner Tenant's use of the Premises as herein permitted, (iii) do not increase the rent and other sums to be paid by Tenant hereunder and (iv) do not diminish in a material manner any of Tenant's other rights under this Lease or increase any of Tenant's other obligations or liabilities under this Lease, Landlord may submit to Tenant a written amendment to this Lease incorporating such required changes, and Tenant hereby covenants and agrees to execute, acknowledge and deliver such amendment to Landlord within five (5) days of Tenant's receipt thereof.

ARTICLE XXII HOLDING OVER

22.1 In the event that Tenant shall not immediately surrender the Premises on the date of the expiration of the Lease Term, Tenant shall become a tenant by the month at a base rent and additional rent equal to one hundred fifty percent (150%) of the amount of the annual base rent and all additional rent in effect during the last month of the Lease Term. Said monthly tenancy shall commence on the first day following the expiration of the Lease Term. As a monthly tenant, Tenant shall be subject to all the terms, conditions, covenants and agreements of this Lease. Tenant shall give to Landlord at least thirty (30) days' written notice of any intention to quit the Premises. Tenant shall be entitled to thirty (30) days' written notice to quit the Premises, which notice shall not be given until the expiration of the Lease Term, unless Tenant is in default hereunder, in which event Tenant shall not be entitled to any notice to quit, the usual thirty (30) days' notice to quit being hereby expressly waived. Notwithstanding the foregoing provisions of this Section 22.1, in the event that Tenant shall hold over after the expiration of the Lease Term, and if Landlord shall desire to regain possession of the Premises promptly at the expiration of the Lease Term, then at any time prior to Landlord's acceptance of rent from Tenant as a monthly tenant hereunder, Landlord, at its option may forthwith re-enter and take possession of the Premises without process, or by any legal process in force in the jurisdiction in which the Building is located.

ARTICLE XXIII COVENANTS OF LANDLORD

23.1 Landlord covenants that it has the right to make this Lease for the term aforesaid, and that if Tenant shall pay all rent when due and punctually perform all the covenants, terms, conditions and agreements of this Lease to be performed by Tenant, Tenant shall, during the term hereby created, freely, peaceably and quietly occupy and enjoy the full possession of the Premises without molestation or hindrance by Landlord or any party claiming through or under Landlord, subject to the provisions of this Lease, including, without limitation, Section 23.2 hereof. Tenant acknowledges and agrees that its leasehold estate in and to the Premises vests on the date this Lease is executed, notwithstanding that the term of this Lease will not commence until a future date.

23.2 Landlord hereby reserves to itself and its successors and assigns the following rights (all of which are hereby consented to by Tenant): (i) to change the street address or name of the Building, or the arrangement or location of entrances, passageways, doors, doorways, corridors, elevators, stairs, toilets, or other public parts of the Building (provided same does not materially, adversely affect Tenant's use of or access to the Premises); (ii) to erect, use and maintain pipes and conduits in and through the concealed portions of the Premises; (iii) to grant to anyone the exclusive right to conduct any particular legal business or undertaking in the Building; (iii) in the event that Tenant vacates the Premises prior to the expiration of the Lease Term, to make alterations to or otherwise prepare the Premises for re-occupancy by another tenant without relieving Tenant of its obligation to pay all base rent, additional rent and other sums due under this Lease through such expiration; and (iv) to grant anyone the exclusive right from time to time on a temporary basis to use any portion of the common public areas of the Building (provided it does not materially, adversely affect Tenant's use of or access to the Premises). Landlord shall use commercially reasonable efforts to minimize any interference to the operation of Tenant's business in the Premises as a result of the exercise of such rights; provided that Landlord shall not be required to incur any additional, unusual risk, cost, or expense in connection therewith. Provided Landlord acts prudently and complies with the immediately preceding sentence, Landlord may exercise any or all of the foregoing rights without being deemed to be guilty of an eviction, actual or constructive, or a disturbance or interruption of the business of Tenant or of Tenant's use or occupancy of the Premises.

ARTICLE XXIV PARKING

24.1 Upon the written request of Tenant received by Landlord on or before the Lease Commencement Date, Landlord agrees to make available to Tenant and its employees and to Tenant's permitted subtenants monthly parking permits in an aggregate amount not to exceed one (1) monthly parking permit for each one thousand three hundred fifty (1,350) square feet of above grade rentable area in the Premises (excluding any storage space leased by Tenant) for the parking of standard-sized and compact passenger vehicles,

including standard-sized sport utility vehicles, in the Office Parking Area of the Garage on a non-exclusive, unassigned, first-come, first-served basis. Tenant hereby elects to so purchase the entire amount of unreserved parking permits available to Tenant in accordance with the foregoing. Notwithstanding the foregoing, by providing written notice to Landlord on or before the Lease Commencement Date, Tenant may elect to convert up to five (5) of such unreserved monthly parking permits from Tenant's parking ratio to reserved parking permits, in locations to be designated by Landlord and/or the garage operator in their sole discretion, which notice shall state the number of permits (up to the aforesaid maximum of 5 permits) that Tenant wishes to convert, it being understood and agreed that any such reserved parking permits shall be a part of, and not in addition to, Tenant's total ratio of one permit for every 1,350 rentable square feet of above grade rentable area in the Premises. Further notwithstanding the foregoing, upon the written request of Tenant delivered to Landlord concurrently with Tenant's execution and delivery of this Lease, Landlord agrees to make available to Tenant up to five (5) temporary additional monthly parking permits for the parking of standard-sized and compact passenger vehicles, including standard-sized sport utility vehicles, in the Garage on a non-exclusive, unassigned, first-come, first-served basis (the "Temporary Permits"). Any such Temporary Permits so elected by Tenant shall be subject to all of the terms and conditions of this Article XXIV that are applicable to Tenant's unreserved permits comprising Tenant's parking ratio, except that, on thirty (30) days prior written notice to Tenant, Landlord shall have the right to recapture all or any of such Temporary Permits if Landlord needs the same in connection with the leasing of the Building (i.e., such Temporary Permits may be recaptured for Building occupants, but not for daily parkers that do not occupy the Building), in which event Tenant shall have no further right or obligation with respect thereto. The charge for all such permits shall be the prevailing rates for unreserved and reserved parking permits (as applicable) charged from time to time by Landlord or the operator of the Garage. Notwithstanding the foregoing, Landlord does not guarantee (a) the right to convert any such unreserved monthly parking permits to reserved parking permits following the Lease Commencement Date if and to the extent that Tenant does not provide written notice to Landlord converting such monthly parking permits (up to the aforesaid maximum of 5 permits) on or before the Lease Commencement Date, or (b) the right to any such reserved or unreserved parking permits hereunder if Tenant fails continuously to maintain such permits.

24.2 It is understood and agreed that the Office Parking Area of the Garage will be operated on a self-parking basis, including without limitation, stacked self-park spaces, and that no specific parking spaces will be allocated for use by Tenant. Landlord reserves the right to establish other parking controls, rules or regulations, at any time and in its commercially reasonable discretion. Each user of the Office Parking Area of the Garage will have the right to park in any available unreserved parking space in accordance with regulations of uniform applicability promulgated by Landlord or the Garage operator. Notwithstanding anything herein to the contrary, Landlord hereby reserves the right from time to time to designate any portion of the Office Parking Area of the Garage to be used exclusively by Building visitors, retail patrons to the Building, other tenants of the Building, and/or members of the public.

24.3 Tenant agrees that it and its employees shall observe reasonable safety precautions in the use of the Garage and shall at all times abide by all rules and regulations promulgated by Landlord or the Garage operator governing its use. Tenant's employees having the use of monthly parking permits shall be required to display an identification or parking sticker at all times in all vehicles parked in the Garage. Any car not displaying such a sticker may be towed away at the vehicle owner's expense in accordance with the Garage rules and regulations. In addition, Landlord's and Tenant's use of the Garage shall be subject to all Legal Requirements. If an employee of Tenant or other person who has been issued a permit for standard unreserved parking in the Garage parks in a reserved parking space or in areas of the Garage that are designated as reserved for the exclusive use of tenants other than Tenant, such employee shall be subject to enforcement measures, which may include violation ticketing at 125% of the daily parking rates then in effect in the Garage. If any violation fee is not paid to an attendant at the time of departure, the violator will be billed and if such violation fee is not paid by the violator within thirty (30) days following the date of invoice, such unpaid sums will be charged to and become the responsibility of Tenant.

24.4 The Office Parking Area of the Garage will remain open on Monday through Friday (excluding legal holidays) during the normal hours of operation of the Building on such days. Landlord reserves the right to close the Garage during periods of unusually inclement weather and portions of the Garage during periods of repair, cleaning and/or maintenance. At all times when the Garage is closed, monthly permit holders shall be afforded access to the Garage by means of a magnetic card or other procedure provided by Landlord or the Garage operator.

24.5 It is understood and agreed that Landlord does not assume any responsibility for, and shall not be held liable for, any damage or loss to any vehicles parked in the Garage or to any personal property located therein, or for any injury sustained by any person in or about the Garage, and in no event shall Landlord or any employee, agent or contractor of Landlord be required to monitor or respond to any panic alarms within the Garage.

ARTICLE XXV GENERAL PROVISIONS

- 25.1 Tenant acknowledges that neither Landlord nor any broker, agent or employee of Landlord has made any representations or promises with respect to the Premises or the Building except as herein expressly set forth, and no rights, privileges, easements or licenses are being acquired by Tenant except as herein expressly set forth.
- 25.2 Nothing contained in this Lease shall be construed as creating a partnership or joint venture of or between Landlord and Tenant, or to create any other relationship between the parties hereto other than that of landlord and tenant.

25.3 Landlord recognizes Studley, Inc. ("Broker") as the sole broker procuring this Lease and shall pay said Broker a commission pursuant to a separate agreement between said Broker and Landlord. Landlord and Tenant each represent and warrant to the other that, except as provided in the preceding sentence, neither of them has employed or dealt with any broker, agent or finder in carrying on the negotiations relating to this Lease. Landlord and Tenant shall indemnify and hold the other harmless from and against any claim or claims for brokerage or other commissions asserted by any broker, agent or finder engaged by Landlord or Tenant or with whom Landlord or Tenant has dealt in connection with this Lease, other than the Broker.

25.4 (a) Tenant agrees, at any time and from time to time, upon not less than ten (10) days' prior written notice from Landlord, to execute, acknowledge and deliver to Landlord a true statement in writing (i) certifying that this Lease is unmodified and in full force and effect (or if there have been any modifications, that this Lease is in full force and effect as modified and stating the modifications), (ii) stating the dates to which the rent and any other charges hereunder have been paid by Tenant, (iii) stating whether or not, to the best knowledge of Tenant, Landlord is in default in the performance of any covenant, agreement or condition contained in this Lease, and if so, specifying the nature of such default, (iv) stating the address to which notices to Tenant are to be sent, and (v) stating such other information as Landlord, Lender or any other holder of a mortgage secured by the Building may request on such form as Landlord, Lender or such holder may reasonably request. If Tenant fails to execute, acknowledge and deliver any such written statement within the aforesaid ten (10) day period, and such failure continues for two (2) business days after a second (2nd) written notice from Landlord, then Tenant shall be deemed to have constituted and appointed Landlord as Tenant's attorney-in-fact to execute any such certificate or other document for or on behalf of Tenant. Any such statement delivered by Tenant may be relied upon by any owner of the Building or the Land, any prospective purchaser of the Building or the Land, any mortgagee or prospective mortgagee of the Building or such Landlord's interest therein, or any prospective assignee of any such mortgagee.

(b) On each anniversary of the Effective Date, Tenant shall deliver to Landlord Tenant's financial statements, audited by a certified independent public accountant, for the fiscal year ending in the previous calendar year stating, among other things, Tenant's revenues and net income; provided, however, that during any such period for which (i) Tenant is not a public company and (ii) Tenant does not prepare audited financial statements in the ordinary course of Tenant's business, then Tenant shall satisfy the requirement hereunder by delivering to Landlord Tenant's financial statements certified by Tenant's chief financial officer. Tenant shall make its chief financial officer available to answer any questions Landlord may have concerning such financial statements and shall deliver any additional information reasonably requested by Landlord to clarify or verify the data shown on the statements provided pursuant to the preceding sentence, provided Landlord agrees to hold the financial statements and other such additional information subject to customary confidentiality conditions. Notwithstanding the foregoing, Tenant shall not be required to deliver to Landlord any such financial statements hereunder during any period in which Tenant's then-current, complete annual financial statements, audited by an independent certified public accountant, are publicly available on Tenant's website.

- 25.5 Landlord and Tenant each hereby waive trial by jury in any action, proceeding or counterclaim brought by either of them against the other in connection with any matter arising out of or in any way connected with this Lease, the relationship of landlord and tenant hereunder, Tenant's use or occupancy of the Premises, or any claim of injury or damage.
- 25.6 All notices or other communications required hereunder shall be in writing and shall be deemed duly given if delivered in person (with receipt therefor), or if sent by certified or registered mail, return receipt requested, postage prepaid, or by recognized overnight courier, when received or refused to the following addresses: (i) if to Landlord at c/o Boston Properties, 505 9th Street, N.W., Suite 800, Washington, D.C. 20004, Attn: Regional Counsel, with a copy to Boston Properties, 800 Boylston Street, Suite 1900, Boston, Massachusetts 02199-8103, Attn: General Counsel, (ii) if to Tenant, at the Premises, except that prior to the Lease Commencement Date, notices to Tenant shall be sent to such address as Tenant shall designate and inform Landlord in accordance with this Section 25.6. Either party may change its address for the giving of notices by notice given in accordance with this Section.
- 25.7 If any provision of this Lease or the application thereof to any person or circumstance shall to any extent be invalid or unenforceable, the remainder of this Lease, or the application of such provision to persons or circumstances other than those as to which it is invalid or unenforceable, shall not be affected thereby, and each provision of this Lease shall be valid and enforceable to the fullest extent permitted by law.
- 25.8 Feminine or neuter pronouns shall be substituted for those of the masculine form, and the plural shall be substituted for the singular number, in any place or places herein in which the context may require such substitution.
- 25.9 The provisions of this Lease shall be binding upon, and shall inure to the benefit of, the parties hereto and each of their respective representatives, successors and assigns, subject to the provisions hereof restricting assignment or subletting by Tenant.
- 25.10 This Lease contains and embodies the entire agreement of the parties hereto and supersedes all prior agreements, negotiations and discussions between the parties hereto. Any representation, inducement or agreement that is not contained in this Lease shall not be of any force or effect. This Lease may not be modified or changed in whole or in part in any manner other than by an instrument in writing duly signed by both parties hereto.
- 25.11 This Lease shall be governed by and construed in accordance with the laws of the jurisdiction in which the Building is located, without regard to the conflicts of laws principles.

- 25.12 Article and section headings are used herein for the convenience of reference and shall not be considered when construing or interpreting this Lease.
- 25.13 The submission of an unsigned copy of this document to Tenant for Tenant's consideration does not constitute an offer to lease the Premises or an option to or for the Premises. This document shall become effective and binding only upon the execution and delivery of this Lease by both Landlord and Tenant.
 - 25.14 Time is of the essence of each provision of this Lease.
 - 25.15 Neither this Lease nor a memorandum thereof shall be recorded.
- 25.16 Except as otherwise provided in this Lease, any amounts (whether referenced herein as "Additional Rent" or "additional rent") owed by Tenant to Landlord, and any cost, expense, damage, or liability shall be paid by Tenant to Landlord no later than the later of (i) twenty (20) days after the date Landlord notifies Tenant of the amount of such additional rent or such cost, expense, damage or liability, or (ii) the day the next monthly installment of annual base rent is due. If any payment hereunder is due after the end of the Lease Term, such additional rent or such cost, expense, damage or liability shall be paid by Tenant to Landlord not later than twenty (20) days after Landlord notifies Tenant of the amount of such additional rent or such cost, expense, damage or liability.
- 25.17 All of Tenant's duties and obligations hereunder, including but not limited to Tenant's duties and obligations to pay annual base rent, additional rent and the costs, expenses, damages and liabilities incurred by Landlord for which Tenant is liable, shall survive the expiration or earlier termination of this Lease for any reason whatsoever. Landlord's obligation to refund to Tenant any Security Deposit or overpayment made by Tenant pursuant to Article IV or Article V shall likewise survive the expiration or earlier termination of this Lease.
- 25.18 In the event Landlord is in any way delayed, interrupted or prevented from performing any of its obligations under this Lease, and such delay, interruption or prevention is due to fire, act of God, governmental act, action or inaction (including, without limitation, government delays in issuing any required building, construction, occupancy or other permit, certificate or approval or performing any inspection or review in connection therewith), act(s) of war, terror or terrorism, strike, labor dispute, inability to procure materials, or any other cause beyond Landlord's reasonable control (whether similar or dissimilar) (each a "Force Majeure Event"), then Landlord shall be excused from performing the affected obligations for the period of such delay, interruption or prevention.
- 25.19 Landlord and Tenant hereby represents and warrants to the other that all necessary action has been taken to enter this Lease and that the person signing this Lease on behalf of Landlord and Tenant has been duly authorized to do so.

25.20 Landlord and Tenant agree that the terms and conditions of this Lease shall remain confidential and shall not be disclosed, directly or indirectly, to any individual or entity by either Landlord or Tenant without the express written consent of the other, with the exception of consultants, employees, agents, lawyers, accountants and other professionals employed or retained directly by either or both of the parties to negotiate or work on this Lease who have a legitimate need to know such information and any other disclosures as may be required to comply with applicable Legal Requirements (including without limitation, SEC reporting and disclosure laws) or otherwise required by a court of law or in connection with any other legal arbitration or dispute resolution proceeding. In the event Tenant is required by a court of law or in connection with any other legal arbitration or dispute resolution proceeding to provide this Lease or disclose any of its terms, Tenant shall give Landlord prompt notice of such requirement prior to making disclosure so that Landlord may seek an appropriate protective order. If failing the entry of a protective order Tenant is compelled to make disclosure, Tenant shall only disclose portions of this Lease which Tenant is required to disclose and will exercise reasonable efforts to obtain assurance that confidential treatment will be accorded to the information so disclosed. Any and all public announcements regarding this Lease and any public announcement using either party's name must be approved in writing by such party prior to publication or other dissemination.

25.21 [Intentionally Omitted].

- 25.22 Landlord and Tenant each hereby covenant and agree that each and every provision of this Lease has been jointly and mutually negotiated and authorized by both Landlord and Tenant; and, in the event of any dispute arising out of any provision of this Lease, Landlord and Tenant do hereby waive any claim of authorship against the other party.
- 25.23 The term "days," as used herein, shall mean actual days occurring, including, Saturdays, Sundays and holidays. The term "business days" shall mean days other than Saturdays, Sundays and holidays. If any item must be accomplished or delivered hereunder on a day that it is not a business day, it shall be deemed to have been timely accomplished or delivered if accomplished or delivered on the next following business day.

25.24 This Lease includes and incorporates Rider No.1 and Exhibits A, B, C, D, E, F, G, H, I and J to this Lease.

25.25 As an inducement to Landlord to enter into this Lease, Tenant hereby represents and warrants that: (a) Tenant is not, nor is it owned or controlled directly or indirectly by, any person, group, entity or nation named on any list issued by the Office of Foreign Assets Control of the United States Department of the Treasury ("OFAC") pursuant to Executive Order 13224 or any similar list or any law, order, rule or regulation or any Executive Order of the President of the United States as a terrorist, "Specially Designated National and Blocked Person" or other banned or blocked person (any such person, group, entity or nation being hereinafter referred to as a "**Prohibited Person**"); (b) Tenant is not (nor is it owned or controlled, directly or indirectly,

by any person, group, entity or nation which is) acting directly or indirectly for or on behalf of any Prohibited Person; and (c) from and after the effective date of the above-referenced Executive Order, Tenant (and any person, group, or entity which Tenant controls, directly or indirectly) has not conducted nor will conduct business nor has engaged nor will engage in any transaction or dealing with any Prohibited Person in violation of the U.S. Patriot Act or any OFAC rule or regulation, including, without limitation, any assignment of this Lease or any subletting of all or any portion of the Premises or the making or receiving of any contribution of funds, goods or services to or for the benefit of a Prohibited Person in violation of the U.S. Patriot Act or any OFAC rule or regulation. In connection with the foregoing, it is expressly understood and agreed that (i) any breach by Tenant of the foregoing representations and warranties shall be deemed a default by Tenant under Article XIX of this Lease and shall be covered by the indemnity provisions of this Lease, (ii) Tenant shall be responsible for ensuring that all assignees of this Lease and all subtenants or other occupants of the Premises comply with the foregoing representations and warranties, and (iii) the representations and warranties contained in this subsection shall be continuing in nature and shall survive the expiration or earlier termination of this Lease.

25.26 Landlord represents and warrants to Tenant, as of the date hereof, that there is no mortgage lien affecting the Building or Landlord's interest therein, and Landlord owns a leasehold estate in and to the Land.

ARTICLE XXVI COMMUNICATIONS AND ACCESS; BUILDING RISERS

26.1 Landlord agrees that, provided there does not exist an Event of Default by Tenant under this Lease, Tenant and its contractor shall be permitted non-exclusive access equal to its proportionate share (as determined from time to time based upon the number of rentable square feet of office space Tenant is leasing in the Building from Landlord) of the available space in the Building risers and telecommunications closets, including without limitation the space above the ceilings and below the floors of the Premises, except such risers or closets being utilized exclusively by Landlord or other tenants in the Building (and excluding, in any event, such Building risers and/or telecommunications closets located in mechanical rooms, basement space or other common and/or public areas of the Building) (collectively, "Risers"), at no additional charge therefor, for the sole purpose of installing cabling and telecommunications equipment therein; provided, however, that:

(a) Tenant shall submit to Landlord for Landlord's prior written approval (which approval shall not be unreasonably withheld or delayed) reasonably detailed plans and specifications showing the locations within the Risers where such cabling and equipment will be installed. Tenant shall appropriately mark and/or tag all such cabling and equipment as reasonably required by Landlord to identify the owner and/or user thereof. If any such cabling and/or equipment are installed without Landlord's prior written approval or without such appropriate identification, and Tenant fails to remove same within thirty (30) days after written

notice from Landlord to do so, then Landlord shall have the right to remove and correct such improvements and restore the Risers to their condition immediately prior thereto, and Tenant shall be liable for all expenses incurred by Landlord in connection therewith. Tenant shall not be entitled to use or occupy a disproportionate amount of the available space in the Risers, based upon the proportion of the rentable area then being leased by Tenant to the aggregate rentable office area in the Building. Landlord makes no representation or warranty that the Risers will be adequate to satisfy Tenant's needs. Tenant has previously inspected the Riser space and has satisfied itself as to the adequacy of such space.

- (b) Tenant and its contractor shall coordinate any access to the Risers with Landlord's property manager for the Building.
- (c) Tenant shall pay, as additional rent, all actual, out of pocket costs and expenses reasonably incurred by Landlord in connection with performance of such work by or on behalf of Tenant or its contractors, agents or employees.
- (d) Tenant and its contractor shall conduct their work in a manner that shall minimize disruption and inconvenience to other tenants and occupants of the Building.
- (e) During the installation, maintenance, repair, replacement, and removal of such cabling and equipment, Tenant shall keep all public areas of the Building where such work is being performed neat and clean at all times and Tenant shall remove or cause all debris to be removed from the Building at the end of each work day.
- (f) Tenant shall promptly repair, at its sole cost and expense, any damage done to the Building or to the premises of any other tenant in the Building and to any electrical, mechanical, HVAC, sprinkler, life safety and other operating system serving the Building or other common areas appurtenant to the Building that are caused by or arise out of any work performed by Tenant or its contractor pursuant to this Section.
- (g) Any contractor performing such work shall be subject to the prior written approval of Landlord, which approval shall not be unreasonably withheld, conditioned or delayed.
- (h) In performing such work, Tenant and its contractor shall observe Landlord's rules and regulations regarding the construction, installation, and removal of Tenant improvements in the Building, which rules and regulations, together with any modifications thereto, shall be provided to Tenant, in writing, prior to enforcement.
- (i) Tenant shall be solely responsible at its sole cost and expense to correct and to repair any work or materials installed by Tenant or Tenant's contractor. Landlord shall have no liability to Tenant whatsoever on account of any work performed or material provided by Tenant or its contractor.
- (j) Tenant shall remove, at Tenant's sole cost and expense, all cabling and equipment installed by or on behalf of Tenant or other occupants of the Premises from the

Risers, by no later than the expiration or earlier termination of this Lease, if and to the extent removal is required by the National Electric Code or other applicable Legal Requirements. All damages and injury to the Risers, the Premises or the Building caused by such removal shall be repaired by Tenant, at Tenant's sole expense and in a manner approved by Landlord.

- (k) Landlord's representative shall have the right to inspect any work performed by Tenant or its contractor during the normal hours of operation of the Building or such other hours as Landlord may request.
- (l) All work done and materials furnished by Tenant and/or its contractor shall be of good quality, shall be performed in a good and workmanlike manner and in accordance and compliance with all applicable Legal Requirements and the other applicable provisions of this Lease.
 - (m) Any casualty or other damage to all or any portion of the Risers shall not affect Tenant's obligations, duties, or responsibilities under this Lease.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK. SIGNATURE PAGE FOLLOWS.]

IN WITNESS WHEREOF, Landlord and Tenant have executed this Lease under seal on or as of the day and year first above written.

LANDLORD:

SQUARE 54 OFFICE OWNER LLC, a Delaware limited liability company

By: BP/DC Properties, Inc., a Maryland corporation, its sole member and manager

By:

/s/ Raymond A. Ritchey (SEAL) Name: Raymond A. Ritchey

Title: Executive Vice President

TENANT:

VANDA PHARMACEUTICALS INC., a Delaware corporation

/s/ James P. Kelly (SEAL)

Name: James P. Kelly

Title: Senior Vice President and Chief Financial Officer

RIDER NO. 1

RENEWAL

THIS RIDER NO. 1, RENEWAL ("**Rider**"), is attached to and made a part of that certain Lease dated July 25, 2011 ("**Lease**"), by and between SQUARE 54 OFFICE OWNER LLC, a Delaware limited liability company ("**Landlord**"), and VANDA PHARMACEUTICALS INC., a Delaware corporation ("**Tenant**"). The terms used in this Rider which are defined in the Lease have the same meanings as provided in the Lease.

WITNESSETH, that for and in consideration of Tenant's entering into the Lease described above, and other good and valuable consideration, and intending to be legally bound hereby, Landlord hereby grants to Tenant the subordinate right to renew the initial term of the Lease upon the following terms and conditions:

- 1. Landlord hereby grants to Tenant the subordinate and conditional right, exercisable at Tenant's option, to renew the term of the Lease for one (1) additional term of five (5) years. If timely exercised and if the conditions applicable thereto have been satisfied, such renewal term ("Renewal Term") shall commence immediately following the end of the initial term provided in Section 2.1 of the Lease. The right of renewal herein granted to Tenant shall be subject to, and shall be exercised in accordance with, the following terms and conditions:
- (a) Tenant shall exercise its right of renewal with respect to the Renewal Term by giving Landlord written notice of the exercise thereof ("Renewal Option Notice") not less than twelve (12) months ("Outside Notice Deadline") and not more than fourteen (14) months prior to the expiration of the initial term of the Lease. In the event that a Renewal Option Notice is not given in a timely manner, Tenant's right of renewal with respect to the Renewal Term shall lapse and be of no further force or effect. If Tenant is in default under the Lease on the date the Renewal Option Notice is given or any time thereafter, on or before the commencement date of the Renewal Term, then, at Landlord's option, the Renewal Option Notice shall be totally ineffective and Tenant's right of renewal as to the Renewal Term shall lapse and be of no further force or effect. Notwithstanding the foregoing, if Tenant is in default under this Lease on the date Tenant delivers to Landlord the Renewal Option Notice or such default occurs following Tenant's delivery of the Renewal Option Notice, and Tenant cures such default in full within the applicable notice and cure period provided pursuant to Section 19.1 of the Lease, but in all events on or before the Outside Notice Deadline, then Tenant's right of renewal hereunder shall not be voided on account of such default.
- (b) Promptly following Landlord's timely receipt of the Renewal Option Notice for the Renewal Term, Landlord and Tenant shall commence negotiations concerning the amount of annual base rent which shall be payable during each year of the Renewal Term and the Lease security that may be required, it being intended that such annual base rent shall be equal to the then prevailing fair market rent for the Premises. The parties shall have thirty (30) days after Landlord's receipt of the Renewal Option Notice in which to agree on the annual base rent which shall be payable during each year of the Renewal Term and the Lease

security that may be required. The parties shall be obligated to conduct such negotiations in good faith. Among the factors to be considered by the parties during such negotiations shall be (i) the general office rental market for Class A office buildings in the Market Area, (ii) rental rates then being obtained (or quoted if comparables are not readily available) by other building owners for office buildings of comparable size, location and quality to the Building in the Market Area, (iii) the rental rates then being obtained by Landlord for comparable office space, in "as is" condition, in the Building, (iv) escalations and pass throughs of Operating Expenses as provided in the Lease, (v) concession packages then being obtained (or offered if comparables are not readily available) by other building owners for office buildings in the Market Area of comparable size, location and quality to the Building, (vi) concession packages then being obtained by Landlord for comparable office space in "as-is" condition in the Building, and (vii) consideration of what would constitute an appropriate security deposit securing the performance of Tenant's obligations with respect to the Renewal Term, given Tenant's creditworthiness at the time, any out-of-pocket expenditures by Landlord in connection with such renewal, and prevailing market conditions at the time, and Tenant shall be required to post any such security as a condition to the Renewal Term. If the parties agree on the base rent payable during each year of the Renewal Term, they shall promptly execute an amendment to the Lease stating the rent so agreed upon.

(c) If, during such thirty (30) day period referred to in subparagraph (b) above, the parties are unable to agree on the base rent payable during the Renewal Term, then (i) the fair market rent and Lease security and (ii) the related fair market concessions, abatements and allowances, if any, that will be applicable thereto shall be determined in accordance with the procedure set forth in this subparagraph (c). Within ten (10) days after expiration of such thirty (30) day period, the parties shall appoint a real estate broker ("Broker") who shall be mutually agreeable to both Landlord and Tenant, shall be a member of the National Association of Realtors or the Greater Washington, D.C. Association of Realtors, and shall have at least ten (10) years relevant experience in office rentals in the Market Area. If the parties are unable to agree on a Broker within such ten (10) day period, then each party, within five (5) days after the expiration of the aforesaid ten (10) day period, shall appoint a Broker (with the same qualifications) and the two Brokers shall together appoint a third Broker with the same qualifications ("Third Broker"). The original agreed upon Broker, if applicable, or two Brokers appointed shall determine, within thirty (30) days after appointment, the then fair market base rent and Lease security (and related fair market concessions, abatements and allowances, if any) that will be applicable to the Premises for the Renewal Term. Among the factors to be considered by the Broker or Brokers in determining the fair market base rent and Lease security for the Premises (and related fair market concessions, abatements and allowances, if any) that will be applicable during the Renewal Term shall be those factors set out in subparagraph (b) above. The fair market rent arrived at by the Broker, if only one, (or if more than one Broker and the original two (2) Brokers appointed by the parties agree on a fair market rent), shall be used as the fair market base rent for the Renewal Term. If more than one Broker is appointed and the Brokers reach different determinations, and the parties are unable to reach agreement within five (5) business days of receipt of both Brokers' determinations, then, the Third Broker shall determine within thirty (30) days of receipt of both Brokers' determinations, which of the Brokers' determination of the fair market base rent and lease security for the Premises (and related fair market

concessions, abatements and allowances, if any) will be applicable for the Renewal Term. The fair market base rent and Lease security (market concessions, abatements and allowances, if any) selected by the Third Broker shall be used for the Renewal Term. Landlord and Tenant shall each bear the cost of its Broker and shall share equally the cost of the Third Broker.

- (d) During the Renewal Term, all the terms, conditions, covenants and agreements set forth in the Lease shall continue to apply and be binding upon Landlord and Tenant, except that (i) the annual base rent payable during each year of the Renewal Term shall be the amount agreed upon by Landlord and Tenant in the manner provided in Paragraphs 1(b) and (c) above, and (ii) in no event shall Tenant have the right to renew the term of the Lease, or any renewal term thereof, beyond the expiration of the Renewal Term, and (iii) no abatements, allowances or other concessions shall apply during the Renewal Term, except to the extent otherwise agreed to by the parties in accordance with this Rider, and (iv) Landlord shall not be responsible for any brokerage commissions in connection with the Renewal Term.
- 2. Tenant's rights under this Rider are personal to and may be exercised only by Tenant and shall not be exercisable by any assignee or subtenant of Tenant (except for any assignee that is a Permitted Transferee pursuant to Section 7.4 of the Lease).
- 3. Tenant shall not be entitled to renew the Term of this Lease, and Tenant's rights under this Rider shall lapse and be of no further force or effect, if, at the time Tenant would otherwise be entitled to exercise its rights of renewal (or at any time thereafter prior to the commencement of the Renewal Term), Tenant is leasing less than one hundred percent (100%) or occupying less than seventy-five percent (75%) of the rentable area contained in the Premises as of the Effective Date.
- 4. Notwithstanding anything herein or in the Lease to the contrary, Tenant's rights under this Rider are subject and subordinate to the right of Hunton & Williams LLP (and its successors and assigns) to expand into the Premises pursuant to expansion rights contained in Hunton & Williams LLP's lease pursuant to the mutual agreement of Landlord and Hunton & Williams LLP.

CERTIFICATION OF PRINCIPAL EXECUTIVE OFFICER PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

- I, Mihael H. Polymeropoulos, certify that:
- 1. I have reviewed this quarterly report on Form 10-Q of Vanda Pharmaceuticals Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a. designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b. designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c. evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d. disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a. all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 7, 2011

/s/ Mihael H. Polymeropoulos, M.D.

Mihael H. Polymeropoulos, M.D.

Chief Executive Officer
(Principal Executive Officer)

CERTIFICATION OF PRINCIPAL EXECUTIVE OFFICER PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, James P. Kelly, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Vanda Pharmaceuticals Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a. designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b. designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c. evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d. disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a. all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Certification

Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002

(Subsections (a) and (b) of Section 1350, Chapter 63 of Title 18, United States Code)

Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (subsections (a) and (b) of section 1350, chapter 63 of title 18, United States Code), each of the undersigned officers of Vanda Pharmaceuticals Inc., (the "Company"), does hereby certify, to the best of such officer's knowledge, that:

The Quarterly Report on Form 10-Q for the quarter ended September 30, 2011 (the Form 10-Q) of the Company fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, and the information contained in the Form 10-Q fairly presents, in all material respects, the consolidated financial condition and results of operations of the Company.

Date: November 7, 2011	/s/ Mihael H. Polymeropoulos, M.D.
	Mihael H. Polymeropoulos, M.D. Chief Executive Officer (Principal Executive Officer)
Date: November 7, 2011	/s/ James P. Kelly
	James P. Kelly Senior Vice President and Chief Financial Officer (Principal Financial Officer and Principal Accounting Officer)

A signed original of this written statement required by Section 906 has been provided to the Company and will be retained by the Company and furnished to the Securities and Exchange Commission (SEC) or its staff upon request. This certification "accompanies" the Form 10-Q to which it relates, is not deemed filed with the SEC and is not to be incorporated by reference into any filing of the Company under the Securities Act of 1933, as amended, or the Securities Exchange Act of 1934, as amended (whether made before or after the date of the Form 10-Q), irrespective of any general incorporation language contained in such filing.